## University of Nebraska - Lincoln DigitalCommons@University of Nebraska - Lincoln

Dissertations and Theses in Biological Sciences

Biological Sciences, School of

Summer 8-2016

## Characterization of the Role of Host and Dietary Factors in the Establishment of Bacteria in the Gastrointestinal Tract

Janina A. Krumbeck University of Nebraska - Lincoln, janina.krumbeck@gmail.com

Follow this and additional works at: http://digitalcommons.unl.edu/bioscidiss Part of the <u>Biology Commons</u>, <u>Dietetics and Clinical Nutrition Commons</u>, and the <u>Microbiology</u> <u>Commons</u>

Krumbeck, Janina A., "Characterization of the Role of Host and Dietary Factors in the Establishment of Bacteria in the Gastrointestinal Tract" (2016). *Dissertations and Theses in Biological Sciences*. 87. http://digitalcommons.unl.edu/bioscidiss/87

This Article is brought to you for free and open access by the Biological Sciences, School of at DigitalCommons@University of Nebraska - Lincoln. It has been accepted for inclusion in Dissertations and Theses in Biological Sciences by an authorized administrator of DigitalCommons@University of Nebraska - Lincoln.

Characterization of the Role of Host and Dietary Factors in the Establishment of Bacteria

in the Gastrointestinal Tract

by

Janina A. Krumbeck

#### A DISSERTATION

Presented to the Faculty of

The Graduate College at the University of Nebraska

In Partial Fulfillment of Requirements

For the Degree of Doctor of Philosophy

Major: Biological Sciences

(Genetics Cell & Molecular Biology)

Under the Supervision of Professors Robert W. Hutkins and Jens Walter

Lincoln, Nebraska

August, 2016

# Characterization of the Role of Host and Dietary Factors in the Establishment of Bacteria in the Gastrointestinal Tract

Janina A. Krumbeck, Ph.D. University of Nebraska, 2016

Advisors: Robert W. Hutkins and Jens Walter

Probiotic bacteria and synbiotics are used as therapeutic and prophylactic agents. The majority of probiotic and synbiotic applications contain bacterial strains that are allochthonous to the human gastrointestinal (GI) tract. Accordingly, many bacterial strains do not survive digestion, or are not capable of persisting and competing the resident gut microbiota, and are therefore washed out of the GI tract shortly after the treatment is discontinued. This might reduce the health effects of these treatments. Therefore, research is needed to address the ecological challenges that probiotic strains encounter in the GI tract in order to develop probiotic regimens. Determining which ecological factors are limiting the colonization of bacteria remains a challenge. To gain insight into the complex interplay between host and microbe, we chose *Lactobacillus reuteri* and its rodent host as a model to investigate which genes of *L. reuteri* contribute to tolerance towards host gastric acid secretion. We established the urease cluster as the predominant factor in mediating resistance to gastric acid, and a mutation of this cluster resulted in substantially decreased population levels of *L. reuteri* in mice.

Secondly, we established a method to select for synergistic synbiotic combinations. Based on *in vivo* selection (IVS), autochthonous putative probiotic strains are enriched in the GI tract of subjects by the continued consumption of a prebiotic. We

used IVS to select a strain of *Bifidobacterium adolescentis* that became enriched in a human feeding trail with galactooligosaccharides (GOS). Here we have shown that the synbiotic combination of *Bifidobacterium adolescentis* IVS-1 and GOS significantly enriched for the putative probiotic component in rats. IVS-1 became the most dominant operational taxonomic unit in the GI tract, outcompeting the resident *Bifidobacterium* species. Similarly, we tested this synbiotic in a human trial with obese adults. In this random, placebo-controlled parallel arm study, the synbiotic combination of IVS-1 and GOS led to establishment of IVS-1 in significantly higher numbers in the GI tract than a commercial synbiotic.

Together, the studies presented in this dissertation allowed new insights into the colonization factors of a true GI symbiont, which could contribute to the development of improved probiotics, and provided novel insight into a rational selection of probiotics and synbiotics.

#### ACKNOWLEDGEMENTS

I would like to send my greatest appreciation to my two advisors Dr. Hutkins and Dr. Walter. Your guidance, patience, support, and advice were invaluable to me throughout these years at the University of Nebraska, Lincoln. You both have been indispensable in your contribution to these projects and my professional and personal development. Thank you.

I would like to thank Dr. Harris for all his help and advice during the rat study. I would also like to thank the other members of my committee, Drs. Brown and Moriyama, for their valuable advice throughout my pursuit of this degree. Your input has helped me to develop as a scientist and improved my work.

Thanks to Dr. Peterson, Dr. Ramer-Tait, Robert Schmaltz, and Brandon White for their help in the mouse facility. Thank you Dr. Marsteller for spending countless hours gavaging mice for me and all your help during the mouse experiments, and thank you Dr. Frese for training me in the lab and all your substantial contributions to the mouse study. Without all of you the mouse study would not have been possible.

Special thanks to Dr. Keshavarzian and Dr. Rasmussen at the Rush University Medical Center in Chicago and all the people in their team for their efforts conducting the human study and their support. Without their engagement it would have not been possible to conduct this trial.

Special thanks to my friends and laboratory colleagues of the Walter and Hutkins Labs, who shared their knowledge with me. It has been great sharing all these years with you. Thank you Dr. Martinez for training me in bioinformatics, and to Dr. Maldonado-Gomez for her development of strain specific primers and help in statistical analyses.

To all my friends in Germany, thank you for always making me feel like I have never left, you all are so special to me. Thank you to my friends in Lincoln, and especially Nathan, you have made these four years so special to me.

I would have not accomplished any of this without the love and endless support of my family. You are so dear to my heart, thank you for always supporting me no matter what.

#### PREFACE

This dissertation is comprised of five chapters. Chapter 1 focuses on the current literature on synbiotic applications, with an emphasis on clinical studies. In particular, the claimed health benefits of synbiotic applications and the implications of recent studies on future design of synbiotics to promote gastrointestinal health are addressed. Chapter 2 describes our published research on the ecological role of genes that mediate acid resistance in *Lactobacillus reuteri* during colonization of the rodent gastrointestinal tract (Krumbeck et al. 2015). In Chapter 3 our published work on the novel concept of *in vivo* selection is introduced, which can be applied to identify bacterial strains that possess enhanced ecological performance in synbiotic applications (Krumbeck et al. 2015). Chapter 4 describes the results of a human clinical trial that applied a synbiotic combination that has been selected by *in vivo* selection. Finally, Chapter 5 provides a conclusion that summarizes the findings provided in this thesis.

## **Table of Contents**

List of Figures			
List of Table	List of Tables		
Chapter 1	Recent developments on formulating synbiotics to improve		
	gastrointestinal health	1	
1.1	Purpose of review	1	
1.2	Introduction	1	
1.3	Functional importance of the colonic microbiota	2	
1.4	Addressing intestinal health with pro-, pre- and synbiotics	5	
1.5	Synbiotic concepts	7	
1.6	Synbiotics and their outcome on human health in clinical studies	12	
1.7	Commercial synbiotics: recent developments and future prospects	38	
1.8	Remaining questions and specific aims	40	
1.9	References	43	
Chapter 2	Characterization of the ecological role of genes mediating acid		
	resistance in Lactobacillus reuteri during colonization of the		

	C	
	gastrointestinal tract	71
2.1	Summary	71
2.2	Introduction	72

2.3	Result	ts	/4
	2.3.1	Selection of genes of L. reuteri 100-23 predicted to be	
		involved in acid resistance	'4
	2.3.2	In vitro characterization of putative acid resistance genes 7	7
	2.3.3	Importance of acid resistance genes during colonization	
		of the mouse GI tract	30
	2.3.4	Urease activity is regulated by pH	33
2.4	Discus	ssion	86
2.5	Experi	imental procedures ethics statement	93
	2.5.1	Bacterial strains and media used in the study 9	13
	2.5.2	Determination of genes predicted to be involved in acid	
		resistance of <i>L. reuteri</i> 100-23	94
	2.5.3	Derivation of mutants	4
	2.5.4	<i>In vitro</i> acid survival assay	5
	2.5.5	Determination of genes' role in <i>in vivo</i> acid resistance 9	5
	2.5.6	Determination of pH regulation of urease activity 9	7
	2.5.7	Measurement of urease activity	7
	2.5.8	RNA extraction from <i>L. reuteri</i> cell cultures	98

	2.5.9	Determination of gene expression by quantitative reverse
		transcription PCR (qRT-PCR)
	2.5.10	Statistical analysis
2.6	Ackno	wledgements
2.7	Refere	nces

Chapter 3	In vivo selection to identify bacterial strains with enhanced
	ecological performance in synbiotic applications
3.1	Abstract
3.2	Introduction
3.3	Materials and methods
	3.3.1 Isolation of <i>in vivo</i> -enriched bifidobacteria from humans 114
	3.3.2 <i>In vitro</i> growth on GOS
	3.3.3. Strain-specific primer design and validation
	3.3.4 Quantitative real-time PCR
	3.3.5 Administration of the probiotic, prebiotic, and synbiotic to rats 117
	3.3.6 Rat study design
	3.3.7 Evaluation of host physiological parameters in rats 119

	3.3.8	Illumina 16S RNA sequencing and sequence analysis 120
	3.3.9	Microbial community analysis
	3.3.10	Statistical analysis
	3.3.11	Nucleotide sequence accession number
3.4	Result	s
	3.4.1	In vivo selection of B. adolescentis IVS-1
	3.4.2	Test of the synbiotic combination using rats on a high-fat diet 127
	3.4.3	Experiments in rats demonstrate strong synergism between
		IVS-1 and GOS
	3.4.4	16S rRNA sequencing confirms synergism between IVS-1
		and GOS <i>in vivo</i>
	3.4.5	Community-wide characterization of effects on gut microbiota 132
	3.4.6	Systematic analyses of associations between members of
		the gut microbiota
3.5	Discus	sion
3.6	Supple	emental material
3.7	Ackno	wledgements
3.8	Refere	nces

Chapter 4	Funct	ional characterization of a rationally selected synbiotic
	applic	ation in obese adults
4.1 /	Abstract .	
4.21	Introducti	on
4.3 1	Methods	
4.4 1	Results .	
	4.4.1	Subject demographics and clinical characteristics
	4.4.2	Test of synergy of GOS in addition to IVS-1 and Bb12 169
	4.4.3	Community-wide characterization of effects on gut microbiota
		confirms ecological advantage of GOS and IVS-1 compared to
		Bb12 and GOS
	4.4.4	Systematic analyses of associations between members of
		the gut microbiota
	4.4.5	Systematic analyses of subjects that showed IVS-1 per-
		sistence after treatment termination
	4.4.6	Analysis of anthropometric markers and
		gastrointestinal symptoms
4.5	Discus	ssion
4.6	Conclu	usion

х

4.7	References	196
4.8	Supplements	208
Chapter 5	Conclusion and future direction	218
	5.1 Conclusions	218
	5.2 Future direction	222

## List of Figures

1.1	Number of publications on the topic "synbiotic" over the past
	15 years
1.2	Health claims made for synbiotics in human populations 13
2.1	Determination of the capacity of genes (Table 2.1) to confer
	survival under acidic conditions
2.2	Mouse competition experiment with mutant and wild-type strains
	in ex-germ-free C57BL/6J mice treated with omeprazole, sham or
	no treatment
2.3	Bacterial growth (OD 600) of L. reuteri 100-23 (continuous line,
	empty symbols, left ordinate axis) and the pH of the media (dotted
	line, full symbols, right ordinate axis) with (triangle symbol) and
	without urea (square symbol)
2.4	Overview of metabolic pathways of genes assessed in this study 88
3.1	In vivo selection to identify putative probiotic strains to be used in
	synbiotic applications
3.2	Test of a synbiotic combination of <i>B. adolescentis</i> IVS-1 and GOS
	in a high-fat-diet rat model
3.3	Characterization of the rat colonic microbiota composition by

	Illumina sequencing of 16SrRNAtags	33
3.4	Correlation analysis of colonic taxa present in rats fed a high-fat	
	diet supplemented with or without a probiotic (IVS-1), a prebiotic	
	(GOS), or a synbiotic (IVS-1 plus GOS) or a standard diet 13	36
3.S1	Growth of Bifidobacterium adolescentis IVS-1 in basal MRS supple-	
	mented with GOS, 0.16% lactose (residual sugar present in the	
	commercial GOS), or basal MRS without carbohydrates 14	44
4.1	Time line for synbiotic trial	55
4.2	Test of <i>in vivo</i> selected synergistic synbiotic application compared	
	to a commercial synbiotic	74
4.3	Quantification of absolute cell numbers of bifidobacteria in fecal	
	samples by genus-specific qPCR	76
4.4	Prediction analysis of IVS-1 persisters and non-persisters	34
4.5	Prediction analysis of IVS-1 persisters and non-persisters	35
4.6	NMDS plot of beta diversity analysis based on Bray-Curtis distance	
	of baseline samples between IVS-1 persisters and non-persisters 18	36
4.S1	Average abundances of taxa in fecal samples of subjects consuming	
	dietary treatments	)8

Correlation analysis of fecal genera present with at least 0.5% 4.S2

	abundance in subjects consuming dietary treatments
4.S3	Correlation analysis of fecal OTUs present with at least 1%
	abundance in subjects consuming dietary treatments
4.S4	Ratio of Bacteroides and Prevetolla taxa present in fecal samples of
	subjects consuming dietary treatments
4.S5	Alpha diversity comparison between persisters and non-persisters 211
4.S.6	Average abundances of taxa in fecal samples of subjects consuming
	dietary treatments based on intend to treat analysis
4.S.7	Ratio of Bacteroides and Prevetolla taxa present in fecal samples of
	subjects consuming dietary treatments based on intend to treat
	analysis
4.S.8	Alpha diversity comparison between persisters and non-persisters
	based on intend to treat analysis

## List of Tables

1.1	Synergistic synbiotics reported in the literature
1.2	Overview of published meta-analyses on synbiotic treatments
	(adapted and updated from Krumbeck et al., 2016)
1.3	Overview of systematic reviews on synbiotic treatments
1.4	Overview of Meta-analyses synbiotic treatments that combined
	pro- and synbiotic trials into one analyses
2.1	Genes selected for functional characterization
2.2	Strains used in this study
3.1	Proportions of bacterial taxa significantly influenced by dietary
	treatments
3.S1	Composition of standard and high-fat diets
3.S2	Body weight, relative epididymal fat pad weight, consumed drinking
	water, consumed IVS-1 and GOS, and host physiological markers 143
4.1	Treatment groups
4.2	Demographic and metabolic characteristics of study subjects 171
4.3	Proportions of bacterial taxa significantly influenced by dietary
	treatments
4.4	Percent change in anthropometrics in participants
4.5	Gastrointestinal symptoms by supplementation group

xvi

## 4.S.1 Proportions of bacterial taxa significantly influenced by dietary

Treatments based on intend to treat analysis	•		•	•	•	2	21	2
--	---	--	---	---	---	---	----	---

#### Chapter 1

#### Recent developments on formulating synbiotics to improve gastrointestinal health.

#### **1.1 Purpose of review**

Research on combining probiotics and prebiotics as synbiotics to enhance human and animal health has accelerated in the past ten years. Included are many clinical trials that have assessed a wide variety of synbiotic formulations. In this review, we summarize those recent clinical studies as well as other research and commercial applications of synbiotics. In particular, we address the claimed health benefits of synbiotic applications and the implications of recent studies on future design of synbiotics to promote gastrointestinal health.

#### **1.2 Introduction**

The impact of the gastrointestinal (GI) microbiota on the health of humans and animals is now one of the most studied fields in biology and medicine. New discoveries made during the past 20 years have dramatically changed the way that clinicians and researchers associate food and diet with health and disease. While the microbiota that resides in the gut has long been considered as important to health, the methods and techniques necessary to gain an appropriate appreciation of this complex microbial ecosystem have just been developed within the last decade. In addition, how this microbiota interacts with the host and how the composition and activity of a healthy state microbiota is distinguished from an unhealthy or dysbiotic state had also been experimentally difficult questions to address. While modern techniques now allow the recognition of a dysbiotic microbial state, the identification of cause and effect between a dysbiotic microbiota and a disease phenotype remains challenge.

#### **1.3 Functional importance of the colonic microbiota**

The human gastrointestinal tract is colonized by approximately  $10^{14}$  microbial cells, with the majority ( $10^{11} - 10^{12}$  per gram) residing in the colon and less than  $10^2 - 10^3$  per gram in the stomach and small intestine (1, 2). This microbiota serves several critical physiological functions. It protects the host from invasive pathogenic microorganisms by competing with them for nutrients and niches, as well as by resistance against infections (3–5). Commensal organisms may also produce a variety of antimicrobial substances, including bacteriocins and other antagonistic peptides and small molecules (6, 7). In addition, the microbiota aids in the development of the adaptive and innate immune system, produces essential vitamins, amino-acids and other metabolites, and facilitates utilization of nutrients, especially polymeric carbohydrates (8). Finally, the microbiota contributes caloric energy to the host. Assuming a typical European diet is consumed, the gut microbiota can potentially yield as much as 140-180 kcal a day via fermentation of the 50-60 g of carbohydrates that escaped host metabolism (9).

The extent and rate of carbohydrate digestion and utilization in humans depends primarily on anatomical location. Initially, complex carbohydrates are hydrolyzed in the mouth via amylases, and starch and glycogen are further hydrolyzed and broken down into sugars, which are absorbed in the stomach. In the large intestine, indigestible substrates, including various dietary fibers and carbohydrates that were not absorbed in the small intestine, are hydrolyzed and fermented by bacteria. The proximal part of the large intestine is responsible for most of the absorption of the short chain fatty acids (mainly acetate, butyrate and propionate) that are produced by the colonic bacteria from fiber fermentation at a rate of approximately 0.5- 0.6 mole per day (1, 9, 10), depending on the microbiota composition, the nature of the fermentable carbohydrate, and the dietary intake (6).

Short chain fatty acids (SCFA) have several beneficial effects on host health (11, 12). As the most important and preferred energy source for colonocytes (13), SCFA promote epithelial integrity (11). Additionally, SCFA affect the thickness of the mucus layer, support epithelial cell survival, and regulate expression of tight junction proteins (6, 14). Disruption of gut integrity has been attributed to serious intestinal diseases, including celiac disease, inflammatory bowel disease, and colorectal cancer (12, 15, 16). The local and systemic immunomodulatory properties of SCFA include the suppression of NF-κB activity (17–19) and support of increased infiltration of immune cells into the lamina propria. In addition, SCFA have anti-inflammatory properties by modulating immune cell chemotaxis, reactive oxygen species, and cytokine release (11). SCFA also regulate colonic mobility and blood flow and can influence colon pH, which has a direct impact on the uptake and absorption of nutrients and electrolytes (20).

Butyrate formation by the colonic microbiota is of particular interest since this compound has been shown to have multiple biological effects. Butyrate has antiinflammatory properties, inhibits IL-12 and up regulates IL-10 in monocytes (19, 21). In addition, butyrate has signaling capacities via G-protein coupled receptors (14, 16, 22, 23) and increases levels of anorectic hormones like PYY and GLP-1, that contribute to energy metabolism and appetite control (6). Butyrate also induces apoptosis of neutrophils (24), and has anticancer activity in several human cell lines (25–27).

The colon offers an especially favorable environment for anaerobic microbes, with high quantities of nutrients that escaped host digestion, a thick mucus layer secreted by a higher number of Goblet cells, reduced intestinal motility, and a favorable pH (28). Since the majority (approximately 70%) of the gut microbiota reside in the large intestine, these organisms may have a profound effect on energy storage, host metabolism, and intestinal health (1). While the microbiota provides many beneficial effects on the host, the composition of an individual's microbiota may also predispose that individual for certain intestinal as well as systemic diseases, including obesity and diabetes. Importantly, the microbiota is not only shaped by host genetic factors and endogenous factors (gastric acid and bile), but also by dietary components that favor specific taxa or groups of bacteria in the colon by promoting their growth or activity. Therefore, establishment of bacteria that are associated with a stable and healthy microbiota may be facilitated by dietary strategies. However, short-term dietary strategies have shown that overall the gut microbiota is exceptionally stable and resilient, since most of the alterations induced to the gut microbiota by dietary treatments are only temporary and the pre-treatment conditions are re-established once the treatment is discontinued (29–31). Still, the dietary approach is now one of the most promising methods for correcting bacterial dysbiosis and restoring homeostasis. In particular, foods and supplements containing prebiotics, or combined with probiotics as a synbiotic, have considerable potential for promoting gut health.

#### 1.4 Addressing intestinal health with pro-, pre- and synbiotics

Among the first dietary products used as intestinal therapeutic agents were probiotics. Indeed, what we now call "probiotics" have been produced and consumed for more than 100 years (32, 33), long before the term was actually defined. Probiotics are currently defined as "live microorganisms that, when administered in adequate amounts, confer a health benefit on the host" (FAO/WHO 2001-2014). There are hundreds of probiotic strains and products in the marketplace, and many clinicians recommend probiotics to patients for a variety of conditions, including antibiotic-associated diarrhea, management of acute gastroenteritis, general gastrointestinal disorders, treatment of mild ulcerative colitis, and for improved lactose digestion (34). While consumers have a general understanding of probiotics (35), the definition itself has been controversial among researchers and regulators (34). It has been criticized, for example, as being too broad or not being broad enough (34). Leaving the discussion about the definition itself aside, it is interesting to note that in Europe, and the U.S., no health claims for probiotic products have been approved by regulators. In contrast, Canada has accepted a limited number of claims about the nature of probiotic microorganisms (36), as has Japan (37). In Australia and New Zealand, products can be labelled as probiotics, but this may change as new legislation was passed in 2013 (effective in 2016) that regulates nutrition content and health claims on food labels and in advertisements (38).

Despite the hesitation of regulators to confer health claims for probiotics, clinical evidence continues to emerge suggesting that probiotics can be effective for a range of conditions, including constipation, irritable bowel syndrome, and lactose intolerance (6, 39). Systematic and meta-analyses have shown that probiotics may aid the treatment of antibiotic associated diarrhea (40), the prevention of necrotizing enterocolitis in preterm neonates (41), the induction of remission and maintenance of IBD (42), the prevention and control of hyperglycemia (43), improve levels of total cholesterol HDL and TNF- $\alpha$  in patients of non-alcoholic fatty liver disease (44), and reduce glucose, insulin, and HOMA-IR in diabetes patients (45). In addition, the effectiveness of probiotics for preventing or reducing severity of infectious and antibiotic-associated diarrhea and respiratory tract infections has also been reported (6).

In contrast to the century-old history of probiotics, the prebiotic concept was more recently formally introduced in 1995 by Gibson and Roberfroid. Defined originally as "a nondigestable food ingredient that beneficially affects the host by selectively stimulating the growth and/or activity of one or a limited number of bacteria in the colon, and thus improves host health" (46), the current criteria for prebiotics is now the subject of considerable debate (47, 48). While the general requirements of a prebiotic have been retained in the most recently proposed definition, some key elements of the definition, including specificity and selectively, have been questioned (48). The Canadian Food Inspection Agency (CFIA) has now established its own definition for prebiotics, but neither the U.S. Food and Drug Administration (FDA) nor Europe's European Food Safety Authority (EFSA) have established a definition.

Prebiotics are comprised of simpler molecules such as inulin, fructooligosaccharides, galactooligosaccharides, isomaltooligosaccharides, and mannan oligosaccharides, or more complex compounds such as pectins, resistant starches, arabinoxylan or human milk oligosaccharides (49). Depending on the nature of the prebiotic, these substrates may support the growth of certain members of the gut microbiota, such as bifidobacteria, lactobacilli, or ruminococcus (50). Prebiotics are thought to aid their health benefits through several different mechanisms. These mechanisms include the induction of compositional or metabolic changes to the resident microorganisms, by stimulating the activity and/or growth of health-promoting bacteria, and by the production of SCFA and other end products that reduce the local pH, induce the production of immunomodulatory cytokines, and stimulate mucin production (51). Substrates like FOS and GOS may also have fermentation independent health effects, such as adherence inhibition of pathogens (52).

Systematic meta-analyses showed that, in some cases, a prebiotic treatment reduces fasting insulin levels (53), can aid the treatment of infectious diseases (54) and diarrhea (55), and restore bowel function (56). There are also reports that prebiotics may contribute to abdominal pain, diarrhea, and increased production of gas depending on the doses, nature of the prebiotic, and the susceptibility of the host (6).

#### **1.5 Synbiotic concepts**

When Gibson and Roberfroid first articulated the prebiotic concept more than twenty years ago, they envisioned that prebiotics and probiotics could be combined as synbiotics. Later, Kolida and Gibson (2011) described the two general ways synbiotics could enhance the effects of their parts. Complementary synbiotics are those that contain probiotics and prebiotics chosen independently of one another, with each responsible for a particular effect or health benefit. Accordingly, the best case scenario for such a synbiotic would be that each constituent, i.e., the probiotic and prebiotic, would have a beneficial effect and that the effects would be additive. For example, the prebiotic would stimulate resident strains of bifidobacteria (presumably stains that improved intestinal health), while the probiotic would be established independently, providing an additional health benefit. In this complementary approach, the prebiotic component is not necessarily preferentially fermented by the probiotic strain and could theoretically support other members of the gastrointestinal microbiota. The probiotic strain would gain no ecological advantage by being combined with the prebiotic, and indeed, may not be capable of fermenting the substrate at all.

When the prebiotic is introduced together with a probiotic that cannot ferment the substrate, the outcome may be highly unpredictable and would likely depend on the composition of an individual's gut microbiota. Already it is apparent from human clinical studies that a bifidogenic response or other changes in the microbiota following prebiotic supplementation occur in some subjects but not in others. The nature of the responder/nonresponder phenotype (i.e., what makes a responder a responder) remains the subject of considerable interest. Davis et al. suggested that specific bacterial strains capable of fermenting the prebiotic or competing well in the colon might be absent in the nonresponder population (29), but that has not been established yet. The inability of members of the gut microbiota to compete in the highly competitive gastrointestinal environment could also affect the responder status of subjects. Indeed, Davis et al. showed that even a high abundance of taxa that would be expected to ferment a given prebiotic substrate was not a reliable predictor of whether or not the prebiotic was fermented (29). Salonen et al. have shown that obese male individuals on a resistant starch diet could be divided into responders and non-responders based on the shifts in the composition of their gut In this case a high microbial diversity correlated with a low dietary microbiota. responsiveness (57). Similarly Martinez et al. have reported a microbial responder and

non-responder phenotype in normal weight human subjects who had consumed resistant starches (58). Kovatcheva et al. divided their study cohort in responder versus nonresponders based on the metabolic response to a dietary fiber treatment (59). A subsequent analysis of the gut microbiota of both groups showed that the *Prevotella* and *Bacteroides* ratio was significantly higher in the responder group. Due to this responder and nonresponder phenomenon, a prediction of whether or not a subject will have a health beneficial effect by a dietary treatment is difficult to establish. The response to the treatment depends not only on the functional and taxonomic composition of the gut microbiota but possibly also on host factors. These factors include the amount of digestive enzymes provided by the host, the food transit time, and other potential environmental constraints, which could limit the increase of certain bacterial numbers, even if the growth substrate is provided (58). While these host factors may be limiting the success of dietary interventions to a certain degree, synergistic synbiotics may provide the functional and taxonomic microorganisms that are not present in non-responders. In contrast to complementary synbiotics, synergistic synbiotics consist of a prebiotic substrate that specifically supports the growth of a cognate probiotic strain in the gastrointestinal ecosystem (32). Assuming the target strain reaches the colon, this approach potentially addresses the responder/non-responder problem by providing the strain and its growth substrate *in situ*. The presence of an autochthonous member of the gut microbiota capable of fermenting the prebiotic prior to the treatment is not necessary. However, the synergistic synbiotics are not limited to addressing the responder/ non-responder phenomenon. One important limitation of many probiotic and synbiotic applications is the low ecological success of the probiotic strain (60). In order to become established in the GI tract, the probiotic must not only secure nutrients and other growth factors, but must also outcompete the resident microbiota for these resources. By providing the probiotic partner organism with a new resource opportunity, in this case a selectively fermentable prebiotic, the probiotic strain's competitive fitness can be significantly increased and its persistence enhanced (32).

A literature search has shown that most of the synbiotic combinations used in clinical studies and reported in the literature have not been synergistically supporting the probiotic strain. Indeed, only a few studies have provided evidence that a prebiotic supports its probiotic counterpart in vivo (Table 1.1), and only one such study was conducted with human subjects. There are a variety of reasons why it is difficult to achieve synergism between a probiotic and a prebiotic *in vivo*. Most importantly, the synbiotics used in these studies have not been rationally designed, and have instead been formulated on rather arbitrary bases, such as shelf life, cost, and industrial performance (60), on the basis of availability, cost, or other marketing reasons. So often probiotic strains do not utilize the respective prebiotic. Even when *in vitro* or *in situ* screenings of synbiotic combinations are applied, these techniques do not account for the ecological factors that will affect the probiotic strain *in vivo*, nor do they account for how other autochthonous members of the gut microbiota may benefit from the prebiotic substrate (29, 61, 62). It can be challenging to identify a prebiotic that will specifically and selectively enhance the probiotic strain of interest. New strategies to develop synergistic synbiotic combinations now include *in vivo* selected synbiotic combinations or Multi-taxon Insertion Sequencing, which have been recently discussed (49).

Authors/ Year	Probiotic component	Prebiotic component	Study subject s	Increase of probiotic abundance	P value	Health outcome
Tanaka et al., 1983 (63)	Bifidobac- terium breve 4006	Transgalac- tosylated oligo- saccharide (TOS)	Healthy adults	Pro: 9-10.2 log/g feces; Syn: 10-10.5 log/g feces§	0.05	Not measured
Wang et al., 1999 (64)	<i>Bifidobac-</i> <i>terium</i> Lafti <sup>™</sup> 8B.	Amylomaize	BALB/ c mice	Pro: 4.3% recovery rate; Syn: 27.92% recovery rate in feces	0.05	Not measured
Femia et al., 2002 (65)	Lactobacillus rhamnosus LGG + Bifidobacteriu m lactis Bb12	Oligofruc- tose enriched inulin	Male F344 rats	LGG: Pro: $4.8 \pm 3.4 \cdot 10^5$ ; Syn: $21.1 \pm 18 \cdot 10^5$ CFU/g of feces; Bb12: Pro: $6.1 \pm 8.1 \cdot 10^5$ ; Syn: $8.4 \pm 12 \cdot 10^5$ CFU/g of feces	Not given	Anti-tumoric activity in azoxymethane induced cancer
Ogawa et al., 2005, 2006 (66, 67)	Lactobacillus casei subsp. casei JCM 1134 <sup>T</sup> (Lcc)	Dextran	BALB/ c mice	Pro: 1·10 <sup>4</sup> CFU/mg of feces; Syn: 1.4·10 <sup>6</sup> CFU/mg of feces§	0.01	Significantly elevated natural killer cell activity in spleen mononuclear cells
Krumbeck et al., 2015 (60)	Bifidobac- terium adolescentis IVS-1	Galacto- oligosac- charide (GOS)	Male Sprague - Dawley rats	Pro: $7.9 \pm 0.1 \log 10$ cells/ g colon content; Syn: $9.47 \pm 0.2 \log 10$ cells/ g colon content	0.0001	None

Table 1.1. Synergistic synbiotics reported in the literature

Pro: Probiotic; Syn: Synbiotic

§: Absolute microbial numbers are not given in the original publication and are estimated by the author by careful evaluation of graphs in the original publication

That there are few reports describing the use of synergistic synbiotics in clinical trials is somewhat surprising considering the many publications on synbiotics. Just in 2015 alone, more than 90 publications on synbiotics were published (Figure 1.1). Most of these studies included lactobacilli and bifidobacteria as the probiotic component, and various oligosaccharides, inulin or dietary fibers as the prebiotic component (68).



Fig. 1.1. Number of publications on the topic "synbiotic" over the past 15 years.

#### 1.6 Synbiotics and their outcome on human health in clinical studies

The reported literature on synbiotics includes studies from a wide-range of subject cohorts. Research subjects have included humans of all ages, as well as companion animals (mainly dogs and cats), and food production animals, such as chicken, cows, pigs, cattle, rabbits and fish (69–77). Rodent animal models have also been widely used (60, 78–80). In this review, we focus on human clinical studies and the health claims made for synbiotic combinations to improve human health.

It is important to note that despite the many health claims made for synbiotic combinations in the literature and in the commercial market (Figure 1.2), no claims have actually been approved by regulatory agencies in the U.S. and Europe.



Fig. 1.2. Health claims made for synbiotics in human populations.

Nonetheless, several meta-analyses and systematic reviews suggest that synbiotic treatments may provide beneficial health effects (Table 1.2, and 1.3).

### Table 1.2. Overview of published meta-analyses on synbiotic treatments (adapted from

Krumbeck et al., 2016 and updated).

Author/ Year	Disease phenotype	Studies and subjects included	P value	Type of synbiotic	Study subjects	Outcome		
Shukla et al., Hepatic 2011 Encephalopat (68) hy (HE)		1 trial (n=55)		Probiotic: PP, LM, LPSP, LP2; Prebiotic: BG, I, P, RS		Syn. use reduced		
	pat 0.004 Probiotic: F l trial (n=60) Prebiotic: F		Probiotic: BL; Prebiotic: FOS, vitamins B1, B2, B6, B12	HE patients	the risk of no improvement of Minimal HE.			
Ford et al.,	IBS and	2 trials (n=198)	0.09	Probiotic: BL, BB, LR, LA, LB,ST , LC; Prebiotic: FOS		No reduced symptoms.		
2014chronicidiopathic(81)constipation	idiopathic	2 trials (n=160)	0.003	Probiotic: BL2, LP, LR, LA; Prebiotic: FOS	IBS patients	Beneficial for chronic idiopathic constipation treatment.		
al., 2013 outcome at elective		8 trials (n=361)	0.002	Probiotic: LC, LP2, LP, LM, LA, LB, BL2, ST, PP, BB, EF, CB, BM, LS, BB2, LL; Prebiotic: OAF, OF, BG, I, P, RS, GOS	Patients undergoing elective surgery			The incidence of postoperative sepsis was reduced by syn.
	Clinical outcome after elective	4 trials (n=135)	0.03	Probiotic: PP, LM, LP2; Prebiotic: BG, I, P, RS		Syn. reduced the length of postoperative antibiotic use.		
	surgery	2 or 3 trails each (n between 198 and 260)	>0.05	Probiotic: LC, LP2, LP, LM, LA, LB, BL2, ST, PP, BB, EF, CB, BM, LS, BB2, LL; Prebiotic: OAF, OF, BG, I, P, RS, GOS		No significant changes observed for prevention of pneumonia, wound infection, urinary tract infection, mortality and length of hospital stay.		
Beserra et insul al., 2015 concentr (53) s and h		2 trials (n=364)	0.04	Probiotic: BL, LC, LR, ST, BB, LA, LB; Prebiotic: FOS	Adults with overweight or obesity	Reduced plasma fasting insulin concentrations.		
	Glycaemia, insulin concentration	3 trials (n=260)	<0.05	Probiotic: LS2, BL, LA, BB; Prebiotic: I, FOS		Reduced plasma triglyceride concentrations.		
	s and lipid parameters	2, 3 or 4 trails each (n between 49 and 104)	Not given	Probiotics: LC, LR, ST, BB, LA, BL, LB, LS2; Prebiotic: I, FOS		No significant changes were observed for total cholesterol, LDL- c, HDL-c and fasting glucose.		

Mugambi et al., 2012 (77)	Growth and stool frequency	2 trials (n=227) 2 trials	0.29	Probiotic: BL, LR, LP; Prebiotic: GOS, ScFOS Probiotic: BL; Probiotic: BL;	Infants	Syn. failed to improve growth rate, but significantly improved stool
Chang et al., 2016 (83)		(n=122) 6 trials	0.03	Prebiotic: GOS, FOS Probiotic: LR,BL2, LA, BB, LC, ST, BI,		frequency. Syn. support the treatment of atopic dermatitis, particularly mixed strains of bacteria are used.
	Atopic dermatitis (AD)	(n=369)	0.048	LB, LS; Prebiotic: FOS, lcFOS, GOS, scGOS, starch	Children	Syn. support the treatment of atopic dermatitis in children older than 1 year.
		2 trials (n=1320)	0.26	Probiotic: BL, BB, LR, PF; Prebiotic: GOS, ScFOS		Syn. do not support prevention of AD.
Sawas et al., 2015 (84)	Prevention of infections after liver transplant	4 trials (n=246)	<0.001	Probiotic: BB, BL2, LP2, PP, LPSP, LM, LA, LC, LR, LB2; Prebiotic: GOS, fiber	Adult patients receiving a liver transplant	Syn. reduced infection rate of urinary tract and intra-abdominal infections. Syn. reduced hospital stay and duration of antibiotic use.
Yang et al., 2016 (85)	Prevention of infections after GI surgery	16 trials (n=1,370)	Not given	Probiotic: LC, ST, BB, LA, BL2, LB, LP2, PP, LM, LP, LS2, BM, CB; Prebiotic: FOS, GOS, OAF, OF, BG, I, P, RS	Patients undergoing GI surgery	Subgroup analysis of synbiotic trials showed no health benefits due to synbiotics

#### Table 1.2 continued

Probiotic type: BB= Bifidobacterium breve; BB2= Bifidobacterium bifidum; BI= Bifidobacterium infantis; BL= Bifidobacterium longum; BL2= Bifidobacterium lactis; BM= Bacillus mesentericus; CB= Clostridium butyricum; EF= Enterococcus faecium; LA= Lactobacillus acidophilus; LA2= Lactobacillus affinolactis; LB= Lactobacillus bulgaricus; LB2= Lactobacillus brevis; LC= Lactobacillus casei; LL= Lactococcus lactis; LM= Leuconostoc mesenteroides; LP= Lactobacillus paracasei; LP2= Lactobacillus plantarum; LPA= Lactobacillus paracasei; LPSP= Lactobacillus paracasei; subsp. paracasei; LR= Lactobacillus rhamnosus; LS= Lactobacillus salivarius; LS2= Lactobacillus sporogenes; PF= Propionibacterium freudenreichii; PP= Pediococcus pentosaceus; ST= Streptococcus thermophilus

Prebiotic type:  $BG=\beta$ -glucan; GOS= galacto-oligosaccharides; I= inulin; P= pectin; OF= oligofructose; OAF= oat fiber; RS= resistant starch; FOS= fructo-oligosaccharides; Sc= short chain; Syn= Synbiotic

Author/ Year	Disease phenotype	Studies and subjects included	<i>P</i> value	Type of synbiotic	Study subjects	Outcome
	Ulcerative colitis (UC) maintenance	1 trial (n=120) &	0.03	Probiotic: BL; Prebiotic: Psyllium		IBDQ score: improved quality of life.
Ghouri et al., 2014	Ulcerative colitis induction and maintenance	1 trial (n=41) &	0.05	Probiotic: BB; Prebiotic: GOS	UC patients	Improvement of endoscopic grading compared to standard therapy group.
and Saez-	Ulcerative	1	0.06			Sigmoidoscopy score not improved.
Lara et al., 2015 (86, 87)	colitis induction	1 trial (n=18) *#&	0.05	Probiotic: BL; Prebiotic: Synergy I (I+OF mix)		Inflammatory markers improved.
	Management of Crohn's Disease (CD)	1 trial (n=35) &	0.02	Probiotic: BL; Prebiotic: Synergy I (I+OF mix)		Improved clinical response compared to placebo.
		1 trial (n=24) *&	>0.05	Synbiotic 2000: Probiotic: LA2, PP, LP2, LPSP; Prebiotic: BG, I, P, RS	CD patients	No improvement in endoscopic, clinical and laboratory parameters.
Saez- Lara et al., 2015 (87)	Ulcerative colitis (UC)	1 trial (n=10)	Not given	Synbiotic therapy; Probiotic: BB, BL, LC; Prebiotic: Psyllium	Patients with active UC	Synbiotic was safe and effective.

#### Table 1.3. Overview of systematic reviews on synbiotic treatments.

\*: also discussed in Hedin et al., 2007 (88); #: also discussed in Zigra et al., 2007 (89);

IBDQ: Inflammatory bowel disease questionnaire

Probiotic type: BB= Bifidobacterium breve; BL= Bifidobacterium longum; LA2= Lactobacillus affinolactis; LP2= Lactobacillus plantarum; LPSP= Lactobacillus paracasei subsp. paracasei; PP= Pediococcus pentosaceus.

Prebiotic type: BG= β-glucan; GOS= galacto-oligosaccharides; I= inulin; P= pectin; OF= oligofructose; RS= resistant starch.

However, comparisons between different trials is rather difficult, since studies often vary between the specific probiotics and prebiotics used, their respective doses, the duration of the study, the targeted population, expected and measured effects of treatment. Even the funding source has been suggested to influence outcomes (90). Of particular concern for synbiotic trials is when investigators did not determine the treatment effects independently. In such an approach it is not possible to establish that improvement of a clinical endpoint in the synbiotic treatment group was indeed more beneficial than just the pro- or prebiotic treatments alone. Therefore the synbiotic concept cannot be validated in that case. The majority of clinical trials have chosen such an approach. A literature research of synbiotic trials published within the last 15 years (see below), showed that out of 26 trials presented here, only one provided a prebiotic only, probiotic only, and synbiotic only group (91), and only one trial provided an additional placebo control group (92).

Out of the 26 trials presented here, only one performed a genus specific analysis of the microbiota of the subjects (93), and only three of the trials used a species specific analysis for the applied probiotic (92, 94, 95). All other studies did not conduct any microbial analyses to confirm the survival or activity of the probiotic component at a strain-specific level or even at higher taxonomical levels. Finally, another common limitation of these studies is the lack of experimental power, which may result in overstating or underestimating the actual health benefits of the applied synbiotic. This phenomenon of disadvantages of synbiotic meta-analyses has been previously recognized and criticized for probiotic meta-analyses as well (96).

In general, most of the meta-analyses have focused on the disease phenotype rather than on the exact nature of the treatment. Consequently, these analyses often do not differentiate between prebiotic, probiotic or synbiotic trials, with very few trials specifically using synbiotics. However, systematic meta-analyses specifically analyzing synbiotics are often impossible due the limited number of trials. Other meta-analyses do not distinguish between a probiotic or synbiotic treatment and combine those trials into one analysis, which is not appropriate for assessing the potential health benefits of a synbiotic treatment (Table 1.4).

 
 Table 1.4.
 Overview of Meta-analyses synbiotic treatments that combined proand synbiotic trials into one analyses.

Authors/Year	Disease phenotype	Overall outcome				
Pitsouni et al., 2009 (97)	Patients undergoing abdominal surgery	Pro-/ synbiotic treatment may reduce postoperative infections after abdominal surgery.				
Rossi et al., 2012 (98)	Patients with chronic kidney disease	Limited but supportive evidence for the effectiveness of pre- and probiotics on reducing uremic toxins. No conclusion about synbiotics.				
Zhang et al., 2010 (99)	Patients with acute pancreatitis	Pre-, pro- or synbiotics treatment shows no statistically significant benefit. Safety and efficacy: Use pre- pro- or synbiotics with caution in critically ill patients and patients with severe acute pancreatitis.				
Watkinson et al., 2007 (101)	Patients admitted to adult intensive units	There is currently a lack of evidence to support the use of pre- pro- or synbiotics.				
He et al., 2013 (102)	Patients undergoing colorectal resection for cancer	Pro-/synbiotics administration had a positive effect on the incidence of diarrhea ( $P = 0.001$ ), the incidence of symptomatic intestinal obstructions ( $P = 0.008$ ), the incidence of operative total infections ( $P = 0.0010$ ), and pneumonia infection ( $P = 0.04$ ).				
		Pro-/synbiotics administration increased numbers of <i>Lactobacillus</i> ( <i>P</i> < 0.00001), and decreased the counts of Enterobacteriaceae.				
Dang et al., 2013 (103)	Prevention of eczema	Pro- and synbiotic treatment may reduce incidence of infant eczema. Prebiotics alone have no effect.				
Lytvyn et al., 2015 (104)	Prevention of postoperative infections following abdominal surgery in adults	Probiotics/synbiotics reduce the risk of surgical site infections compared to placebo or standard of care and potentially benefit for urinary tract infections with no increased risk of adverse events, and no occurrence of serious adverse events reported as related to study product.				
Petrof et al., 2012 (105)	Critically ill patients, including burn, multiple trauma, pancreatis, diarrhea patients and general intensive care unit patients	Clinical trials suggest that probiotics patients may reduce overall infection rates in critically ill patients.				
Arumugam et al., 2016 (106)Decrease of postoperative sepsis GI surgical patients		Pro-/synbiotics significantly reduced risk of postoperative sepsis by 38% $(P = < 0.0001)$				

#### Review of selected clinical trials on synbiotic treatments

As introduced earlier, the study design is crucial when synbiotic treatments are being assessed for their health beneficial effects. To justify the application of a synbiotic treatment instead of a probiotic or prebiotic only treatment, appropriate controls must be used to assess if the synbiotic treatment acts synergistically. Thus, controls must include a placebo, the probiotic treatment alone, the prebiotic treatment alone, and the synbiotic combination. Only this study design allows investigators to assess whether or not a synbiotic treatment is more effective than the probiotic and prebiotic treatments alone and whether or not synergy is given. In addition, the survival and/or metabolic activity of the probiotic component should be quantified in a strain-specific manner to ensure survival and establishment of the probiotic. However, this requires a significantly higher number of subjects, and only a small number of trials fulfill these criteria.

In the next sections, recent human trials that applied synbiotics to treat specific clinical disorders are reviewed. These include trials on metabolic syndrome, inflammatory bowel disease, diarrhea, colon cancer, and inflammatory bowel syndrome

#### Metabolic syndrome

Metabolic syndrome refers to a group of metabolic disorders that collectively contribute to heart and other health problems. Risk factors include central obesity, impaired glucose tolerance, dyslipidemia and hypertension (107, 108). This syndrome is associated with obesity, type II diabetes, heart disease, and cancer (109).
Several synbiotic formulations have been used in clinical trials. Eslamparast et al. conducted a prospective, randomized, double-blind, and placebo-controlled study analyzing the effect of 250 mg fructooligosaccharide (FOS) and a probiotic cocktail of seven different strains (Lactobacillus casei, Lactobacillus rhamnosus, Streptococcus thermophilus, Bifidobacterium breve, Lactobacillus acidophilus, Bifidobacterium longum, and *Lactobacillus bulgaricus*) on 38 subjects suffering from metabolic syndrome (110). Subjects were instructed to consume the supplement or placebo (maltodextrin) twice daily for 28 weeks, and were instructed to follow strict dietary recommendations, lower their energy intake, and increase their physical activity. At the end of the study, individuals in the synbiotic treatment had significantly improved levels of insulin resistance, fasting blood sugar, triacylglyceride, and serum high-density lipoprotein levels compared to the placebo treatment. No difference was observed in body mass index (BMI), low-density lipoprotein levels, anthropometric parameters, and energy intake/expenditure. The authors concluded that the synbiotic treatment can increase the efficacy of a dietary therapy in the management of metabolic syndrome and insulin resistance. While the design and analysis of this study were adequate for the study goals, an analysis of the gut microbiota was not included. The authors stated that "because previous studies had shown beneficial effects of VLS#3 and Lactobacillus longum and fructooligosaccharide, demonstrating their beneficial effects on intestinal microbiota, this synbiotic capsule was chosen for the present study as it contained all these strains in addition to others." However, the mentioned "Lactobacillus longum" is not a strain that has ever been described in the literature before and is most likely either a *B. longum* or a different *Lactobacillus* strain. Additionally, the authors claim that VLS#3 was chosen for its previously reported health benefits. But in the study conducted here, the applied synbiotic mixture did not contain all the strains

present in VLS#3 and left, seemingly randomly, two strains, *B. infantis* and *L. delbrueckii*, out. Therefore no comparison to other studies using VLS#3 can be made. Leaving these issues aside, no comparison was made between a pro- and prebiotic treatment alone, thus not confirming that the synbiotic treatment was more effective than the probiotic or prebiotic treatment alone.

Lactobacilli, bifidobacteria and FOS were combined as synbiotic treatments for gastric bypass patients. The synbiotic treatment contained *Lactobacillus paracasei* LPC-37, *Lactobacillus rhamnosus* HN001, *Lactobacillus acidophilus* NCFM, and *Bifidobacterium lactis* HN019 (each at  $10^9$  CFU) and 6 g of FOS (111). Patients were divided into a placebo, prebiotic, and synbiotic treatment group. The supplement was taken for 15 days after which a significant weight loss was seen in the prebiotic group. The BMI was also lower in the prebiotic and placebo group than in the synbiotic group. Other blood and inflammatory markers were not different among the groups, and the fecal microbiota was not addressed. Effects of SCFA production induced by FOS treatment were only indirectly measured by assessing plasma cytokines, and no effect was detected. Although clinical benefits for each of the tested strains are reported by the authors, there was no rationale given by the authors for combining them as a synbiotic. Also, the low number of bypass patients (n = 3 per group) makes the interpretation of this data rather difficult and further investigation will be needed.

The effect of a synbiotic consisting of *Lactobacillus sporogenes*  $(2.7 \times 10^8 \text{ CFU})$ and 1.08 g of inulin was tested in 62 type II diabetic patients in a randomized doubleblinded cross-over controlled clinical trial (112). The study lasted three weeks, and patients were instructed to consume the treatments three times a day. The synbiotic was compared to a placebo, and results suggested the synbiotic led to significantly decreased serum insulin levels (P = 0.03) and serum high-sensitivity C-reactive protein levels (P = 0.01). Increases in plasma total glutathione (P < 0.001) and serum uric acid levels (P = 0.04) were also reported. However, no effects on cholesterol levels were observed, and the fecal microbiota of the patients were not investigated. The same research group tested this synbiotic in combination with beta-carotene, and a reduced amount of *L. sporogenes* and inulin in 102 diabetic patients (113). In this case cholesterol levels were affected significantly in addition to insulin compared to a control. Pro- or prebiotic-alone treatments were not included, making it impossible to determine if the observed effects were due to the synbiotic or the individual synbiotic components. Inulin has been previously shown to have beneficial effects in the treatment of diabetes (114–116). Finally, as noted previously (117), "*Lactobacillus sporogenes*" is not a validly named species and the applied species here is more likely a *Bacillus coagulens*. However, without a classification analysis, the identity of the strain used in this study cannot be established.

The same synbiotic combination of *Lactobacillus sporogenes* ( $40 \times 10^8$  CFU, consumed three times a day) and inulin (2.8 g, consumed three times a day) was tested in 81 diabetic patients (118). The synbiotic, probiotic, or placebo was delivered in form of bread. After eight weeks of treatment, plasma nitric oxide was increased (P = < 0.0001) and malondialdehyde levels were significantly reduced (P = 0.001) compared to both placebo and probiotic only treatments. Since a prebiotic only treatment only was not included it remains unclear if this effect could have been achieved by inulin only. The survival of the probiotic strain during the bread making process or storage was not reported, nor any analysis of the gut microbiota. The actual dose of *L. sporogenes* per day therefore remains speculative.

A synbiotic cocktail containing *Lactobacillus casei, Lactobacillus rhamnosus, Streptococcus thermophilus, Bifidobacterium breve, Lactobacillus acidophilus, Bifidobacterium longum, Lactobacillus bulgaricus* (2 x 10<sup>8</sup> CFU total), and FOS (unknown amount) was tested for its potential to support a weight loss regime (119). Forty-six patients of metabolic syndrome consumed the synbiotic or placebo (maltodextrin) for 12 weeks twice a day. All patients experienced significant weight loss, and the synbiotic treatment reduced the systolic blood pressure (P < 0.05). However, 90% of the subjects in the synbiotic group used medication to reduce blood pressure compared to 75% in the placebo group before and throughout the duration of the study. While this difference was not significant, and could have affected the outcome. The fecal microbiota was not examined and the nature of the study design precludes any conclusions about the efficacy of the prebiotic or the probiotic strains applied here.

A study conducted with patients suffering from type II diabetes claimed a health benefit of a synbiotic combination compared to a placebo (120). However, the value of this study is limited by the fact that no further details of the nature of either the synbiotic or placebo treatment was given.

## Inflammatory bowel disease

Ulcerative colitis (UC) and Crohn's disease (CD), the two types of inflammatory bowel diseases (IBD), are chronic inflammatory pathologies of the gastrointestinal tract. Both conditions occur in individuals who are genetically susceptible and exposed to environmental risk factors (121). Even though the etiology of IBD has been extensively studied, the disease pathogenesis is not fully known, nor is there a cure (122). The characteristics of the inflammation are different, with CD being scattered throughout the GIT, typically involving the distal small intestine and colon with transmural inflammation and occasionally associated with granulomas, whereas in UC the inflammation is usually confined to the mucosa of the colon (123, 124). Both UC and CD are characterized by a relapsing and remitting course leading to a very significant reduction in life quality during the disease (125).

Several synbiotic formulations have been used in clinical studies to treat IBD. In Furrie et al., Synergy 1 (6 g of inulin/oligofructose) and 2 x  $10^{11}$  CFU *Bifidobacterium* longum were administered to UC patients (93). One of the strengths of this study was that the probiotic strain was isolated from a healthy human subject and had been assessed for its aerotolerance, acid tolerance, resistance to bile salt, and adherence to epithelial cells. Its ability to use the prebiotic substrate as an energy source was also established *in vitro*. The organism was further shown to alter the cytokine expression in a HT29 epithelial cell line and reduce proinflammatory cytokine levels *in vitro*. For the clinical study, 18 patients were divided into synbiotic and placebo groups, each receiving the respective treatments twice daily for four weeks. The synbiotic treatment led to reduced inflammation and regeneration of epithelial tissue compared to the placebo group, reduced mRNA levels of human beta defensins, and lowered levels of tumor necrosis factor  $\alpha$  and interleukin 1  $\alpha$ . Although survival of the probiotic strain was not measured in a strain-specific manner, bifidobacteria-specific rRNA levels were increased 42-fold in the synbiotic group compared to approximately 5-fold in the placebo group. Unfortunately, this study did not investigate the effects of the probiotic independently.

Another *B. longum* synbiotic clinical trial was performed by Fujimori et al. In this study, a *B. longum* strain (2 x  $10^9$  CFU) was combined with eight grams of psyllium as the prebiotic component. Subjects were UC outpatients (n = 120) (91). This trial did include probiotic and prebiotic only treatment groups in the study design to allow for comparisons. While most tested blood markers showed no differences among the three treatments, C-reactive protein was significantly decreased (*P* = 0.04) and the total protein level in the blood samples increased (*P* = 0.03) in the synbiotic group. Hemoglobin and hematocrit only increased in the probiotic group (*P* = 0.04). Total inflammatory bowel disease questionnaire scores showed significant improvement only for the synbiotic group. The investigators concluded that the synbiotic treatment led to a greater life-quality than the pre- and probiotic treatments alone. However, this study did not investigate the mechanisms responsible for this improvement nor was the survival of the probiotic determined. Therefore no conclusion about the nature of the synbiotic can be drawn, i.e. if the synbiotic acted synergistic or complementary.

In another clinical study, the effect of a *Bifidobacterium breve*-GOS synbiotic on subjects with mild to moderate UC was assessed (94). Synbiotics contained *B. breve* (10<sup>9</sup> CFU, Yakult) and 5.5 g of galactooligosaccharide. Forty-one patients were treated with a placebo or the synbiotic for one year. End-points included endoscopic scores and myeloperoxidase levels in lavage solutions; both were significantly lower in the synbiotic treated group. An analysis of the fecal microbiota by plate counting was also performed for subjects in the synbiotic group before and after the treatment. Of all assessed microbes, only *Bacteroidaceae* were significantly decreased after the synbiotic treatment. The abundance of *Bifidobacterium* remained the same, and *B. breve* was only detected after the treatment but not before ( $5.75 \pm 1.65 \log_{10}$  CFU/g feces). Therefore, it can be concluded

that *B. breve* survived the passage through the GI tract. Whether or not the applied prebiotic was supporting the probiotic could not be determined from this study. Interestingly, here no bifidogenic effect was observed due to galactooligosaccharide treatment. This is contrary to results previously reported (126).

A combination of 6 g of Synergy 1 (inulin/oligofructose) and 2 x  $10^{11}$  CFU *Bifidobacterium longum* showed improvements in Crohn's disease in patients that continued using their conventional CD medication (95). Compared to a placebo, the synbiotic significantly reduced TNF- $\alpha$  gene expression (P = 0.041), disease activity indexes (P = 0.02), and histological scores (P = 0.018) after six months of treatment. The microbiota of tissue biopsies was analyzed in both species- and genus-specific manner. Interestingly, 8 out of 13 patients had increased numbers of *Bifidobacterium longum* and bifidobacteria at the three month time point compared to the baseline in the synbiotic group, increasing to 11 patients after six months. The nature of this responder/non-responder phenomenon was not addressed.

Chermesh et al. investigated the potential of Synbiotic 2000 to prevent postoperative recurrence of CD (127). This formulation contained  $10^{10}$  CFU *Pediococcus pentoseceus*,  $10^{10}$  CFU *L. raffinolactis*,  $10^{10}$  CFU *L. paracasei* subsp. *paracasei* 19, and  $10^{10}$  CFU *L. plantarum* 2362 and as fermentable fibers 2.5 g of  $\beta$ -glucans, 2.5 g of inulin, 2.5 g of pectin, and 2.5 g of resistant starch. The frequency of the treatment was not stated. Of 30 enrolled patients, nine patients completed the study, which lasted 24 months. Synbiotic 2000 had no effect compared to the placebo on endoscopic or clinical relapse, nor the postoperative occurrence of CD. However, it significantly improved weight increase and normalization of hemoglobin levels at the three month follow up time point. No analysis of the gut microbiota was done, nor were reasons given for the selection of the synbiotic combination. Of the prebiotic components none were correlated to any health benefits by the authors. Only fructooligosaccharides were mentioned as beneficial in general. Unfortunately, this study had a small sample size, lacked a pro- or prebiotic control arm, and used a seemingly arbitrarily selection process to determine the synbiotic combination.

The plant fiber, psyllium, was combined as a synbiotic with bifidobacteria and lactobacilli, and used to treat CD patients (128). The synbiotic contained 9.9 g of psyllium and  $3 \times 10^{10}$  CFU *Bifidobacterium breve*,  $3 \times 10^{10}$  CFU of *Lactobacillus casei*, and  $1.5 \times 10^{10}$  CFU of *Lactobacillus casei*, and 10<sup>10</sup> CFU of *Bifidobacterium longum*. Ten active CD patients were enrolled in this 13 month trial. The trial was not placebo controlled and subjects were allowed to discontinue using psyllium during the trial if abdominal bloating occurred. All subjects were allowed to reduce the pro- and/or prebiotic treatment on their own will, thereby individualizing the treatments. Subjects also received aminosalicylates and prednisolone at varying doses. All subjects continued taking the probiotic treatment, but their doses varied between 12 and 73  $\times 10^{10}$  CFU. Four subjects discontinued the prebiotic treatment, and for the remaining six patients the doses varied between 3.3 and 9.9 g per day. Based on the clinical outcome, the authors divided the subjects into complete responder, partial responder, and nonresponders. Eight complete responders lowered their Crohn's Disease activity index scores by more than 70 points and six achieved remission. Patients who discontinued the synbiotic treatment and those following it through the whole duration of the study were found in the responder as well as the non-responder group. Therefore, no correlation between a synbiotic treatment and a health improvement could be determined.

Rossi et al. conducted a observational study applying SynGut<sup>TM</sup> (*Bifidobacterium lactis* W51, *Lactobacillus acidophilus* W22, *Lactobacillus plantarum* W21, *Lactococcus lactis* W19, and inulin) to 96 IBS patients for two months (129). While this study had no control group, no pro- or prebiotic group, and no standardized scoring system, it reported an improvement of IBS symptoms in 74% of the participants.

#### Diarrhea

Diarrheal diseases are often caused by infectious agents, which lead to lose, or liquid, bowel movements with increased frequency, water content, and volume. Worldwide, diarrhea is the leading cause of hospitalizations, morbidity, and mortality (130).

The potential of *Lactobacillus acidophilus*, *Lactobacillus rhamnosus*, *Bifidobacterium bifidum*, *Bifidobacterium longum*, *Enterococcus faecium* (2.5 x 10<sup>9</sup> CFU total), combined with 625 mg of fructooligosaccharide as a synbiotic, was tested in children with acute diarrhea (130). Treatment with oral rehydration salts (ORS) and intravenous therapy was also provided. Compared to a control group (receiving only ORS and/or intravenous therapy), the synbiotic combination significantly shortened the duration of diarrhea (P < 0.0001) and shortened the hospital stay (P = 0.002). The gut microbiota of these children was not analyzed.

In a similar study, children with acute rotavirus diarrhea were treated with *Lactobacillus* sp., *Streptococcus* sp., *Bifidobacterium* sp. (1 x  $10^9$  CFU total) and 990 mg of FOS (131). A standard fluid therapy and nutritional support were provided. A total of

35 children were enrolled in the synbiotic group and were compared to a placebo group. The duration of diarrhea was significantly shorter in the synbiotic group (P < 0.0001), and for half of the patients receiving a synbiotic treatment, intestinal mucosal healing was reported 50 hours after the synbiotic administration. The gut microbiota was not analyzed. No further descriptions were given to explain why half of the group responded to the treatment.

An arabinogalactan and xylooligosaccharide mixture was used to formulate a synbiotic that also included *Lactobacillus paracasei* B21060 (2.5 x  $10^9$  CFU). The prebiotics were present at 500 mg and 700 mg, respectively. Subjects included 55 children with acute diarrhea, who also received ORS treatment. The synbiotic showed a significantly higher resolution rate (P = 0.005) than the placebo group after the first 72 hours (132). This study allowed for additional treatments (e.g., diosmectite, domperidone or racecadotril) given by the parents after the first 72 hours, which may have influenced the total duration of diarrhea. No analysis of the fecal microbiota was performed. The additional treatments may have influenced the efficacy of the synbiotic treatment.

A combination of *Streptococcus thermophilus*, *Lactobacillus bulgaricus* (9.7 x  $10^8$  CFU), *Bifidobacterium animalis* subspecies *lactis* (Bb-12) (5 x  $10^9$  CFU), and 1g of inulin was assessed for its potential to prevent diarrhea, vomiting and other infections in young children (133). One hundred and forty-nine children participated in this placebo controlled double-blind study. After 16 weeks of treatment, synbiotic treated children had significantly fewer days of fewer, but significantly more days with watery stools (*P* < 0.05). No analysis of the microbiota was performed and it remains unknown which components of the treatment might have caused this effect.

To assess the potential to prevent and/or shorten the occurrence of traveler's diarrhea, 196 healthy adults received a synbiotic combination named Agri-King Synbiotic (AKSB). This preparation contained fructooligosaccharide, 4.5 x 10<sup>9</sup> CFU Enterococcus faecium SF68 and 5 x 10<sup>9</sup> CFU Saccharomyces cerevisiae CNCM I 4444 consumed twice daily (134). This paper does not state the amount of fructooligosaccharide present in the AKSB capsules. A literature research suggests that this preparation contains 115 mg of fructooligosaccharide per capsule (135). A phase I study had shown that AKSB was safe and also that both strains were washed out within seven days after the treatment was discontinued. Study subjects traveling to Asia, Africa, South and Central America were instructed to consume the synbiotic or placebo treatment one to two times daily, and to continue the treatment if diarrhea should occur. Approximately half of the study cohort experienced traveler's diarrhea, but no benefit of the synbiotic treatment was detected. To justify the combination of E. faecium and S. cerevisiae the authors stated that E. faecium was capable of competing with other Gram-negative bacteria. However, the authors do not show evidence for this here and instead refer to another paper that analyzed this phenomenon in stored meat samples and not for the gut microbiota (136). No further comments were made regarding S. cerevisiae or fructooligosaccharide.

## Colon cancer

Colorectal cancer is the third most common form of cancer and has a very high mortality. In addition to genetic factors, environmental factors including radiation, chemical carcinogens, and diet contribute to tumorigenesis in the colon (137). Current treatments are associated with a high risk of complications and a low success rate.

Investigators have suggested that by maintaining a healthy weight, diet, and physical activity, up to one third of colon cancers may be prevented (138). Numerous pro-, pre-, and synbiotic studies using rodent models suggest that these treatments may have preventive and therapeutic properties. However, human studies are difficult to perform and therefore rare.

Rafter et al. and Roller et al. assessed a combination of Lactobacillus rhamnosus GG (1 x 10<sup>10</sup> CFU), *Bifidobacterium lactis* Bb12 (1 x 10<sup>10</sup> CFU) and 12 or 10 g of Synergy 1 on colon cancer and polypectomized patients in two similar phase II anti-cancer studies (139, 140). The synbiotic treatment was compared to a placebo in a 12 week trial. Fecal water obtained from the cancer patients did not improve barrier function in Caco-2 cells, but did increase production of interferon  $\gamma$ . For polypectomized patients, several benefits were observed among the synbiotic group, including significant decreased DNA damage in colonic mucosa, reduced proliferation, and decreased secretion of IL-2. The fecal water improved barrier function in Caco-2 cells and significantly reduced necrosis in HCT116 cells. The investigators also assessed survival of each of the probiotic strains in an independent study with healthy human subjects who consumed rifampicin resistant mutants of each strain (139). Rifampicin resistance was used to identify the probiotics in a strain specific manner in the fecal samples. A full recovery of both strains in healthy subjects was reported. For the study patients, a fecal analysis was not performed at the strain level, but at the genus level. The number of Lactobacillus and Bifidobacterium consistently increased in the synbiotic group for both cancer and polypectomized patients over the 12 week trial, while *Clostridium* numbers decreased. Since a probiotic only treatment was not applied, a synergy between the probiotic strains and Synergy 1 could not be confirmed based on this data. Roller et al. did not analyze the fecal microbiota, but referred to another study that reported that only 10 % of the consumed amount of LGG and Bb12 survived the GI tract in the same synbiotic treatment (141). Neither study analyzed the probiotic and prebiotic components alone.

In a four week cross-over trial the effect of the synbiotic combination of 12.5 g of resistant starch and 5 x 10<sup>9</sup> CFU *Bifidobacterium lactis* was investigated on twenty healthy subjects (92). Even though these patients were healthy, the effect of the dietary treatments on markers of early colorectal carcinogenesis was assessed. A placebo, a prebiotic only, and a probiotic only arm were also included. Full analyses of the fecal microbiota were conducted using DGGE and quantitative real-time PCR to assess levels of Bifidobacterium The DGGE banding patterns showed that the synbiotic treatment introduced lactis. significantly more changes to the gut microbiota than the placebo or the pro- or prebiotic treatments alone. Interestingly the probiotic treatment led to higher numbers of B. lactis than the synbiotic treatment (8.8 x  $10^7$  versus 5.4 x  $10^7$  B. lactis/g feces). Therefore a synergistic relationship between this strain of *B. lactis* and the applied resistant starch is not likely. No differences were detected for the SCFA profile, fecal ammonia or pH, serum inflammatory markers, or epithelial variables among the treatments. This study demonstrates how even significant changes introduced to the gut microbiota by a dietary intervention do not necessarily lead to a change in disease associated phenotypes.

## Irritable bowel syndrome

Irritable bowel syndrome (IBS) is an intestinal disorder characterized by abdominal pain, bloating, diarrhea, alternate constipation, distention, or a combination of these symptoms. The cause of this illness has not been established, but visceral hypersensitivity, genetics, the gut microbiota, constant low-grade inflammation, and environment are contributing factors (81). Approximately 11 % of the world's population may be affected by IBS, with higher occurrences among women and younger individuals (142, 143). Physiological interventions, dietary manipulations, pharmacologic agents, and modulation of the gut microbiota are part of current treatments for IBS (144).

In one double-blinded, randomized and placebo-controlled study, a synbiotic mixture of  $5 \times 10^9$  *Lactobacillus plantarum*,  $2 \times 10^9$  *Lactobacillus casei* subp. *rhamnosus*,  $2 \times 10^9$  *Lactobacillus gasseri*,  $1 \times 10^9$  *Bifidobacterium infantis*,  $1 \times 10^9$  *Bifidobacterium longum*,  $1 \times 10^9$  *Lactobacillus acidophilus*,  $1 \times 10^9$  *Lactobacillus salivarus*,  $1 \times 10^9$  *Lactobacillus sporogenes*, and  $5 \times 10^9$  *Streptococcus thermophilus* in combination with 2.2 g Synergy 1 (inulin/oligofructose) was tested for its potential to reduce symptoms of IBS (145). Sixty-four patients were enrolled and treated for four weeks. No overall satisfactory relief was achieved with the synbiotic treatment. However the synbiotic did improve quality of life scores, and the severity of flatulence was significantly decreased. Interestingly, the authors provided a rationale for selecting this particular synbiotic combination. Namely, the product is readily available, has a history of safe use, and there was only one other study using single-strain synbiotic mixture for the treatment of IBS. Nonetheless, effect on the gut microbiota was not studied.

A recent study examined the effect of *Lactobacillus acidophilus* (1.8 x  $10^7$  CFU/g), *Bifidobacterium animalis* subsp. *lactis* Bb-12 (2.5 x  $10^7$  CFU/g), and Beneo dietary fibers (2%) on the quality of life and IBS symptoms of 76 constipation-predominant IBS patients (146). The synbiotic was delivered twice daily in 180 g of fermented milk for four weeks. Several markers of IBS symptoms improved after four weeks, but there was no difference between the synbiotic and the placebo (fermented milk).

Bittner et al. tested the efficacy of the synbiotic Prescript-Assit<sup>TM</sup> in a two week randomized, placebo-controlled study, followed by a two week open label treatment and a follow up 60 weeks later (147, 148). Prescript-Assit<sup>TM</sup> is a combination of 29 soil-based microorganisms, including several Anthrobacter, Bacillus, Brevibacterium, Pseudomonas, and Streptomyces strains. The prebiotic components are not well defined, except that one of them is leonardite. A total of 25 patients completed the first two week study, and 22 completed the 60 week follow up. The authors concluded that Prescript-Assit<sup>TM</sup> was capable of reducing short-term and long-term symptoms of IBS in the study cohort. These symptoms included general ill feeling/nausea, indigestion/flatulence, and colitis. Since the synbiotic composition was not clearly defined, an assessment of the synergy of this synbiotic cannot be made. The fecal microbiota of the patients was not analyzed and therefore no conclusions regarding the mode of action of this synbiotic can be inferred. Interestingly, the applied organisms are mostly found in soil are not considered members of the human gut microbiota. Whether or not these organisms are capable of reaching the colon or interacting with the autochthonous microbiota remains unknown. Unfortunately, a rationale for choosing soil organisms in a human trial is not given.

The efficacy of Flortec, a synbiotic combination containing 5 x  $10^9$  CFU *Lactobacillus paracasei* B21060, xylo-oligosaccharides, glutamine, and arabinogalactone, was tested in a parallel-arm, double blind study in patients of IBS (149). Patients were instructed to consume the synbiotic or prebiotic treatment twice a day for twelve weeks and were asked to report GI symptoms on a daily basis. No placebo control or probiotic

only arm was included. A total of 167 IBS patients were enrolled in this study, however, 55 discontinued the treatment prematurely. The main reason to withdraw was a perceived lack of benefit (33 % of the cases). For the remaining patients, no differences were observed between the prebiotic and the probiotic treatment groups. Compared to the baseline, both treatments led to a decrease in IBS scores after one week. There was, however, a significant difference among the number of patients that had at least one bowel movement per day with 70 % in the synbiotic group and 35 % in the prebiotic group. An analysis to assess how a responder differed from a non-responder was not conducted. Subsequently, 47 patients who had a diarrhea predominance were analyzed as a subgroup. The number of bowel movements and IBS score significantly decreased due to the synbiotic treatment compared to the baseline and the prebiotic group. In this study, patients were allowed to use a "rescue treatment". Approximately 8 % of the study subjects used such a treatment, which was not closer defined then "medications effecting gastrointestinal motility and/or perception". These additional treatments were not taken into account during the assessment of the syn- and prebiotic treatments. Collectively, the absence of both a placebo control arm and an analysis of the gut microbiota makes it difficult to establish the effectiveness of the synbiotic.

Dughera et al. conducted an open-label, uncontrolled, and multi-center study in ten Italian gastroenterological centers (150). The applied synbiotic was Zir Fos®, containing  $5 \ge 10^9$  CFU of *Bifidobacterium longum* W11 and 2.5 g of Fos-Actilight, a short-chain fructooligosaccharide. A total of 129 patients with constipation-predominant IBS were enrolled and received the treatment for three months. Measured symptoms included abdominal pain, bloating, well-being, stool shape, stool frequency, concomitant treatments and comorbidity. The results showed that the synbiotic treatment significantly reduced abdominal pain and bloating, and induced a higher stool frequency (P < 0.0001).

This same synbiotic (i.e., *Bifidobacterium longum* W11 and Fos-Actilight) was tested on 636 patients with constipation-variant IBS (151). A clear statement about duration of the treatment was not given, only that the treatment lasted for at least 36 days. The dose was the same as in the study above (150), and the treatment effects were evaluated at the end of the study and at a one month follow up visit. The results are similar to those reported by Dughera et al., in that the treatment resulted in significantly decreased bloating and abdominal pain (P < 0.0001) and increased stool frequency significantly. Most (~84%) of the patients reported improved symptoms at the end of the treatment, but at the follow-up evaluation the number decreased to 63 %.

In neither trial were the fecal microbiota analyzed nor were control groups with probiotic, prebiotic, or placebo treatment included. Based on these studies no conclusion can be drawn about a synergy between the probiotic and prebiotic.

### The ideal human trial to assess health benefits of a synbiotic treatment

The development and selection of successful synbiotic combinations is a very complex issue (32). As noted above and previously recommended by Kolida and Gibson (32), synbiotic trials have to be carefully designed and controlled in order to demonstrate the additive effect of each component, and to assess the minimum effective dose of each component of the synbiotic in order to achieve the desired health benefit, while avoiding side effects. As demonstrated here and previously (32), most studies did not provide the

necessary controls to confirm an additive or synergistic effect of the synbiotic. Importantly, the rationale for how the synbiotic had been formulated is rarely stated. In contrast, an ideal clinical trial would include synergistic synbiotics that had been shown to survive passage through the GI tract and also had an ability to become established in the GI environment. Considering ecological criteria is also important when formulating synbiotic combinations, including demonstrating that the probiotic strain is capable of metabolizing the given prebiotic under competitive conditions. Changes introduced to the gut microbiota should also be assessed to determine if cross feeding or other ecological events had occurred, e.g. niche competition, niche partitioning, or niche exclusion with the resident microbiota. It is also critical to validate that the probiotic had been enriched using strain-specific probes or primers in quantitative PCR assays. The experimental design should include treatments consisting of each component of the synbiotic. Once the characteristics of the pro- and prebiotic are established independently as well as in combinations and an additive effect of the components has been demonstrated, randomized, controlled and double-blinded human trials should be conducted, with prebiotic, probiotic, and synbiotic treatments and a placebo control. Survival of the test strains and changes in the composition of the gut microbiota should be assessed, in addition to measuring the health or clinical biomarkers of interest. The study cohort needs to be sufficiently large to assure adequate power for the statistical analysis. Ideally, the synbiotic should be compared to another a similar synbiotic containing a different strain.

#### 1.7 Commercial synbiotics: recent developments and future prospects

The functional foods market, that includes gastrointestinal health products, is estimated to be worth more than US \$ 43 billion. Currently, the US, Europe, and Japan represent 90 % of the global functional food market (152). Within the US market, probiotic products had an estimated revenue of US \$ 3.4 billion in 2013, and already in 2014 sales of probiotic products were the fasted growing of all supplements with a 22 % increase and US \$ 10 billion revenue (153). It is therefore the most popular functional food ingredient after minerals and vitamins (154). The market revenue for prebiotics is estimated at US \$ 334 million, and synbiotics at 69 million. These numbers are expected to increase with the new markets in the Middle East, China, India and New Zealand (35).

Despite the substantial market opportunities for these products, researchers, clinicians, and regulatory agencies continue to emphasize several important issues. In particular, demonstrating that products are safe and effective remains a top priority. Manufacturers are especially motivated to develop appropriate health claim strategies (155). This situation is complicated by different health claim regulations that vary from one country to another and the degree of evidence required to support a health claim (156). Currently, neither EFSA nor the FDA have approved any health claims made for pre-, pro- or synbiotic combinations. Moreover, probiotic and synbiotic products must be distinguished between a pharmaceutical product and a food product (157). The FDA guidelines state that if any agent, including probiotics, is ingested for the purpose of curing, mitigating, treating, diagnosing or preventing disease, it is classified as a "drug" and must undergo the regulatory process similar to any new pharmaceutical. This can be a burdensome process for probiotic/synbiotic foods, especially since the majority of food

products are not produced in pharma grade facilities and the FDA may require phase I safety studies for new synbiotic products. In Europe the European Nutrition and Health Claims regulations intend to: (i) ensure that claims are "clear, accurate and based on scientific evidence"; and (ii) prohibit foods that bear "claims that could mislead consumers". Ultimately, high-quality human intervention studies are necessary to support any health claims of a product (156).

Another obstacle for the food industry is the lack of consumer understanding of these products. Although consumers apparently understand probiotics, at least in general, they are less familiar with prebiotics (or confuse it with probiotics), and even fewer understand synbiotics (35, 47). Therefore some products are not even marketed as synbiotics, but rather as probiotics. Nonetheless many synbiotic products are on the market. To date, most synbiotic-containing foods claim to improve general gut health or the body's natural defense mechanisms by supporting the immune system or lowering blood cholesterol.

Synbiotic products are most commonly presented to consumers as powders or cultured dairy products like yogurts or smoothies. Prebiotics can be used in most food applications, but the environmental sensitivity of probiotics limits their practical use in non-refrigerated foods, since the survivability of the probiotic strains dictates which synbiotics can be developed (35). However, new microencapsulation technologies that protect the bacteria against otherwise detrimental processing treatments could lead to a variety of new synbiotic products, including desserts, candy, juices, cheeses, or chocolate (158–162). Interestingly many pre- and synbiotic products contain rather small amounts of the prebiotic component (on a per serving basis), which may be too low to induce a

health effect. Low doses are used, in part, to avoid adverse gastrointestinal complaints (163), but perhaps also for cost reasons.

Interestingly, despite the new products introduced into the marketplace and questions concerning safety and efficacy, synbiotic foods have a long history of save consumption. As reviewed by Ashwani et al., many indigenous synbiotic foods can be found around the world, including Central America, India, Eastern Europe, China and Africa (164). Most of these synbiotic preparations are fermented beverages, either with defined starter cultures, or by "spontaneous fermentation" (165). However, the uncertainty of the actual microbial composition likely results in inconsistent final food products, unsuitable for industrial sale. Nonetheless, these indigenous synbiotic foods have market potential if produced on an industrial scale and with appropriate quality standards (164).

# 1.8 Remaining questions and specific aims

Disturbances of the microbial composition in the gastrointestinal tract have been associated with deterioration of host health and functions. These developments may either be directly induced by the gut microbiota, via an altered metabolite synthesis, or via the host immune system (6). While the microbiota composition in the human gastrointestinal tract is remarkably stable, it can be successfully modulated by certain synbiotic treatments. These treatments could offer great advantages to human health when selected on a rational basis. However, there are currently many potential limitations that hinder the development of synergistic synbiotic formulations. There is a need to validate the potential health benefits of synbiotics in carefully controlled human clinical trials. Additionally the future development of new synbiotic combinations should focus on the development of synergistic synbiotics that prioritize the ecological properties and requirements of the probiotic strain.

The work presented in this thesis aims to answer three important questions, which are relevant for our understanding of how the gut microbiota is shaped and how we can establish successful concepts for the modulation of the gut microbiota.

The first objective of this thesis was to gain insight into the complex interplay between the host and intestinal microbes. For this approach we chose *Lactobacillus reuteri*, which is a commonly used probiotic strain. *L. reuteri* has been shown to be a true symbiont in rodents and densely colonizes the forestomach of mice (166, 167). The aim of this study presented in Chapter 2 was to systematically determine which genes of *L. reuteri* 100-23 contribute to tolerance towards host gastric acid secretion. A better understanding of microbial colonization factors in their host contributes to our understanding of ecological requirements for novel probiotic strains. If these and other findings are truly understood and applied to the rational selection of synbiotic strains, the next generation of synbiotic combinations may have a greater ecological advantage and could be more competitive than current synbiotic combinations on the market.

The second objective of this thesis was to test if rationally selected synbiotic applications have a higher potential to establish probiotic strains in the gastrointestinal tract. To answer this question, Chapter 3 analyses the potential of a novel technique to select synergistic synbiotic combinations. In this study we characterized the potential of an *in vivo* selected combination of *Bifidobacterium adolescentis* and GOS in a rat model.

After the ecological advantages of *in vivo* selected synbiotics had successfully been established, the third objective was to test the potential of this synbiotic combination in

human subjects in Chapter 4 in comparison to a commercial synbiotic. The rationally, *in vivo* selected synbiotic was established in the GI tract of the subjects in significantly higher numbers then the commercial synbiotic.

Together, the studies performed for this thesis present a comprehensive examination of the role of the stomach and dietary factors, such as probiotics, prebiotics and synbiotics, on the establishment of bacteria in the gastrointestinal tract.

# **1.9 References**

- Vyas U, Ranganathan N. 2012. Probiotics, prebiotics, and synbiotics: Gut and beyond. Gastroenterology Research and Practice 2012:1–16.
- Sartor RB. 2008. Microbial influences in inflammatory bowel diseases. Gastroenterology 134:577–94.
- Yasmin A, Butt MS, Afzaal M, van Baak M, Nadeem MT, Shahid MZ. 2015.
   Prebiotics, gut microbiota and metabolic risks: Unveiling the relationship. Journal of Functional Foods 17:189–201.
- Stecher B, Hardt W-D. 2008. The role of microbiota in infectious disease. Trends in Microbiology 16:107–114.
- Belkaid Y, Hand TW. 2014. Role of the microbiota in immunity and inflammation. Cell 157:121–141.
- Jonkers D. 2016. Microbial perturbations and modulation in conditions associated with malnutrition and malabsorption. Best Practice & Research Clinical Gastroenterology 30:161–172.
- Buffie CG, Pamer EG. 2013. Microbiota-mediated colonization resistance against intestinal pathogens. Nature Reviwes in Immunology 13:790–801.
- Walsh CJ, Guinane CM, O'Toole PW, Cotter PD. 2014. Beneficial modulation of the gut microbiota. FEBS Letters 588:4120–4130.
- McNeil NI. 1984. The contribution of the large-intestine to energy supplies in man. American Journal of Clinical Nutrition 39:338–342.

- 10. Angelakis E, Lagier J-C. 2016. Samples and techniques highlighting the links between obesity and microbiota. Microbial Pathogenesis.
- Tan J, McKenzie C, Potamitis M, Thorburn AN, Mackay CR, Macia L. 2014.
   The Role of Short-Chain Fatty Acids in Health and Disease, p. 91–120. *In* Alt, FW (ed.), Advances in Immunology, Volume 121 First Edit. Elsevier.
- 12. **Puertollano E, Kolida S, Yaqoob P**. 2014. Biological significance of short-chain fatty acid metabolism by the intestinal microbiome. Current opinion in clinical nutrition and metabolic care **17**:139–44.
- Suzuki T, Yoshida S, Hara H. 2008. Physiological concentrations of short-chain fatty acids immediately suppress colonic epithelial permeability. The British Journal of Nutrition 100:297–305.
- Krishnan S, Alden N, Lee K. 2015. Pathways and functions of gut microbiota metabolism impacting host physiology. Current Opinion in Biotechnology 36:137– 145.
- den Besten G, Eunen K van, Groen AK, Venema K, Reijngoud D-J, Bakker BM. 2013. The role of short-chain fatty acids in the interplay between diet, gut microbiota, and host energy metabolism. Journal of Lipid Research 54:2325–2340.
- 16. Tolhurst G, Heffron H, Lam YS, Parker HE, Habib AM, Diakogiannaki E, Cameron J, Grosse J, Reimann F, Gribble FM. 2012. Short-chain fatty acids stimulate glucagon-like peptide-1 secretion via the G-protein-coupled receptor FFAR2. Diabetes 61:364–371.
- 17. Tedelind S, Westberg F, Kjerrulf M, Vidal A. 2007. Anti-inflammatory

- Vinolo MAR, Rodrigues HG, Hatanaka E, Sato FT, Sampaio SC, Curi R. 2011.
   Suppressive effect of short-chain fatty acids on production of proinflammatory mediators by neutrophils. Journal of Nutritional Biochemistry 22:849–855.
- Park J-S, Lee E-J, Lee J-C, Kim W-K, Kim H-S. 2007. Anti-inflammatory effects of short chain fatty acids in IFN-gamma-stimulated RAW 264.7 murine macrophage cells: involvement of NF-kappaB and ERK signaling pathways. International immunopharmacology 7:70–77.
- Tazoe H, Otomo Y, Kaji I, Tanaka R, Karaki SI, Kuwahara A. 2008. Roles of short-chain fatty acids receptors, GPR41 and GPR43 on colonic functions. Journal of Physiology and Pharmacology 59:251–262.
- Hamer HM, Jonkers D, Venema K, Vanhoutvin S, Troost FJ, Brummer RJ.
   2008. Review article: The role of butyrate on colonic function. Alimentary Pharmacology and Therapeutics 27:104–119.
- Tang Y, Chen Y, Jiang H, Robbins GT, Nie D. 2011. G-protein-coupled receptor for short-chain fatty acids suppresses colon cancer. International Journal of Cancer 128:847–856.
- 23. Brown AJ, Goldsworthy SM, Barnes AA, Eilert MM, Tcheang L, Daniels D, Muir AI, Wigglesworth MJ, Kinghorn I, Fraser NJ, Pike NB, Strum JC, Steplewski KM, Murdock PR, Holder JC, Marshall FH, Szekeres PG, Wilson

**S**, **Ignar DM**, **Foord SM**, **Wise A**, **Dowell SJ**. 2003. The orphan G protein-coupled receptors GPR41 and GPR43 are activated by propionate and other short chain carboxylic acids. Journal of Biological Chemistry **278**:11312–11319.

- 24. Aoyama M, Kotani J, Usami M. 2010. Butyrate and propionate induced activated or non-activated neutrophil apoptosis via HDAC inhibitor activity but without activating GPR-41/GPR-43 pathways. Nutrition **26**:653–661.
- 25. Wang HG, Huang XD, Shen P, Li LR, Xue HT, Ji GZ. 2013. Anticancer effects of sodium butyrate on hepatocellular carcinoma cells *in vitro*. International journal of molecular medicine **31**:967–974.
- 26. Foglietta F, Serpe L, Canaparo R, Vivenza N, Riccio G, Imbalzano E, Gasco P, Zara GP. 2014. Modulation of Butyrate Anticancer Activity by Solid Lipid Nanoparticle Delivery: An *in Vitro* Investigation on Human Breast Cancer and Leukemia Cell Lines. Journal of Pharmacy & Pharmaceutical Sciences 17:231–247.
- Yamamura T, Matsumoto N, Matsue Y, Okudera M, Nishikawa Y, Abiko Y, Komiyama K. 2014. Sodium butyrate, a histone deacetylase inhibitor, regulates lymphangiogenic factors in oral cancer cell line HSC-3. Anticancer Research 34:1701–1708.
- Wlodarska M, Kostic AD, Xavier RJ. 2015. An integrative view of microbiomehost interactions in inflammatory bowel diseases. Cell Host and Microbe 17:577– 591.
- 29. Davis LMG, Martínez I, Walter J, Goin C, Hutkins RW. 2011. Barcoded pyrosequencing reveals that consumption of galactooligosaccharides results in a

highly specific bifidogenic response in humans. PloS One 6:e25200.

- 30. Wu GD, Chen J, Hoffmann C, Bittinger K, Chen Y, Keilbaugh SA, Bewtra M, Knights D, Walters WA, Knight R, Sinha R, Gilroy E, Gupta K, Baldassano R, Nessel L, Li H, Bushman FD, Lewis JD. 2011. Linking Long-Term Dietary Patterns with Gut Microbial Enterotypes. Science 334:105–109.
- 31. Rattanaprasert M, Roos S, Hutkins RW, Walter J. 2014. Quantitative evaluation of synbiotic strategies to improve persistence and metabolic activity of *Lactobacillus reuteri* DSM 17938 in the human gastrointestinal tract. Journal of Functional Foods 10:85–94.
- Kolida S, Gibson GR. 2011. Synbiotics in health and disease. Annual Review of Food Science and Technology 2:373–393.
- 33. Mechnikoff E. 1908. The prolongation of life.
- Sanders ME. 2015. Probiotics in 2015: Their Scope and Use. Journal of Clinical Gastroenterology 49:S2–S6.
- Saulnier DMA. 2008. Synbiotics: making the most of prbiotics by their combinations?, p. 9–19. *In* Food Science and Technology Bulletin: Functional Foods, Volume 4.
- 36. Guidance Document The Use of Probiotic Microorganisms in Food. http://wwwinspectiongcca/food/labelling/food-labelling-for-industry/healthclaims/eng/1392834838383/1392834887794?chap=9, and http://wwwhc-scgcca/fnan/label-etiquet/claims-reclam/probiotics-probiotiques-engphp.

- Farnworth E. 2008. The evidence to support health claims for probiotics. The Journal of nutrition 138:1250–1254.
- 38. Food Standards Australia and New Zeeland, Food Standard code as of 2016. http://wwwfoodstandardsgovau/consumer/labelling/nutrition/Pages/defaultaspx.
- 39. Sanders ME, Lenoir-Wijnkoop I, Salminen S, Merenstein DJ, Gibson GR, Petschow BW, Nieuwdorp M, Tancredi DJ, Cifelli CJ, Jacques P, Pot B. 2014. Probiotics and prebiotics: prospects for public health and nutritional recommendations. Annals of the New York Academy of Sciences 1309:19–29.
- D'Souza AL, Rajkumar C, Cooke J, Bulpitt CJ. 2002. Probiotics in prevention of antibiotic associated diarrhoea: meta-analysis. British Medical Association 324:1–6.
- Deshpande G, Rao S, Patole S, Bulsara M. 2010. Updated meta-analysis of probiotics for preventing necrotizing enterocolitis in preterm neonates. Pediatrics 125:921–930.
- 42. Shen J, Zuo ZX, Mao AP. 2014. Effect of probiotics on inducing remission and maintaining therapy in ulcerative colitis, Crohn's disease, and pouchitis: meta-analysis of randomized controlled trials. Inflammatory Bowel Diseases 20:21–35.
- Ruan Y, Sun J, He J, Chen F, Chen R, Chen H. 2015. Effect of probiotics on glycemic control: A systematic review and meta-analysis of randomized, controlled trials. PLoS ONE 10:1–15.
- 44. **Gao X**, **Zhu Y**, **Wen Y**, **Liu G**, **Wan C**. 2016. Efficacy of probiotics in nonalcoholic fatty liver disease in adult and children: A meta-analysis of randomized controlled

trials. Hepatology Research. In press.

- 45. **Sun J, Buys NJ**. 2016. Glucose- and glycaemic factor-lowering effects of probiotics on diabetes: a meta-analysis of randomised placebo-controlled trials. British Journal of Nutrition **115**:1167–1177.
- 46. Gibson GR, Roberfroid MB. 1995. Dietary Modulation of the Human Colonic Microbiota : Introducing the Concept of Prebiotics. Journal of Nutrition 125:1401– 1412.
- 47. Hutkins RW, Krumbeck J a., Bindels LB, Cani PD, Fahey G, Goh YJ, Hamaker B, Martens EC, Mills D a., Rastal R a., Vaughan E, Sanders ME. 2016. Prebiotics: Why definitions matter. Current Opinion in Biotechnology 37:1–7.
- Bindels LB, Delzenne NM, Cani PD, Walter J. 2015. Towards a more comprehensive concept for prebiotics. Nature Reviews Gastroenterology & Hepatology 12:303–310.
- Krumbeck JA, Maldonado-Gomez MX, Ramer-Tait AE, Hutkins RW. 2016.
   Prebiotics and synbiotics: dietary strategies for improving gut health. Current Opinion in Gastroenterology 32:110–119.
- Walker AW, Lawley TD. 2013. Therapeutic modulation of intestinal dysbiosis.
   Pharmacological Research 69:75–86.
- Preidis GA, Versalovic J. 2009. Targeting the Human Microbiome With Antibiotics, Probiotics, and Prebiotics: Gastroenterology Enters the Metagenomics Era. Gastroenterology 136:2015–2031.

- Quintero-Villegas MI, Wittke A, Hutkins R. 2014. Adherence Inhibition of Cronobacter sakazakii to Intestinal Epithelial Cells by Lactoferrin. Current Microbiology 69:574–579.
- 53. Beserra BTS, Fernandes R, do Rosario V a., Mocellin MC, Kuntz MGF, Trindade EBSM. 2015. A systematic review and meta-analysis of the prebiotics and synbiotics effects on glycaemia, insulin concentrations and lipid parameters in adult patients with overweight or obesity. Clinical Nutrition 34:845–858.
- 54. Lohner S, Küllenberg D, Antes G, Decsi T, Meerpohl JJ. 2014. Metabolic benefits of dietary prebiotics in human subjects: a systematic review of randomised controlled trials. Clinical Nutrition 34:958–965.
- Zaman SA, Sarbini SR. 2015. The potential of resistant starch as a prebiotic. Critical Reviews in Biotechnology 8551:1–7.
- 56. Collado Yurrita L, San Mauro Martín I, Ciudad-Cabañas MJ, Calle-Purón ME, Hernández Cabria M. 2014. Effectiveness of inulin intake on indicators of chronic constipation; a meta-analysis of controlled randomized clinical trials. Nutrición Hospitalaria 30:244–252.
- 57. Salonen A, Lahti L, Salojärvi J, Holtrop G, Korpela K, Duncan SH, Date P, Farquharson F, Johnstone AM, Lobley GE, Louis P, Flint HJ, de Vos WM.
  2014. Impact of diet and individual variation on intestinal microbiota composition and fermentation products in obese men. The ISME journal 8:2218–2230.
- 58. **Martínez I, Kim J, Duffy PR, Schlegel VL, Walter J**. 2010. Resistant starches types 2 and 4 have differential effects on the composition of the fecal microbiota in

human subjects. PloS one 5:e15046.

- 59. Kovatcheva-Datchary P, Nilsson A, Akrami R, Lee YS, De Vadder F, Arora T, Hallen A, Martens E, Bjoerck I, Baeckhed F. 2015. Dietary Fiber-Induced Improvement in Glucose Metabolism Is Associated with Increased Abundance of Prevotella. Cell Metabolism 22:971–982.
- 60. Krumbeck JA, Maldonado-Gomez MX, Martínez I, Frese SA, Burkey TE, Rasineni K, Ramer-Tait AE, Harris EN, Hutkins RW, Walter J. 2015. In Vivo Selection to identify bacterial strains with enhanced ecological performance in synbiotic applications. Applied and Environmental Microbiology 81:2455–2465.
- Ventura M, O'Flaherty S, Claesson MJ, Turroni F, Klaenhammer TR, van Sinderen D, O'Toole PW. 2009. Genome-scale analyses of health-promoting bacteria: probiogenomics. Nature Reviews Microbiology 7:61–71.
- Sonnenburg ED, Zheng H, Joglekar P, Higginbottom SK, Firbank SJ, Bolam DN, Sonnenburg JL. 2010. Specificity of Polysaccharide Use in Intestinal Bacteroides Species Determines Diet-Induced Microbiota Alterations. Cell 141:1241–1252.
- 63. Tanaka R, Takayama H, Morotomi M, Kuroshima T, Ueyama S, Matsumoto K, Kuroda A, Masahiko M. 1983. Effects of Administration of TOS and Bifidobacterium breve 4006 on the Human Fecal Flora. Bifidobacteria and Microflora 2:17–24.
- 64. Wang X, Brown IL, Evans AJ, Conway PL. 1999. The protective effects of high amylose maize (amylomaize) starch granules on the survival of *Bifidobacterium*

spp. in the mouse intestinal tract. Journal of applied microbiology 87:631–639.

- 65. Femia A Pietro, Luceri C, Dolara P, Giannini A, Biggeri A, Salvadori M, Clune Y, Collins KJ, Paglierani M, Caderni G. 2002. Antitumorigenic activity of the prebiotic inulin enriched with oligofructose in combination with the probiotics *Lactobacillus rhamnosus* and *Bifidobacterium lactis* on azoxymethane-induced colon carcinogenesis in rats. Carcinogenesis 23:1953–1960.
- 66. Ogawa T, Asai Y, Yasuda K, Sakamoto H. 2005. Oral immunoadjuvant activity of a new synbiotic *Lactobacillus casei* subsp *casei* in conjunction with dextran in BALB/c mice. Nutrition Research 25:295–304.
- 67. Ogawa T, Asai Y, Tamai R, Makimura Y, Sakamoto H, Hashikawa S, Yasuda K. 2006. Natural killer cell activities of synbiotic *Lactobacillus casei* ssp. *casei* in conjunction with dextran. Clinical and experimental immunology 143:103–109.
- 68. **Shukla S, Shukla A, Mehboob S, Guha S**. 2011. Meta-analysis: the effects of gut flora modulation using prebiotics, probiotics and synbiotics on minimal hepatic encephalopathy. Alimentary Pharmacology & Therapeutics **33**:662–671.
- Cerezuela R, Meseguer J, Esteban MA. 2011. Current Knowledge in Synbiotic Use for Fish Aquaculture: A Review. Journal of Aquaculture Research & Development 01:1–7.
- 70. Hasunuma T, Kawashima K, Nakayama H, Murakami T, Kanagawa H, Ishii T, Akiyama K, Yasuda K, Terada F, Kushibiki S. 2011. Effect of cellooligosaccharide or synbiotic feeding on growth performance, fecal condition and hormone concentrations in Holstein calves. Animal Science Journal 82:543–

- 548.
- 71. Sopková D, Hertelyová Z, Andrejčáková Z, Vlčková R, Gancarčíková S, Petrilla V, Ondrašovičová S, Krešáková L. 2016. The application of probiotics and flaxseed promotes metabolism of n-3 polyunsaturated fatty acids in pigs. Journal of Applied Animal Research 2119:1–6.
- 72. Abdel-Raheem S, Abd-Allah S, Hassanein K. 2012. The effects of prebiotic, probiotic and synbiotic supplementation on intestinal microbial ecology and histomorphology of broiler chickens. International Journal for Agro Veterinary and Medical Sciences 6:277–289.
- 73. Hart ML, Suchodolski JS, Steiner JM, Webb CB. 2012. Open-label trial of a multi-strain synbiotic in cats with chronic diarrhea. Journal of feline medicine and surgery 14:240–245.
- 74. Gagné JW, Wakshlag JJ, Simpson KW, Dowd SE, Latchman S, Brown DA, Brown K, Swanson KS, Fahey GC. 2013. Effects of a synbiotic on fecal quality, short-chain fatty acid concentrations, and the microbiome of healthy sled dogs. BMC Veterinary Research 9:246–256.
- 75. **Bengmark S**. 2012. Integrative medicine and human health the role of pre-, proand synbiotics. Clinical and Translational Medicine **1**:1–13.
- 76. Awad W, Ghareeb K, Böhm J. 2008. Intestinal structure and function of broiler chickens on diets supplemented with a synbiotic containing *Enterococcus faecium* and oligosaccharides. International Journal of Molecular Sciences 9:2205–2216.
- 77. Mugambi MN, Musekiwa A, Lombard M, Young T, Blaauw R. 2012.

Synbiotics, probiotics or prebiotics in infant formula for full term infants: a systematic review. Nutrition journal **11**:81–103.

- 78. Simeoli R, Raso GM, Lama A, Pirozzi C, Santoro A, Di Guida F, Sanges M, Aksoy E, Calignano A, D'Arienzo A, Meli R. 2015. Preventive and Therapeutic Effects of *Lactobacillus Paracasei* B21060-Based Synbiotic Treatment on Gut Inflammation and Barrier Integrity in Colitic Mice. Journal of Nutrition 145:1202– 1210.
- 79. Bindels LB, Neyrinck AM, Claus SP, Le Roy CI, Grangette C, Pot B, Martinez I, Walter J, Cani PD, Delzenne NM. 2015. Synbiotic approach restores intestinal homeostasis and prolongs survival in leukaemic mice with cachexia. The ISME Journal 1–15.
- Asahara T, Takahashi A, Yuki N, Kaji R, Takahashi T, Nomoto K. 2016. Protective Effect of a Synbiotic against Multidrug-Resistant *Acinetobacter baumannii* in a Murine Infection Model. Antimicrobial Agents and Chemotherapy 60:3041–3050.
- 81. Ford AC, Quigley EMM, Lacy BE, Lembo AJ, Saito YA, Schiller LR, Soffer EE, Spiegel BMR, Moayyedi P. 2014. Efficacy of Prebiotics, Probiotics, and Synbiotics in Irritable Bowel Syndrome and Chronic Idiopathic Constipation: Systematic Review and Meta-analysis. The American journal of gastroenterology 5498:1–15.
- Kinross JM, Markar S, Karthikesalingam A, Chow A, Penney N, Silk D, Darzi
  A. 2013. A Meta-Analysis of Probiotic and Synbiotic Use in Elective Surgery: Does

Nutrition Modulation of the Gut Microbiome Improve Clinical Outcome? Journal of Parenteral and Enteral Nutrition **37**:243–253.

- Chang Y-S, Trivedi MK, Jha A, Lin Y-F, Dimaano L, García-Romero MT.
   2016. Synbiotics for Prevention and Treatment of Atopic Dermatitis. Journal of the American Medical Association of Pediatrics 170.3:236-242.
- 84. Sawas T, Al halabi S, Hernaez R, Carey WD, Cho WK. 2015. Patients Receiving Prebiotics and Probiotics Before Liver Transplantation Develop Fewer Infections Than Controls: A Systematic Review and Meta-analysis. Clinical Gastroenterology and Hepatology 13:1567–1574.e3.
- 85. Yang Z, Wu Q, Liu Y, Fan D. 2016. Effect of Perioperative Probiotics and Synbiotics on Postoperative Infections After Gastrointestinal Surgery: A Systematic Review With Meta-Analysis. Journal of parenteral and enteral nutrition 0148607116629670.
- 86. Ghouri YA, Richards DM, Rahimi EF, Krill JT, Jelinek K a, DuPont AW. 2014. Systematic review of randomized controlled trials of probiotics, prebiotics, and synbiotics in inflammatory bowel disease. Clinical and experimental gastroenterology 7:473–487.
- 87. Saez-Lara MJ, Gomez-llorente C, Plaza-Diaz J, Gil A. 2015. The Role of Probiotic Lactic Acid Bacteria and Bifidobacteria in the Prevention and Treatment of Inflammatory Bowel Disease and Other Related Diseases : A Systematic Review of Randomized Human Clinical Trials. BioMed research international. In press.
- 88. Hedin C, Whelan K, Lindsay JO. 2007. Evidence for the use of probiotics and
prebiotics in inflammatory bowel disease: a review of clinical trials. Proceedings of the Nutrition Society **66**:307–315.

- 89. **Zigra PI, Maipa VE, Alamanos YP**. 2007. Probiotics and remission of ulcerative colitis: A systematic review. Netherlands Journal of Medicine **65**:411–418.
- 90. **Mugambi MN**, **Musekiwa A**, **Lombard M**, **Young T**, **Blaauw R**. 2013. Association between funding source, methodological quality and research outcomes in randomized controlled trials of synbiotics, probiotics and prebiotics added to infant formula: a systematic review. BMC medical research methodology **13**:137.
- 91. Fujimori S, Gudis K, Mitsui K, Seo T, Yonezawa M, Tanaka S, Tatsuguchi A, Sakamoto C. 2009. A randomized controlled trial on the efficacy of synbiotic versus probiotic or prebiotic treatment to improve the quality of life in patients with ulcerative colitis. Nutrition 25:520–525.
- 92. Worthley DL, Le Leu RK, Whitehall VL, Conlon M, Christophersen C, Belobrajdic D, Mallitt K-A, Hu Y, Irahara N, Ogino S, Leggett BA, Young GP. 2009. A human, double-blind, placebo-controlled, crossover trial of prebiotic, probiotic, and synbiotic supplementation: effects on luminal, inflammatory, epigenetic, and epithelial biomarkers of colorectal cancer. The American journal of clinical nutrition 90:578–586.
- 93. Furrie E, Macfarlane S, Kennedy A, Cummings JH, Walsh SV, O'Neil DA, Macfarlane GT. 2005. Synbiotic therapy (*Bifidobacterium longum*/Synergy 1) initiates resolution of inflammation in patients with active ulcerative colitis: a randomised controlled pilot trial. Gut 54:242–249.

- 94. Ishikawa H, Matsumoto S, Ohashi Y, Imaoka A, Setoyama H, Umesaki Y, Tanaka R, Otani T. 2011. Beneficial effects of probiotic *Bifidobacterium* and galacto-oligosaccharide in patients with ulcerative colitis: A randomized controlled study. Digestion 84:128–133.
- 95. Steed H, Macfarlane GT, Blackett KL, Bahrami B, Reynolds N, Walsh S V., Cummings JH, Macfarlane S. 2010. Clinical trial: the microbiological and immunological effects of synbiotic consumption - a randomized double-blind placebo-controlled study in active Crohn's disease. Alimentary Pharmacology & Therapeutics 32:872–883.
- 96. Glanville J, King S, Guarner F, Hill C, Sanders ME. 2015. A review of the systematic review process and its applicability for use in evaluating evidence for health claims on probiotic foods in the European Union. Nutrition journal **14**:16.
- 97. **Pitsouni E**, **Alexiou V**, **Saridakis V**, **Peppas G**, **Falagas ME**. 2009. Does the use of probiotics/synbiotics prevent postoperative infections in patients undergoing abdominal surgery? A meta-analysis of randomized controlled trials. European journal of clinical pharmacology **65**:561–570.
- 98. Rossi M, Klein K, Johnson DW, Campbell KL. 2012. Pre-, pro-, and synbiotics: Do they have a role in reducing uremic toxins? a systematic review and metaanalysis. International Journal of Nephrology 2012.
- 99. Zhang MM, Cheng JQ, Lu YR, Yi ZH, Yang P, Wu XT. 2010. Use of pre-, proand synbiotics in patients with acute pancreatitis: A meta-analysis. World Journal of Gastroenterology 16:3970–3978.

- Watkinson PJ, Barber VS, Dark P, Young JD. 2007. The use of pre- pro- and synbiotics in adult intensive care unit patients: Systematic review. Clinical Nutrition 26:182–192.
- 101. Watkinson PJ, Barber VS, Dark P, Young JD. 2007. The use of pre- pro- and synbiotics in adult intensive care unit patients: Systematic review. Clinical Nutrition 26:182–192.
- 102. He D, Wang HY, Feng JY, Zhang MM, Zhou Y, Wu XT. 2013. Use of pro-/synbiotics as prophylaxis in patients undergoing colorectal resection for cancer: A meta-analysis of randomized controlled trials. Clinics and Research in Hepatology and Gastroenterology 37:406–415.
- 103. Dang D, Zhou W, Lun Z, Mu X, Wang D, Wu H. 2013. Meta-analysis of probiotics and/or prebiotics for the prevention of eczema. The Journal of international medical research 41:1426–1436.
- 104. Lytvyn L, Quach K, Banfield L, Johnston BC, Mertz D. 2015. Probiotics and synbiotics for the prevention of postoperative infections following abdominal surgery: a systematic review and meta-analysis of randomized controlled trials. Journal of Hospital Infection 1.
- Petrof EO, Dhaliwal R, Manzanares W, Johnstone J, Cook D, Heyland DK.
   2012. Probiotics in the critically ill: a systematic review of the randomized trial evidence. Critical care medicine 40:3290–302.
- 106. Arumugam S, Lau CSM, Chamberlain RS. 2016. Probiotics and Synbiotics Decrease Postoperative Sepsis in Elective Gastrointestinal Surgical Patients: a

Meta-Analysis. Journal of Gastrointestinal Surgery 20:1123–1131.

- 107. Van Vliet-Ostaptchouk JV, Nuotio ML, Slagter SN, Doiron D, Fischer K, Foco L, Gaye A, Gögele M, Heier M, Hiekkalinna T, Joensuu A, Newby C, Pang C, Partinen E, Reischl E, Schwienbacher C, Tammesoo ML, Swertz MA, Burton P, Ferretti V, Fortier I, Giepmans L, Harris JR, Hillege HL, Holmen J, Jula A, Kootstra-Ros JE, Kvaløy K, Holmen TL, Männistö S, Metspalu A, Midthjell K, Murtagh MJ, Peters A, Pramstaller PP, Saaristo T, Salomaa V, Stolk RP, Uusitupa M, van der Harst P, van der Klauw MM, Waldenberger M, Perola M, Wolffenbuttel BH. 2014. The prevalence of Metabolic Syndrome and metabolically healthy obesity in Europe: a collaborative analysis of ten large cohort studies. BMC Endocrine Disorders 14:1–13.
- 108. Andersen CJ, Fernandez ML. 2013. Dietary strategies to reduce metabolic syndrome. Reviews in Endocrine and Metabolic Disorders 14:241–254.
- 109. Cornier M-A, Dabelea D, Hernandez TL, Lindstrom RC, Steig AJ, Stob NR, Van Pelt RE, Wang H, Eckel RH. 2008. The Metabolic Syndrome. Endrocrine Reviews 29:777–822.
- 110. Eslamparast T, Zamani F, Hekmatdoost A, Sharafkhah M, Eghtesad S, Malekzadeh R, Poustchi H. 2014. Effects of synbiotic supplementation on insulin resistance in subjects with the metabolic syndrome: a randomised, double-blind, placebo-controlled pilot study. British Journal of Nutrition FirstView:1–8.
- 111. Fernandes R, Beserra BTS, Mocellin MC, Kuntz MGF, da Rosa JS, de Miranda RCD, Schreiber CSO, Froede TS, Nunes EA, Trindade EBSM. 2016. Effects of

Prebiotic and Synbiotic Supplementation on Inflammatory Markers and Anthropometric Indices After Roux-en-Y Gastric Bypass. A randomized, Tripleblind, Placebo-controlled Pilot Study. Journal of Clinical Gastroenterology **50**:208– 217.

- 112. Asemi Z, Khorrami-Rad A, Alizadeh SA, Shakeri H, Esmaillzadeh A. 2014. Effects of synbiotic food consumption on metabolic status of diabetic patients: A double-blind randomized cross-over controlled clinical trial. Clinical Nutrition 33:198–203.
- 113. Asemi Z, Alizadeh S-A, Ahmad K, Goli M, Esmaillzadeh A. 2015. Effects of beta-carotene fortified synbiotic food on metabolic control of patients with type 2 diabetes mellitus: A double-blind randomized cross-over controlled clinical trial. Clinical nutrition (Edinburgh, Scotland) 35:819–825.
- 114. Pourghassem Gargari B, Dehghan P, Aliasgharzadeh A, Asghari Jafar-Abadi
  M. 2013. Effects of high performance inulin supplementation on glycemic control and antioxidant status in women with type 2 diabetes. Diabetes Metab J 37:140–148.
- 115. Dehghan P, Pourghassem Gargari B, Asghari Jafar-abadi M. 2014. Oligofructose-enriched inulin improves some inflammatory markers and metabolic endotoxemia in women with type 2 diabetes mellitus: A randomized controlled clinical trial. Nutrition 30:418–423.
- 116. Dehghan P, Pourghassem Gargari B, Asghari Jafar-abadi M, AliasgharzadehA. 2014. Inulin controls inflammation and metabolic endotoxemia in women with

type 2 diabetes mellitus: a randomized-controlled clinical trial. Journal of Food Sciences and Nutrition **65**:117–123.

- 117. Cutting SM. 2011. Bacillus probiotics. Food Microbiology 28:214–220.
- 118. Bahmani F, Tajadadi-Ebrahimi M, Kolahdooz F, Mazouchi M, Hadaegh H, Jamal AS, Mazroii N, Asemi S, Asemi Z. 2015. The Consumption of Synbiotic Bread Containing *Lactobacillus sporogenes* and Inulin Affects Nitric Oxide and Malondialdehyde in Patients with Type 2 Diabetes Mellitus: Randomized, Double-Blind, Placebo-Controlled Trial. Journal of the American College of Nutrition 5724:1–8.
- 119. Rabiei S, Shakerhosseini R, Saadat N. 2015. The effects of symbiotic therapy on anthropometric measures, body composition and blood pressure in patient with metabolic syndrome: a triple blind RCT. Medical journal of the Islamic Republic of Iran 29:213.
- 120. Akram Kooshki A, Tofighiyan T, Rakhshani MH. 2015. Effects of Synbiotics on Inflammatory Markers in Patients With Type 2 Diabetes Mellitus. Global Journal of Health Science 7:6–10.
- 121. Haller D, Antoine J, Bengmark S, Enck P, Rijkers GT. 2010. Guidance for Substantiating the Evidence for Beneficial Effects of Probiotics : Probiotics in Chronic Inflammatory Bowel Disease and the Functional Disorder Irritable Bowel Syndrome 1 – 3. The Journal of Nutrition 140:690–698.
- 122. Goyal N, Rana A, Ahlawat A, Bijjem KR V, Kumar P. 2014. Animal models of inflammatory bowel disease: a review. Inflammopharmacology 22:219–33.

- 123. Molodecky NA, Soon IS, Rabi DM, Ghali WA, Ferris M, Chernoff G, Benchimol EI, Panaccione R, Ghosh S, Barkema HW, Kaplan GG. 2012. Increasing incidence and prevalence of the inflammatory bowel diseases with time, based on systematic review. Gastroenterology 142:46–54.e42; quiz e30.
- Ardizzone S, Bianchi Porro G. 2005. Biologic Therapy for Inflammatory Bowel Disease. Drugs 65:2253–2286.
- Mahida YR, Rolfe VE. 2004. Host bacterial interactions in inflammatory bowel disease. Clinical Science 107:331–341.
- 126. Davis LMG, Martínez I, Walter J, Hutkins R. 2010. A dose dependent impact of prebiotic galactooligosaccharides on the intestinal microbiota of healthy adults. International Journal of Food Microbiology 144:285–292.
- 127. Chermesh I, Tamir A, Reshef R, Chowers Y, Suissa A, Katz D, Gelber M, Halpern Z, Bengmark S, Eliakim R. 2007. Failure of synbiotic 2000 to prevent postoperative recurrence of Crohn's disease. Digestive Diseases and Sciences 52:385–389.
- 128. Fujimori S, Tatsuguchi A, Gudis K, Kishida T, Mitsui K, Ehara A, Kobayashi T, Sekita Y, Seo T, Sakamoto C. 2007. High dose probiotic and prebiotic cotherapy for remission induction of active Crohn's disease. Journal of Gastroenterology and Hepatology (Australia) 22:1199–1204.
- 129. Rossi R, Rossi L, Fassio F. 2015. Clinical Follow-up of 96 Patients Affected by Irritable Bowel Syndrome Treated with a Novel Multi-strain Symbiotic. Journal of Contemporary Immunology 2:49–58.

- 130. Dinleyici EC, Dalgic N, Guven S, Ozen M, Kara A, Metin-Timur VAO, Sancar M, Kurugol Z, Tanir G, Ozturk D, Aydogdu S, Tutanc M, Eren M, Vandenplas Y. 2013. The effect of a multispecies synbiotic mixture on the duration of diarrhea and length of hospital stay in children with acute diarrhea in Turkey: Single blinded randomized study. European Journal of Pediatrics 172:459–464.
- 131. **Dewi MR**, **Soenarto Y**, **Karyana IPG**. 2015. Efficacy of synbiotic treatment in children with acute rotavirus diarrhea. Paediatrica Indonesiana **55**:74–78.
- 132. Passariello A, Terrin G, Cecere G, Micillo M, De Marco G, Di Costanzo M, Cosenza L, Leone L, Nocerino R, Berni Canani R. 2012. Randomised clinical trial: Efficacy of a new synbiotic formulation containing *Lactobacillus paracasei* B21060 plus arabinogalactan and xilooligosaccharides in children with acute diarrhoea. Alimentary Pharmacology and Therapeutics 35:782–788.
- 133. Ringel-Kulka T, Kotch JB, Jensen ET, Savage E, Weber DJ. 2015. Randomized, double-blind, placebo-controlled study of synbiotic yogurt effect on the health of children. The Journal of pediatrics 166:1475–1481.e3.
- 134. Virk A, Mandrekar J, Berbari EF, Boyce TG, Fischer PR, Kasten MJ, Orenstein R, Rosenblatt JE, Sampathkumar P, Sia I, Springer D, Witzig TE. 2013. A randomized, double blind, placebo-controlled trial of an oral synbiotic (AKSB) for prevention of travelers' diarrhea. Journal of Travel Medicine 20:88–94.
- 135. Lieske JC, Tremaine WJ, De Simone C, O'Connor HM, Li X, Bergstralh EJ, Goldfarb DS. 2010. Diet, but not oral probiotics, effectively reduces urinary oxalate excretion and calciumoxalate supersaturation. Kidney international 78:1178–1185.

- 136. Jones RJ, Hussein HM, Zagorec M, Brightwell G, Tagg JR. 2008. Isolation of lactic acid bacteria with inhibitory activity against pathogens and spoilage organisms associated with fresh meat. Food Microbiology 25:228–234.
- 137. Willett WC. 2000. Diet and Cancer. The Oncologist 5:393–404.
- 138. Raman M, Ambalam P, Kondepudi KK, Pithva S, Kothari C, Patel AT, Purama RK, Dave JM, Vyas BRM. 2013. Potential of probiotics , prebiotics and synbiotics for management of colorectal cancer. Gut microbes 0976:181–192.
- 139. Rafter J, Bennett M, Caderni G, Clune Y, Hughes R, Karlsson PC, Klinder A, O'Riordan M, O'Sullivan GC, Pool-Zobel B, Rechkemmer G, Roller M, Rowland I, Salvadori M, Thijs H, Van Loo J, Watzl B, Collins JK. 2007. Dietary synbiotics reduce cancer risk factors in polypectomized and colon cancer patients. Am J Clin Nutr 85:488–496.
- 140. Roller M, Clune Y, Collins K, Rechkemmer G, Watzl B. 2007. Consumption of prebiotic inulin enriched with oligofructose in combination with the probiotics *Lactobacillus rhamnosus* and *Bifidobacterium lactis* has minor effects on selected immune parameters in polypectomised and colon cancer patients. The British journal of nutrition 97:676–684.
- 141. Van Loo J, Clune Y, Bennett M, Collins JK. 2005. The SYNCAN project: goals, set-up, first results and settings of the human intervention study. The British journal of nutrition 93 Suppl 1:S91–S98.
- 142. Canavan C, West J, Card T. 2014. The epidemiology of irritable bowel syndrome. Clinical Epidemiology 6:71–80.

- 143. Endo Y, Shoji T, Fukudo S. 2015. Epidemiology of irritable bowel syndrome. Annals of Gastroenterology 28:158–159.
- 144. Gibson PR, Varney J, Malakar S, Muir JG. 2015. Food Components and Irritable Bowel Syndrome. Gastroenterology 148:1158–1174.e4.
- 145. Cappello C, Tremolaterra F, Pascariello A, Ciacci C, Iovino P. 2013. A randomised clinical trial (RCT) of a symbiotic mixture in patients with irritable bowel syndrome (IBS): Effects on symptoms, colonic transit and quality of life. International Journal of Colorectal Disease 28:349–358.
- 146. Šmid A, Strniša L, Bajc K, Vujić-Podlipec D, Bogovič Matijašić B, Rogelj I. 2016. Randomized clinical trial: The effect of fermented milk with the probiotic cultures *Lactobacillus acidophilus* La-5® and *Bifidobacterium* BB-12® and Beneo dietary fibres on health-related quality of life and the symptoms of irritable bowel syndrome in adult. Journal of Functional Foods 24:549–557.
- 147. Bittner AC, Croffut RM, Stranahan MC, Yokelson TN. 2007. Prescript-Assist TM probiotic-prebiotic treatment for irritable bowel syndrome: An open-label, partially controlled, 1-year extension of a previously published controlled clinical trial. Clinical Therapeutics 29:1153–1160.
- 148. Bittner AC, Croffut RM, Stranahan MC. 2005. Prescript-Assist TM probioticprebiotic treatment for irritable bowel syndrome: A methodologically oriented, 2week, randomized, placebo-controlled, double-blind clinical study. Clinical Therapeutics 27:755–761.
- 149. Andriulli A, Neri M, Loguercio C, Terreni N, Merla A, Cardarella MP,

Federico A, Chilovi F, Milandri GL, De Bona M, Cavenati S, Gullini S, Abbiati R, Garbagna N, Cerutti R, Grossi E. 2008. Clinical trial on the efficacy of a new symbiotic formulation, Flortec, in patients with irritable bowel syndrome: a multicenter, randomized study. Journal of clinical gastroenterology **42 Suppl 3**:S218–223.

- 150. Dughera L, Elia C, Navino M, Cisaro F, Aimo G, Bertele A, Calabrese C, De Felici I, Diligente G, Febbraro I, Franze A, Fries W, Martorano M, Scifo V, Villardo L. 2007. Effects of symbiotic preparations on constipated irritable bowel syndrome symptoms. Acta Biomedica de l'Ateneo Parmense 78:111–116.
- 151. Colecchia A, Vestito A, La Rocca A, Pasqui F, Nikiforaki A, Festi D. 2006. Effect of symbiotic preparation on the clinical manifestations of irritable bowel syndrome, constipation-variant. Minerva Gastroenterology and Dietology 52:349– 358.
- 152. Buriti FCA, Bedani R, Saad SMI. 2016. Probiotic and Prebiotic Dairy Desserts,
  p. 345–360. *In* Watson, RR, Preedy, VR (eds.), Probiotics, Prebioticos, and
  Synbiotics: Bioactive Foods in Health Promotion. Elsevier.
- Social Policy Indicators database (SPIN). SPINscan on Functional Food, 52 Weeks Ending 10/5/2014.
- 154. Hudson E. 2011. Global Probiotic Opportunities. Natural products insider.
- 155. **Young J**. 1998. European market developments in prebiotic- and probioticcontaining foodstuffs. The British journal of nutrition **80**:S231–233.
- 156. van Loveren H, Sanz Y, Salminen S. 2012. Health Claims in Europe: Probiotics

and Prebiotics as Case Examples. Annual Review of Food Science and Technology **3**:247–261.

- 157. Govender M, Choonara YE, Kumar P, du Toit LC, van Vuuren S, Pillay V. 2014. A review of the advancements in probiotic delivery: Conventional vs. nonconventional formulations for intestinal flora supplementation. AAPS PharmSciTech 15:29–43.
- 158. Konar N, Toker OS, Oba S, Sagdic O. 2016. Improving Functionality of Chocolate: A review on Probiotic, Prebiotic, and/or Synbiotic Characteristics. Trends in Food Science & Technology 49:35–44.
- 159. Ahmadi A, Milani E, Madadlou A, Mortazavi SA, Mokarram RR, Salarbashi
   D. 2014. Synbiotic yogurt ice cream produced via incorporation of microencapsulated *Lactobacillus acidophilus* (la-5) and fructooligosaccharide. Journal of Food Science and Technology 51:1568–1574.
- 160. Angiolillo L, Conte A, Faccia M, Zambrini AV, Del Nobile MA. 2014. A new method to produce synbiotic Fiordilatte cheese. Innovative Food Science and Emerging Technologies 22:180–187.
- 161. Petreska-Ivanovska T, Petrushevska-Tozi L, Grozdanov A, Petkovska R, Hadjieva J, Popovski E, Stafilov T, Mladenovska K. 2014. From optimization of synbiotic microparticles prepared by spray-drying to development of new functional carrot juice. Chemical Industry and Chemical Engineering Quarterly 20:549–564.
- 162. Fratianni F, Cardinale F, Russo I, Iuliano C, Tremonte P, Coppola R, NazzaroF. 2014. Ability of synbiotic encapsulated *Saccharomyces cerevisiae* boulardii to

grow in berry juice and to survive under simulated gastrointestinal conditions. Journal of microencapsulation **31**:299–305.

- 163. de Vrese M, Schrezenmeir J. 2008. Probiotics, prebiotics, and synbiotics.Advances in biochemical engineering/biotechnology 111:1–66.
- 164. Ashwani K, Vidisha T, Amarjeet K, Joshi VK. 2015. Synbiotics: A culinary Art to Creative Health Foods. International Journal of Food and Fermentation Technololgy 5:1–14.
- 165. Blandino A, Al-Aseeri ME, Pandiella SS, Cantero D, Webb C. 2003. Cerealbased fermented foods and beverages. Food Research International 36:527–543.
- 166. Frese SA, Mackenzie DA, Peterson DA, Schmaltz R, Fangman T, Zhou Y, Zhang C, Benson AK, Cody LA, Mulholland F, Juge N, Walter J. 2013. Molecular characterization of host-specific biofilm formation in a vertebrate gut symbiont. PLoS Genetics 9:e1004057.
- 167. Frese SA, Benson AK, Tannock GW, Loach DM, Kim J, Zhang M, Oh PL, Heng NCK, Patil PB, Juge N, Mackenzie DA, Pearson BM, Lapidus A, Dalin E, Tice H, Goltsman E, Land M, Hauser L, Ivanova N, Kyrpides NC, Walter J. 2011. The evolution of host specialization in the vertebrate gut symbiont *Lactobacillus reuteri*. PLoS Genetics 7:e1001314.

### Chapter 2

# Characterization of the ecological role of genes mediating acid resistance in *Lactobacillus reuteri* during colonization of the gastrointestinal tract.

# Preface

This chapter has been previously published: Characterization of the ecological role of genes mediating acid resistance in *Lactobacillus reuteri* during colonization of the gastrointestinal tract. Janina A. Krumbeck, Nathan L. Marsteller, Steven A. Frese, Daniel A. Peterson, Amanda E. Ramer-Tait, Robert W. Hutkins, and Jens Walter. *Environmetal Microbiology* (2015). doi:10.1111/1462-2920.1310

#### 2.1 Summary

Rodent-derived strains of *Lactobacillus reuteri* densely colonize the forestomach of mice and possess several genes whose predicted functions constitute adaptations towards an acidic environment. The objective of this study was to systematically determine which genes of *L. reuteri* 100-23 contribute to tolerance towards host gastric acid secretion. Genes predicted to be involved in acid resistance were inactivated, and their contribution to survival under acidic conditions was confirmed in model gastric juice. Fitness of five mutants that showed impaired *in vitro* acid resistance were then compared through competition experiments in ex-germ-free mice that were either treated with omeprazole, a proton-pump inhibitor that suppresses acid secretion in the stomach, or left untreated. This analysis revealed that the urease cluster was the predominant factor in mediating resistance to gastric acid production. Population levels of the mutant, which were substantially decreased in untreated mice, were almost completely restored through omeprazole, demonstrating that urease production in *L. reuteri* is mainly devoted to overcome gastric acid. The findings provide novel information on the mechanisms by which *L. reuteri* colonizes its gastric niche and demonstrate that in silico gene predictions and *in vitro* tests have limitations for predicting the ecological functions of colonization factors in bacterial symbionts.

# **2.2 Introduction**

A complex and diverse collection of microorganisms colonizes the gastrointestinal (GI) tract of mammals, affecting the health and immune status of the host. Among other functions, these microbial communities enhance energy absorption from ingested food, contribute to the development of their host's immune system, and provide colonization resistance against pathogens (Sekirov *et al.*, 2010). As a result of co-evolution, the bacteria that reside in the mammalian gut have developed a high degree of ecological fitness and specialization (Oh *et al.*, 2010; Frese *et al.*, 2011; O'Callaghan and O'Toole, 2013). Given the importance of the gut microbiota to the health of its host, there is currently much interest in formulating strategies that modulate its composition. However, remodeling this complex ecosystem requires an understanding of the mechanisms by which specific gut microbes colonize the GI tract and the factors that distinguish resident autochthonous members of the microbiota from allochthonous ones (Walter, 2008).

Among the bacteria that are autochthonous to several mammalian species is *Lactobacillus reuteri* (Walter, 2008). *L. reuteri* forms high populations in the rodent stomach that are maintained throughout the life of the animal. Colonization is achieved, in part, by the ability of the organism to adhere to the surface of the non-secretory epithelium present in the forestomach, resulting in formation of a biofilm-like structure (Walter *et al.*, 2007; Frese *et al.*, 2013). The ability of *L. reuteri* strains to form these biofilms is strictly dependent on their host origin, with only rodent isolates being capable of forming biofilms (Frese *et al.*, 2013). This translates to a higher ecological fitness of rodent strains when colonizing the mouse gastrointestinal tract (Frese *et al.*, 2011). *L. reuteri* is therefore an example of a bacterium that maintains a tight, host-specific relationship with its mammalian host (Oh *et al.*, 2010), and hence serves as a model to study ecologically important traits that facilitate host-microbe symbiosis in mammals at the molecular level (Frese *et al.*, 2011; Tannock *et al.*, 2005).

A combination of comparative genomic and transcriptomic analyses have been used to identify genes that were overexpressed during gut colonization and contributed to host specificity and biofilm formation in rodent strains of *L. reuteri* (Frese *et al.*, 2011, 2013; Schwab *et al.*, 2014; Wilson *et al.*, 2014). Several of these genes (Table 2.1) are predicted to be involved in acid resistance, reflecting the acidic pH in the gastric niche, which varies depending on food loading and emptying (McConnell *et al.*, 2008) from pH 4 and 5.7 in the lumen (Ward and Coates, 1987), and between pH 3.5 and 4 in the forestomach (Ward and Coates, 1987; Gärtner, 2001; Teixeira *et al.*, 2014). In particular, the presence of the urease gene cluster is mostly specific to rodent strains and its expression was induced during colonization of the mouse gut, without contributing to biofilm formation (Frese *et al.*, 2011, 2013). Wilson and colleagues confirmed its induction *in vivo* and showed that the cluster contributed to ecological performance in *Lactobacillus*-free mice (Wilson *et al.*, 2014). In addition, genes encoding glutamate decarboxylase and glutaminase were also found to be overexpressed during stomach colonization (Schwab *et al.*, 2014; Wilson *et al.*, 2014). The *dlt* operon, which contributes to acid resistance through the incorporation of D-alanine

esters into cell wall-associated teichoic acids, is essential for *L. reuteri* colonization of the gastrointestinal tract (Walter *et al.*, 2007). Several other acid resistance mechanisms (glutamate decarboxylase, glutaminase and arginine deaminase) support growth of *L. reuteri* during sour dough fermentation (Su *et al.*, 2011; Teixeira *et al.*, 2014).

Although it is established that gastric acid constitutes a potent barrier to bacterial pathogens (Tennant *et al.*, 2008), little is known about how lactobacilli autochthonous to the stomach overcome this environmental filter. Of the genes predicted to be involved in acid resistance, only the urease cluster and the *dlt* operon have been studied in colonization experiments in mice (Walter *et al.*, 2007; Wilson *et al.*, 2014). However, the mechanisms by which these factors facilitate colonization have not been determined, and other functions, independent of acidity, could explain the importance of these factors *in vivo*. In addition, it is unknown which of the other acid resistance factors present in *L. reuteri* contribute to acid resistance during stomach colonization. The goal of this study was therefore to determine the ecological significance of acid-resistance genes present in *L. reuteri* 100-23 during gut colonization, and to systematically determine to what degree they contribute to tolerance to host gastric acid secretion. To achieve this, we compared the ecological fitness of mutants in ex-germ-free mice treated with omeprazole, a proton-pump inhibitor that raises the pH of the stomach from approximately pH 3 to 5 (depending on food loading) to approximately 6.8 to 7.0 (Betton *et al.*, 1988), with mice that were left untreated.

### 2.3 Results

### 2.3.1 Selection of genes of L. reuteri 100-23 predicted to be involved in acid resistance

Genes selected for this study are listed in Table 2.1 and included: (i) the gene encoding for the  $\alpha$ -subunit of the urease enzyme (*ureC*). This gene cluster, which hydrolyses urea to ammonia, which increases the pH (Cotter and Hill, 2003), is mainly found in rodent strains of *L. reuteri* but is absent in other isolates and is overexpressed

during gut colonization (Frese et al., 2011; 2013; Wilson et al., 2014); (ii) arginine deiminase (Adi), which increases acid resistance by intracellular consumption of protons and ammonia production (Arena et al., 2002; Cotter and Hill, 2003; Rollan et al., 2003; Vrancken et al., 2009; Teixeira et al., 2014); (iii) the glutamate decarboxylase (GadB), which is specific to *L. reuteri* strains isolated from rodents (Frese *et al.*, 2011) and induced in vivo (Wilson et al., 2014), and further implicated in acid resistance during growth in sourdoughs (Su *et al.*, 2011; Teixeira *et al.*, 2014); (iv) the cystathionine  $\gamma$ -lyase (Cgl), which catalyses several reactions transforming compounds such as L-cystine, Lcystathionine, L-homoserine, or L-cysteine (De Angelis et al., 2002; Wang, 2002). Lcysteine is degraded into pyruvate, hydrogen sulfide and ammonia, the latter of which could increase the pH (Wang, 2002); (v) the *dltA* gene, which is involved in D-alanyl esterification of teichoic acids associated with cell walls, is an important colonization factor of L. reuteri (Walter et al., 2007) associated with in vitro acid resistance in L. reuteri (Walter et al., 2007) and other organisms (Boyd et al., 2000; Kristian et al., 2005); and (vi) homologues of a two-component regulatory system consisting of a histidine sensor kinase (*lisK*, lr69622) and response regulator (*lisR*, lr69623), which has been previously shown to be involved in acid response regulation in *Listeria monocytogenes* and *Lactobacillus* acidophilus (Kallipolitis and Ingmer, 2001; Cotter and Hill, 2003; Azcarate-Peril et al., 2005). At the protein level, the LisR and LisK homologues have 76 % and 47 % similarity to the proteins in L. acidophilus, whereas they show less than 32 % similarity to other twocomponent systems (cemAKR, bfrKRT, and lr70529/lr70530) described for L. reuteri 100-23 (Frese *et al.*, 2011; Su and Gänzle, 2014). As a negative control, a mutant with an inactivated high molecular mass surface protein (*lsp* mutant) was included in our studies,

as this adhesin contributes to ecological performance *in vivo* but is not predicted to be involved in acid resistance (Walter *et al.*, 2005).

Gene	Protein	Description	Putative Function	Reason for Study	
<i>lr</i> 70114	UreC	Urease enzyme, α subunit	Acid resistance	Host specific (Frese <i>et al.</i> , 2011; Wilson <i>et al.</i> , 2014), upregulated in biofilms (Frese <i>et al.</i> , 2013) and <i>in vivo</i> (Frese <i>et al.</i> , 2011; Schwab <i>et al.</i> , 2014; Wilson <i>et al.</i> , 2014), and involved in acid resistance and critical for ecological success (Kakimoto <i>et al.</i> , 1990; Cotter and Hill, 2003; Wilson <i>et al.</i> , 2014).	
Ir71325	GadB	Glutamate decarboxylase	Acid resistance	Upregulated in biofilm in <i>L. reuteri</i> 100-23 (Wilson <i>et al.</i> , 2014) and involved in acid resistance (Su et al., 2011; Teixeira et al., 2014).	
Ir69360	Cgl	Cystathionine γ- lyase	Reactive oxygen resistance (Lo <i>et al.</i> , 2009)	Upregulated in biofilm in <i>L. reuteri</i> 100-23 (Frese <i>et al.</i> , 2013) and in acid-adapted <i>B.</i> <i>longum</i> biotype <i>longum</i> (Sánchez <i>et al.</i> , 2007); pathway produces ammonia (Lo <i>et al.</i> , 2009), which may have buffering capacity.	
Ir71377	Adi	Arginine deiminase	Acid resistance	Upregulated in the stomach when compared to the cecum in conventional mice (Schwab <i>et al.</i> , 2014) and involved in acid resistance in <i>L. reuteri</i> 100-23 (Teixeira <i>et al.</i> , 2014).	
Ir69622	LisK	Histidine sensor kinase of two- component regulatory system	Two- component regulatory system	Involved in acid response regulation in <i>Listeria monocytogenes</i> (Kallipolitis and Ingmer, 2001; Cotter and Hill, 2003) and <i>Lactobacillus acidophilus</i> (Azcarate-Peril <i>et al.</i> , 2005).	
Ir69623	LisR	Response regulator of two- component regulatory system	Two- component regulatory system	Involved in acid response regulation in <i>Listeria monocytogenes</i> (Kallipolitis and Ingmer, 2001; Cotter and Hill, 2003) and <i>Lactobacillus acidophilus</i> (Azcarate-Peril <i>et al.</i> , 2005).	
Ir1649- Ir1652	DltA	D-alanylation of lipoteichoic acids	Acid resistance, biofilm formation	Involved in acid resistance in several organisms (Boyd <i>et al.</i> , 2000; Kristian <i>et al.</i> , 2005; Lebeer <i>et al.</i> , 2008) and strongly contributes to ecological performance in <i>L. reuteri</i> 100-23 during gut colonization (Walter <i>et al.</i> , 2007).	

 Table 2.1. Genes selected for functional characterization

# 2.3.2 In vitro characterization of putative acid resistance genes

Isogenic mutants (Table 2.2) of each gene were generated by insertional mutagenesis and compared with the wild- type for survival in simulated gastric juice at pH 1.5 and 2 (Fig. 2.1A–F). Depending on the gene tested, the gastric fluid was supplemented with the substrate necessary for the particular pathway. The analysis revealed that the *ureC*, *adi*, *cgl*, *gadB* and *dlt* mutants were all impaired in their ability to tolerate acidic pH. For the *ureC*, *Cgl*, *gadB* and *dltA* mutants, the inhibitory effect of acidic conditions appeared to be similar at pH 1.5 and 2. Exceptions were the *adi* mutant, which was more impaired at pH 2 (Fig. 2.1B), and the *dlt* mutant, which was considerably more impaired in its survival at pH 1.5 than at pH 2 (Fig. 2.1E). The omission of urea, arginine, glutamic acid or cysteine reduced the survival rates of the wild-type to those of the respective mutants (grown with the substrates), showing that acid resistance is facilitated by these substrates. The two-component system with similarity to LisRK did not contribute to acid resistance (Fig. 2.1F). As expected, the *lsp* mutant was not impaired in survival in gastric juice (data not shown).

Lactobacillus reuteri 100-23	Isolate of rat gastrointestinal tract	Wesney and Tannock (1979)
Lactobacillus reuteri 100-23c	Plasmid-cured derivate of strain 100-23	McCOnnell and colleagues (1991)
Lactobacillus reuteri 100-23 ure C mutant	Urease $\alpha$ -subunit inactivated	Frese and colleagues (2013)
Lactobacillus reuteri 100-23 lsp mutant	Large surface protein inactivated	Walter and colleagues (2005)
Lactobacillus reuteri 100-23 cgl mutant	Cystathionine $\gamma$ -lyase inactivated	Frese and colleagues (2013)
Lactobacillus reuteri 100-23 gadB mutant	Glutamate decarboxylase inactivated	This study
Lactobacillus reuteri 100-23c lisR mutant	Response regulator of two-component regulatory system involved in acid resistance in <i>Listeria monocytogenes</i> and <i>Lactobacillus acidophilus</i>	This study
Lactobacillus reuteri 100-23c lisK mutant	Histidine sensor kinase of two-component regulatory system involved in acid resistance in <i>Listeria monocytogenes</i> and <i>Lactobacillus acidophilus</i>	This study
Lactobacillus reuteri 100-23c dlt mutant	D-alanylation of lipoteichoic acids in the bacterial cell wall inactivated	Walter and colleagues (2007)
Lactobacillus reuteri 100-23c adi mutant	Arginine deiminase inactivated	This study
Escherichia coli EC1000	Contains copy of pVW01 repA gene	Russell and Klaenhammer (2001)



**Fig. 2.1.** Determination of the capacity of genes (Table 2.1) to confer survival under acidic conditions. The viability of strains was determined after incubation in artificial gastric fluid at pH 2 and 1.5 for 6 h at 37°C. Survival of (I) wild-type strains incubated without the substrate of the respective enzyme and (II) mutant strains incubated with the respective substrate are shown relative (%) to that of the wild-type strain incubated with the respective substrate. (A) *ureC*; (B) *adi*; (C) *cgl*; (D) *gadB*; (E) *dltA*; and (F) *lisR* and *lisK*. Because there is no added substrate for the *dltA*, *lisR* and *lisK* genes, only the survival of the mutants was compared with the wild-type. Data are shown as means with standard deviations of triplicate independent experiments (biological replicates).

#### 2.3.3 Importance of acid resistance genes during colonization of the mouse GI tract

The ecological importance of the five genes found to contribute to acid resistance *in vitro* (see above) was subsequently tested via competition experiments of mutant and wild-type strains in germ-free mice that were treated with omeprazole or left untreated. Therefore, for genes that contribute to acid resistance *in vivo*, omeprazole would lead to an increase in competitive fitness of the mutants. Controls received either a sham treatment [containing only the dimethylsulphoxide (DMSO), polyethylene glycol and water used to dissolve the omeprazole] or no treatment (bacteria only). A schematic summary of the experimental design is depicted in Fig. 2.2A.

The analysis revealed that the inactivation of *ureC* had a large impact on the tolerance of *L. reuteri* 100-23c towards host gastric acid secretion (Fig. 2.2B). Without the neutralizing effect of omeprazole, the *ureC* mutant represented around 0.1% of the *L. reuteri* population in the gut after 8 days of colonization. Omeprazole treatment restored the population of the *ureC* mutant to 29.8%  $\pm$  11 of the total lactobacilli population detected in the forestomach and 50.2%  $\pm$  15 in the cecum. The omeprazole solvent, polyethylene glycol, has weak buffering capacity, which likely is responsible for the increase of mutant abundance in mice on the sham treatment (Fig. 2.2B).

The *adi* mutant was only slightly impaired *in vivo*. When in direct competition with the wild-type strain, the mutant represented 26.9%  $\pm$  13 of the total lactobacilli in the forestomach and 35.7%  $\pm$  17 in the cecum. No significant differences were observed for omeprazole or sham treatment (Fig. 2.2C). The inactivation of *gadB* also only led to a slight impairment *in vivo* (Fig. 2.2D), with the mutant comprising 19.5%  $\pm$  11 and 13.0%  $\pm$  10 of the total lactobacilli population in forestomach and in the cecum, respectively.

Omeprazole treatment had no detectable effect in the forestomach (23.5%  $\pm$  14), but significantly enhanced survival in the cecum (31.3%  $\pm$  25).



**Fig. 2.2.** Mouse competition experiment with mutant and wild-type strains in ex-germ-free C57BL/6J mice treated with omeprazole, sham or no treatment. A. Conceptual summary of the experimental design for mouse experiments. Mice were divided into three groups. Group 1 mice were treated daily with omeprazole, whereas group 2 mice were gavaged

with the polyethylene glycol, water and DMSO sham. Group 3 mice served as the control animals and received no treatment. All treatments were administered for 8 days. On day two, all mice received a single gavage with a 1:1 mixture of wild-type and mutant. The proportions of total lactobacilli composed of each mutant in the forestomach and cecum of mice co-inoculated with wild-type and mutant strains were shown. B–G. *In vivo* competition experiment between wild-type and *ureC* mutant (B); wild-type and *adi* mutant (C); wild-type and *gadB* mutant (D); wild- type and *cgl* mutant (E); wild-type and *dltA* mutant (F); wild-type and *lsp* mutant (G). Data are shown as means with standard errors of the mean. Significance of  $P \le 0.05$  is denoted by a single asterisk (\*),  $P \le 0.01$  as two asterisks (\*\*), and  $P \le 0.001$  by three asterisks (\*\*\*). Circles and triangles represent the forestomach and cecum, respectively, of a single mouse.

The *cgl* mutant showed a high degree of impairment when competing with the wildtype in both the forestomach (4.8%  $\pm$  3) and the cecum (7.5%  $\pm$  6), demonstrating that this gene is ecologically relevant *in vivo* (Fig. 2.2E). Similar findings were obtained for the *dlt* mutant, which was highly impaired (Fig. 2.2F), an observation consistent with previous findings in *Lactobacillus*-free mice (Walter *et al.*, 2007). In both mutants, omeprazole did not influence the ecological performance. Thus, it appears that both *cgl* and *dltA* encode for ecologically relevant colonization factors that are not involved in providing resistance to gastric acid secretion.

The *lsp* mutant lacks a putative adhesin that is not involved in acid resistance. We included this mutant to determine unspecific effects of omeprazole on the competiveness of mutant strains in general. As shown previously (Walter *et al.*, 2005), the *lsp* mutant was impaired *in vivo*, forming around 10% of the population. However, as expected, no difference between the three treatments was observed (Fig. 2.2G).

Altogether, these experiments demonstrate that the urease gene cluster is the only factor that mediates resistance against gastric acid secretion in *L. reuteri* 100-23 during stomach colonization, and that no other acid resistance factor was able to compensate for its loss under the given experimental and dietary conditions.

### 2.3.4 Urease activity is regulated by pH

The *in vivo* competition experiments demonstrated the importance of the urease gene cluster as an acid-related colonization factor for *L. reuteri* 100-23 in the rodent forestomach. To characterize the regulation of this cluster, the wild-type strain was grown

in mMRS broth supplemented with 1% urea and growth and urease activity was monitored for 24 h (Fig. 2.3A). The addition of urea caused a slightly decreased growth rate, but led to a rapid increase in pH after 12 h of incubation. The final pH after 24 h was 7.4 with urea supplementation, compared with pH 3.9 without urea in the media (P < 0.0001). This alkalization of the supernatant was not observed for the *ureC* mutant (data not shown). There was no detectable urease activity in the first 8 h. However, urease activity became detectable after 10 h when the pH approached pH 4 (Fig. 2.3A).



**Fig. 2.3.** A. Bacterial growth (OD 600) of *L. reuteri* 100-23 (continuous line, empty symbols, left ordinate axis) and the pH of the media (dotted line, full symbols, right ordinate axis) with (triangle symbol) and without urea (square symbol). B. Urease activity over time in cell lysates of wild-type strain 100-23 grown with and without 1% urea in mMRS media, gray and white bars respectively. C. Urease activity in cell lysates of wild-type strain 100-23 in induction experiment. mMRS media adjusted to pH 4, 5 or 6 with and

without 1% urea supplementation, gray and white bars respectively. n = 3, means and standard deviations are shown. Treatments with different letters (a, b, c) are significantly different from one another ( $P \le 0.05$ ).

Urea supplementation had no significant effect on the urease activity at any time point. These findings suggest that urease activity in *L. reuteri* was not induced by the substrate but by rather acidic conditions. To confirm that induction of urease activity in *L. reuteri* 100-23 occurs via acidity and not urea, cells were grown for 6 h in mMRS, centrifuged and re-suspended in fresh mMRS media adjusted to pH 4, 5 or 6. Cells were incubated for another 2 h before urease activity was measured. Cells after 6 h of growth were used for these experiments as *L. reuteri* did not show any urease activity until 10 h of growth (Fig. 2.3A), allowing the determination of conditions that induce urease activity. This experiment demonstrated that urease activity was induced at pH 4, unaffected at pH 5 and not detectable at pH 6 (P < 0.01). The presence of urea did not enhance urease activity. Transcript analysis revealed that expression of *ureC* was 124 times higher at pH 4 compared with pH 6 independently of the presence of urea. These findings demonstrate that the urease activity in *L. reuteri* is regulated on the transcriptional level.

Two-component systems are commonly used by lactic acid bacteria for environmental sensing and signal transduction and are often involved in the acid stress response (Cotter and Hill, 2003). *Lactobacillus reuteri* 100-23 possess homologues to the LisRK system, which has been implicated in mediating acid resistance in *L. monocytogenes* (Cotter et al., 1999) and *L. acidophilus* (Azcarate-Peril et al., 2005). To test if the *lisRK* genes are involved in regulating urease activity, we compared culture supernatant pH of the *lisK* and *lisR* mutants during growth in the presence of urea. The *lisR* mutant was also tested in the same pH induction experiment described above. Neither the *lisK* nor the *lisR* mutation had an effect on the buffering capacity during growth in the presence of urea, and urease activity was still induced by low pH in the *lisR* mutant in the pH induction experiment (data not shown). Hence, it was concluded that this two-component regulatory system is not involved in the regulation of the urease gene cluster, which is consistent with the finding that the *lisR* and *lisK* mutants were not impaired in simulated gastric juice (Fig. 2.1K). Therefore, it is currently unknown how *L. reuteri* senses acidic pH and induces gene expression of the urease cluster.

# **2.4 Discussion**

The rodent stomach consists of two parts: forestomach and corpus. The forestomach represents about two thirds of the total stomach volume and is lined by a squamous stratified epithelium. The corpus is lined by a glandular and secretory epithelium covered by a mucus layer (Gärtner, 2001) and harbours the H<sup>+</sup> /K<sup>+</sup> proton-pumps responsible for the low pH in the stomach (Fig. 2.2A). *Lactobacillus reuteri* colonizes the forestomach epithelium, but it is also found throughout the digestive tract, including the cecum, where pH values are closer to neutral. However, the spatial patterns of *L. reuteri* populations throughout the mouse digestive tract suggest that cells in the cecum are likely allochthonous to this site and originated from cells colonizing the stomach (Walter, 2008). This notion was supported by the findings of this study; cecal mutant proportions always mirrored those of the forestomach, independently of gene function (Fig. 2.2B–G). This agrees with previous findings concerning the *dlt* (Walter *et al.*, 2007), *gtfA*,

87

*inu*, (Walter *et al.*, 2008), *lsp*, *msrB* (Walter *et al.*, 2005) and *ftf* (Sims *et al.*, 2011) mutants. The forestomach of mice is therefore the primary habitat of *L. reuteri*, which makes acid resistance a key factor for successful colonization.

Accordingly, several pathways and factors have been identified and functionally characterized to contribute to acid resistance in L. reuteri (Fig. 2.4). However, our experiments in omeprazole-treated mice identified the urease gene cluster as the predominant factor necessary for L. reuteri 100-23 to tolerate host gastric acidic secretion (Figs 2.2B and 2.4A). Inactivation of the *ureC* gene resulted in the lowest levels of colonization (around 0.1%) of all mutants tested here. This finding, consistent with observations in *Lactobacillus*-free mice (Wilson *et al.*, 2014), demonstrated the paramount ecological importance of the urease cluster. Restoration of mutant proportions to around 30% and to 50% with omeprazole in the forestomach and cecum, respectively, indicated that host acid secretion is the main ecological factor decreasing mutant levels, and that urease production of L. reuteri is almost completely devoted towards resistance to host gastric acid production. Furthermore, the percentage of mutant strains was significantly lower in the forestomach in omeprazole treated mice compared with the cecum. One could speculate that *ureC* has a residual function in the forestomach that is unrelated to host acid production. Instead, *ureC* may contribute to resistance against the build-up of acidic metabolic end-products in the biofilm generated through bacterial fermentation. Overall, our findings show that host acid secretion exerts a substantial selective pressure on the L. reuteri population, even in the non-secretory forestomach, and that urease production serves as an adaptive phenotype to overcome this pressure.



Fig. 2.4. Overview of metabolic pathways of genes assessed in this study. A. The urease gene converts urea to ammonia and CO<sub>2</sub>. Ammonia is exported from the cytoplasm using the UreI transporter, thus buffering the cell from its surrounding environment. B. Glutamate is imported into the cell by an antiport system and converted to GABA using the glutamate decarboxylase pathway while generating a  $\Delta \Psi$  and  $\Delta pH$ . H<sup>+</sup> consumption raises the intracellular pH. Adapted from Su and colleagues (2011) and Price and colleagues (2012). C. L-arginine is imported using an L-arginine–ornithine antiporter and converted into citrulline and ammonia by the arginine deiminase enzyme. Citrulline is further catalysed to ornithine and ammonia, while consuming H<sup>+</sup>. Ammonia is exported from the cytoplasm potentially using the UreI transporter. D. D-Alanine is coupled to a DltC carrier protein, exported across the cytoplasmic cell membrane and used for esterification of teichoic acids associated with the cell wall. This esterification results in a positive charge of the cell wall. Adapted and simplified from Peschel and colleagues (1999). E. Cysteine is converted to ammonia, hydrogen sulfide and pyruvate by the cystathionine  $\gamma$ -lyase. Ammonia is exported from the cytoplasm using the potentially via the UreI transporter, thus buffering the cell from its surrounding environment. F.  $F_1$  - $F_0$  -

ATPase-producing ATP using a  $\Delta \Psi$  and  $\Delta pH$ , which is generated, in part, by the glutamate decarboxylase pathway. Dashed arrows indicate that more than one step is involved in the pathway.

Although the other four genes (gadB, cgl, adi and dltA) evaluated here also contributed to both survival in the in vitro gastric model and ecological performance in mice, our findings indicate that they do not contribute to tolerance of host gastric acid secretion during forestomach colonization. Mutants for two of the genes, gadB and adi, were only marginally impaired, comprising >20% of the population in competition experiments. The gadB gene has been identified as the most important mechanism of acid resistance in *Escherichia coli* (Feehily and Karatzas, 2013) and was previously shown to contribute to acid resistance (Teixeira et al., 2014) and ecological performance of L. reuteri 100-23 during growth in sourdough (Su et al., 2011). In our experiments, omeprazole treatment did lead to small but significantly higher levels of the gadB mutant in the cecum, suggesting that the gene contributed to acid survival during transit into the cecum but not the forestomach. These findings may be explained by GadB of L. reuteri being active primarily at pH 2.5 (Teixeira et al., 2014), a feature that may confer a survival benefit when the stomach lumen becomes very acidic. Expression of *gadB* is required for the conversion of glutamate to  $CO_2$  and g-aminobutyric acid (GABA) (Su *et al.*, 2011); this function is independent of its role in acid resistance. When glutamate is exchanged with GABA by an antiporter- system, a  $\Delta\Psi$  and  $\Delta pH$  are generated. This proton motive force generated by GadB provides a mechanism for conserving ATP that would otherwise be required to fuel the F<sub>1</sub> F<sub>0</sub> ATPase (Su et al., 2011) (Fig. 2.4B and F). The loss of the proton motive

force may also explain the impairment of this mutant during transit because less energy may be available for the cells to launch a stress response towards the acidic conditions in the stomach lumen.

The arginine deiminase pathway is widely distributed among bacteria (Senouci-Rezkallah *et al.*, 2011), triggered in *L. reuteri* CRL 1098 by low pH (Rollan *et al.*, 2003), and over-expressed in the *Lactobacillus* population colonizing the stomach of conventional mice when compared with the cecum (Schwab *et al.*, 2014). This pathway consumes intracellular protons and raises the cytoplasmatic pH when converting L-arginine and H<sub>2</sub>O to ammonia and citrulline, which is further catalyzed to ornithine, ammonia and CO<sub>2</sub> (Konings, 2002; Teixeira *et al.*, 2014) (Fig. 2.4C). *In vivo*, however, the *adi* mutant was only marginally impaired, and the gene did not confer resistance against host acid secretion.

In contrast to the *gadB* and *adi* mutants, the ecological performance of the *dlt* and *cgl* mutants was substantially impaired *in vivo*. Contrary to the consistent involvement of the *dltA* gene in acid resistance of *L. reuteri* and other organisms *in vitro* (Boyd *et al.*, 2000; Kristian *et al.*, 2005; Walter *et al.*, 2007; Lebeer *et al.*, 2008), host acid production was not the factor that reduced mutant populations in mice. An alternative function of the *dlt* operon is to increase resistance to cationic antimicrobial peptides by generating a positive net charge of the cell surface (Kristian *et al.*, 2005; Walter *et al.*, 2005; Walter *et al.*, 2007). This positive charge leads to a decreased binding of positively charged antimicrobial peptides, e.g. defensins, which may result in increased cell lysis and impaired ecological performance *in vivo* (Walter *et al.*, 2007) (Fig. 2.4D). A recent study showed that a reduction of the negative cell surface charge through Lipid A dephosphorylation mediates resistance to

antimicrobial peptides in the Gram-negative Bacteroides thetaiotaomicron (Cullen *et al.*, 2015). The *dlt* operon could have a similar function in *L. reuteri*.

Although bacterial cystathionine  $\gamma$ -lyases have not been associated with acid resistance, the *cgl* mutant of *L. reuteri* 100-23c was impaired in the *in vitro* acid resistance assays (Fig. 2.1C). Among other reactions, these enzymes catalyze the transformation of L-cysteine and water to hydrogen sulfide, pyruvate and NH4<sup>+</sup> (Wang, 2002; Lo *et al.*, 2009) (Fig. 2.4E). Although expression of the *cgl* gene is upregulated in *L. reuteri* 100-23 growing in biofilms *in vitro* (Frese *et al.*, 2013) and in acid-adapted *B. longum* subsp. *longum* (Sánchez *et al.*, 2007), our mouse experiments did not support a role for *cgl* in overcoming gastric acid secretion. In *L. reuteri* BR11, this pathway was shown to improve oxidative stress defense and is required for thiol production (Lo *et al.*, 2009); it could be important during forestomach colonization. This study establishes the *cgl* gene as an important colonization factor of *L. reuteri* 100-23, but further research is needed to elucidate the mechanism by which this gene contributes to gut colonization.

Together with our previous phylogenetic and comparative genomic studies on *L. reuteri* (Oh *et al.*, 2010; Frese *et al.*, 2011), this work provides novel insight into the ecology and evolution of a vertebrate gut symbiont, and the mechanisms by which a hostspecific lifestyle can emerge. Urease is commonly used by bacteria from different phyla to tolerate stomach acidity (e.g. in Helicobacter pylori) and in some pathogens (e.g. Clostridium perfringens and Yersinina enterocolitica), urease is considered a virulence factor that facilitates survival during gastric transit (Mora and Arioli, 2014). *Lactobacillus reuteri* has acquired the urease cluster, which is extremely rare in the genus *Lactobacillus* (Zheng *et al.*, 2015), by horizontal gene transfer (Frese *et al.*, 2011). The cluster has then

been stably maintained within rodent lineages of the species (Walter et al., 2011). The findings presented here now provide an explanation for the conservation of this trait among rodent strains – it constitutes an essential colonization factor that provides a key adaptation to the gastric niche in rodents. During the evolutionary process, it appears that L. reuteri has tailored transcriptional regulation of the cluster towards the environmental conditions of the murine stomach. Transcriptional expression of the urease cluster is strictly regulated by pH (Fig. 2.3), allowing the organism to respond to the variation in gastric pH and only produce urease when the habitat becomes too acidic. Urea, in contrast, is always present as it enters the stomach by diffusion and through the saliva (Burne and Chen, 2000), and it was therefore not required for L. reuteri to evolve a mechanism of substrate induction. Substrate availability through the rodent host is also a likely reason why urease formation evolved to become more important than GadB and Adi, as the latters' substrates (glutamate and arginine) must be provided in the diet where supply is not reliable. Urea hydrolysis is therefore a key facet of host adaptation (and potentially even co-evolution) in the L. reuteri-rodent symbiosis, and the absence of the phenotype in most non-rodent strains (Walter *et al.*, 2011) is likely an important reason for their low ecological performance in the mouse GI tract (Oh et al., 2010; Frese et al., 2011).

In conclusion, the findings obtained during this study demonstrated that urease production is essential and sufficient for *L. reuteri* 100-23 to cope with host gastric acid secretion. Other genes, such as *adi*, *clg*, *dltA* and *gadB*, and genes encoding for glutaminase [which were overexpressed in acid resistance tests *in vitro* (Teixeira *et al.*, 2014) and in the forestomach (Wilson *et al.*, 2014) but were not studied here due to the presence of several copies in the genome] might contribute to resistance against acidic bacterial metabolic end-

products, or may become more important in a different dietary context. However, in the experiments conducted here, none of these genes was able to complement the loss of *ureC* in mediating resistance to host gastric acid secretion, which appears to exert a major selective pressure. This study provides a better understanding of the phenotypic adaptations of vertebrate gut symbionts that contribute to both a highly successful lifestyle and specialization towards a particular host. Most importantly, it demonstrates that gene annotations and *in vitro* tests have limitations to predict the exact ecological functions of colonization factors of bacterial gut symbionts.

# 2.5 Experimental procedures ethics statement

All mouse experiments were approved by the Institutional Animal Care and Use Committee of the University of Nebraska (Project ID 731).

# 2.5.1 Bacterial strains and media used in the study

All strains of *L. reuteri* and *E. coli* are listed in Table 2. Lactobacilli were grown anaerobically at 37°C in de Man, Rogosa and Sharpe (MRS) medium medium (Difco<sup>TM</sup>; Le point-de-Claix, France) supplemented with 10 g l<sup>-1</sup> maltose and 5 g l<sup>-1</sup> fructose (referred to as mMRS). For gene inactivation in *L. reuteri* 100-23c (plasmid-free derivative of strain 100-23), *E. coli* EC1000 was used as a cloning vector and grown aerobically in Luria– Bertani media (Difco<sup>TM</sup>; Sparks, MD, USA) at 37°C. Erythromycin (200 µgml<sup>-1</sup> for *E. coli*, 5 µgml–1 for lactobacilli), kanamycin (40 µgml<sup>-1</sup> for *E. coli*) and chloramphenicol (7.5 µgml<sup>-1</sup> for lactobacilli) were used for the propagation of recombinant strains.
# 2.5.2 Determination of genes predicted to be involved in acid resistance of *L. reuteri* 100-23

Several different approaches were used to select genes of interest for this study. First, we identified genes that were specific to rodent strains of *L. reuteri* (Frese *et al.*, 2011) and predicted to be involved in acid resistance. Second, putative acid resistance genes that were upregulated *in vivo* compared with *in vitro* cultures were identified (Frese *et al.*, 2013; Wilson *et al.*, 2014). Third, genes coding for metabolic path- ways that produce ammonia (Lo *et al.*, 2009) and two- component systems involved in acid resistance in other bacteria were also considered (Kallipolitis and Ingmer, 2001; Cotter and Hill, 2003). One additional criterion for the selection of genes was that the gene had to be a single copy gene to generate the knock-out mutants according to the method described by Walter and colleagues (2005).

# 2.5.3 Derivation of mutants

Genes of interest were inactivated by insertional mutagenesis by site-specific integration of the plasmid pORI28 into the target sites in the *L. reuteri* 100-23c genome (Walter *et al.*, 2005). Internal regions of the genes of interest were amplified using the primers in Table S1 for each mutant. Each knockout mutation was confirmed by polymerase chain reaction (PCR) using primers flanking the target region of each gene. Strains were routinely maintained in mMRS medium containing 5  $\mu$ gml<sup>-1</sup> of erythromycin, unless the mutant was used for the *in vitro* acid survival assay. Growth curves showed no growth impairments in any of the mutants (data not shown).

#### 2.5.4 In vitro acid survival assay

To simulate the acidic conditions in the mouse stomach, an artificial gastric fluid developed by Cotter and colleagues (2001) was used. The experiment was performed with wild- type L. reuteri 100-23c and all mutants; bacterial survival was monitored over time. To evaluate acid resistance, lactobacilli were grown in mMRS (pH 6.5) for 12-16 h, harvested by centrifugation and washed in PBS. Pre-warmed gastric fluid was adjusted to pH 1.5 and 2 with HCl, and inoculated with approximately 108 cells ml<sup>-1</sup>. Samples were incubated at 37°C and quantified by serial plating after 0, 1, 2, 3 and 6 h. To assess the importance of the *ureC*, *adi*, *gadB* and *LisK/R* genes, assays were performed in gastric fluid with and without 1% urea, 20 mM arginine, 20 mM glutamic acid or 20 mM cysteine respectively. The role of *lsp* and *dlt* genes was assessed without supplementation of the gastric fluid. To gain an insight on the effect of gene inactivation on survival and to allow for a better comparison between experiments, cell numbers of the wild-type strain plus supplement was set to 100%. Based on that value the cell numbers of wild-type without the respective substrate and mutant strains with the substrate was expressed as percent of those obtained with the wild-type incubated with the substrate. Experiments were done in triplicate of biological replicates.

#### 2.5.5 Determination of genes' role in *in vivo* acid resistance

Germ-free C57BL/6J mice (males and females) were bred and reared in flexible film isolators and maintained under gnotobiotic conditions at the University of Nebraska-Lincoln and were randomly assigned to one of three treatment groups. Mice in group 1 received a daily oral gavage of 400  $\mu$ mol of omeprazole kg<sup>-1</sup> (6-methoxy-2-[(4-methoxy-3,5- dimethylpyridin-2-yl)methanesulfinyl]-1H-1,3-benzodiazole; Sigma) for 8 days (Tennant et al., 2008). Omeprazole was dissolved in 50 µl of a DMSO-polyethylene glycol solution (90% DMSO, 4.5% polyethylene glycol and 5.5% water) and was filter sterilized (Zavros et al., 2002). Mice in group 2 were orally gavaged daily with the DMSOpolyethylene glycol vehicle and otherwise treated the same way as group 1 animals. Mice in group 3 did not receive any treatment. On day two, each mouse was inoculated with  $10^6$ cells in a 1:1 ratio of 100-23c wild-type and a mutant strains in a single oral gavage. The inoculum was also plated on mMRS plates with and without erythromycin (5  $\mu$ gml<sup>-1</sup>) to confirm equal representation of the two strains. Mice had access to food and water ad libitum. After 8 days, mice were euthanized and forestomach and cecum contents were serially diluted and plated on mMRS with and without erythromycin (5 µgml<sup>-1</sup>) to determine the ratio of the wild-type and mutant strains in the samples. A total of 6-11 mice per each group (omeprazole, sham, control) were used per experiment. The experiment was repeated twice with the gadB mutant because the first experiment showed a trend towards a higher survival rate due to omeprazole treatment compared with the sham in the forestomach. However, this tendency was not confirmed (Fig. 2.2D). It should be noted that polyethylene glycol possesses weak buffering capacity, which may therefore impact acid exposure to the lactobacilli. Therefore, the amount of solution was kept as low as possible (50 µl total). For all analysed gene clusters, it was assumed that the corresponding substrates, i.e. glutamic acid, arginine, urea etc., were present in the forestomach as they are supplied by the diet, or in the case of urea, enter the stomach by diffusion (Burne and Chen, 2000).

#### 2.5.6 Determination of pH regulation of urease activity

*Lactobacillus reuteri* 100-23c was grown for 6 h in mMRS at 37°C, centrifuged and re-suspended in fresh mMRS media adjusted to pH 4, 5 or 6 with HCl (before sterile filtration). Cells were incubated for another 2 h at 37°C, and subsequently 10 ml of *L*. *reuteri* 100-23c culture was centrifuged for 5 min at 15 000 × g and stored in 10% glycerol at -20°C until determination of urease activity.

# 2.5.7 Measurement of urease activity

Cell solutions were thawed on ice, washed twice with citrate buffer (pH 4) and disrupted with 0.3 g sterile silica beads (0.5 mm) at maximum speed in a cell mill (Mini-Beadbeater Biospec product) for three 1 min intervals. Tubes were cooled on ice for 2 min between intervals to prevent overheating. Samples were centrifuged at 10 000  $\times$  g for 2 min. Supernatant was collected and stored at  $-20^{\circ}$ C until the assays were performed. Urease activity in the supernatant was determined by conversion of urea to ammonia, as described previously (Chaney and Marbach, 1962). Citrate buffer (pH 4) containing 167 mM urea was mixed in equal volumes with cell supernatant and incubated at 30°C for 30 min. Ammonia was quantified by the Berthelot reaction (Chaney and Marbach, 1962). To determine protein, cell pellets were washed twice with 10 mM Tris (pH 8) and disrupted as described above. Protein concentration was determined according to Lowry and colleagues (1951). Urease activity is expressed as microgram of ammonia formed per microgram of protein.

#### 2.5.8 RNA extraction from L. reuteri cell cultures

Lactobacillus reuteri 100-23 was grown for 6 h in mMRS media at 37°C, and cells were collected by centrifugation for 10 min at  $3214 \times g$  and re-suspended in fresh mMRS media adjusted to pH 4, 5 or 6 with HCl. Cells were incubated for another 30 min at 37°C, and subsequently mixed with RNAprotect bacterial reagent (Qiagen, Valencia, CA, USA) at a ratio of 1 to 5. The solution was incubated for 5 min at room temperature, centrifuged and stored at -80°C until used for RNA isolation. Total RNA was isolated after the cell pellet was washed with RNase-free PBS buffer and re-suspended in 100 µl of lysis buffer (30 mM Tris-HCl; 1 mM EDTA, pH 8.0; 15 mg ml<sup>-1</sup> lysozyme; 10 U ml-1 mutanolysin; and 100  $\mu$ gml<sup>-1</sup> Proteinase K). Samples were treated as previously described (Rattanaprasert et al., 2014) and subsequently transferred to an RNeasy Mini spin column (Qiagen, Hilden, Germany). Mixtures were centrifuged for 15 s at 14 000  $\times$  g and the eluate discarded. 350 µl of Buffer RW1 was added and centrifuged as before. There was 80 µl of DNase I incubation mix applied to the RNeasy column and incubated at room temperature for 15 min. And, 350 µl of RW1 buffer was added and centrifuged as described above. The flow-through was discarded, 500 µl of Buffer RPE added and centrifuged. 500  $\mu$ l of Buffer RPE was added again and centrifuged for 2 min at 14 000  $\times$ g. RNeasy column was placed in a new 2.0 ml collection tube and centrifuged for 1 min at 14 000  $\times$  g. RNeasy column was placed in a new 1.5 ml collection tube and RNA eluted with 50  $\mu$ l of RNase-free water. Samples were centrifuged for 1 min at 12 000  $\times$  g. According to the manufacturer's protocol (Applied Biosystems/Ambion, Austin, TX, USA) the purified RNAwas subsequently treated with the TURBO DNA-free kit. RNA was quantified using the Qubit® RNABRAssay kit (Invitrogen, Carlsbad, CA, USA), and RNA integrity was validated on a 1% agarose gel. The absence of DNA contamination was confirmed by real-time PCR.

# 2.5.9 Determination of gene expression by quantitative reverse transcription PCR (qRT-PCR)

The purified RNA was reverse transcribed using the SuperScript<sup>®</sup> VILO<sup>TM</sup> cDNA Synthesis Kit (Invitrogen) according to the manufacturer's instructions with minor modifications as described by Frese and colleagues (2013). qRT PCR was performed using an Eppendorf Mastercycler Realplex2 machine (Eppendorf AG, Hamburg, Germany) and Quanti-Fast SYBR Green PCR kits (Qiagen, Valencia, CA, USA). The ureC and glyceraldehyde 3-phosphate dehydrogenase primers (Table S1) were previously validated using serial 10-fold dilutions of pooled cDNA to determine specificity and efficiency (Frese et al., 2013). For each 25 µl qRT-PCR reaction, 12.5 µl of 2x Quantifast SYBR Green Mastermix, 1 µl of cDNA and 10 µMol of each primer were used. The DNA was denatured at 95°C for 5 min and followed by 40 two-step cycles of 10 s at 95°C, then 30 s at 60°C. Each PCR product was validated on an agarose gel and by inspection of their melting curves. Gene transcripts of the urease  $\alpha$ -subunit were quantified relative to the glyceraldehyde-3-phosphate dehydrogenase housekeeping relative gene, and quantification was performed using the method by Pfaffl (2001).

# 2.5.10 Statistical analysis

Data are expressed as means  $\pm$  standard deviations unless otherwise stated. Statistical analyses were carried out using GRAPHPAD PRISM 5 (GraphPad Software, California, USA). If only two groups were compared, Student's t-tests were per- formed. ANOVA and Tukey's post-tests were used for multiple comparisons. Significance of P  $\leq$ 0.05 is denoted by a single asterisk (\*), P  $\leq$  0.01 as two asterisks (\*\*), and P  $\leq$  0.001 by three asterisks (\*\*\*).

# 2.6 Acknowledgements

This study was funded in part by BioGaia AB. We thank Robert Schmaltz, Brandon White and the staff at the UNL Gnotobiotic Mouse Facility for technical expertise and skillful animal husbandry. JW thanks Michael Gänzle (University of Alberta, Canada) for helpful discussions and critical reading of the manuscript.

#### **2.7 References**

Arena, M.E., Manca de Nadra, M.C., and Muñoz, R. (2002) The arginine deiminase pathway in the wine lactic acid bacterium *Lactobacillus hilgardii* X1B: structural and functional study of the arcABC genes. *Gene* 301: 61–66.

Azcarate-Peril, M.A., McAuliffe, O., Altermann, E., Lick, S., Russell, W.M., and Klaenhammer, T.R. (2005) Microarray analysis of a two-component regulatory system involved in acid resistance and proteolytic activity in *Lactobacillus acidophilus*. *Appl Environ Microbiol* 71: 5794–5804.

Betton, G.R., Dormer, C.S., Wells, T., Pert, P., Price, C.A., and Buckley, P. (1988) Gastric ECL-cell hyperplasia and carcinoids in rodents following chronic administration of H2-antagonists SK&F 93479 and oxmetidine and omeprazole. *Toxicol Pathol* 16: 288–298.

Boyd, D.A., Cvitkovitch, D.G., Bleiweis, A.S., Kiriukhin, M.Y., Debabov, D.V., Neuhaus, F.C., et al. (2000) Defects in d-alanyl-lipoteichoic acid synthesis in *Streptococcus mutans* results in acid sensitivity. *J Bacteriol* 182: 6055–6065.

Burne, R.A., and Chen, Y.Y. (2000) Bacterial ureases in infectious diseases. *Microbes Infect* 2: 533–542.

Chaney, A.L., and Marbach, E.P. (1962) Modified reagents for determination of urea and ammonia. *Clin Chem* 8: 130–132.

Cotter, P.D., and Hill, C. (2003) Surviving the acid test: responses of gram-positive bacteria to low pH. *Microbiol Mol Biol* Rev 67: 429–453.

Cotter, P.D., Emerson, N., Gahan, C.G.M., and Hill, C. (1999) Identification and disruption of lisRK, a genetic locus encoding a two-component signal transduction system involved in stress tolerance and virulence in *Listeria monocytogenes*. *J Bacteriol* 181: 6840–6843.

Cotter, P.D., Gahan, C.G.M., and Hill, C. (2001) A glutamate decarboxylase system protects *Listeria monocytogenes* in gastric fluid. *Mol Microbiol* 40: 465–475.

Cullen, T.W., Schofield, W.B., Barry, N.A., Putnam, E.E., Rundell, E.A., Trent, M.S., et al. (2015) Antimicrobial peptide resistance mediates resilience of prominent gut commensals during inflammation. *Science* 347: 170–175.

De Angelis, M., Curtin, A.C., McSweeney, P.L.H., Faccia, M., and Gobbetti, M. (2002) *Lactobacillus reuteri* DSM 20016: purification and characterization of a cystathionine  $\gamma$ lyase and use as adjunct starter in cheese making. *J Dairy Res* 69: 255–267.

Feehily, C., and Karatzas, K.A.G. (2013) Role of glutamate metabolism in bacterial responses towards acid and other stresses. *J Appl Microbiol* 114: 11–24.

Frese, S.A., Benson, A.K., Tannock, G.W., Loach, D.M., Kim, J., Zhang, M., et al. (2011)The evolution of host specialization in the vertebrate gut symbiont *Lactobacillus reuteri*.*PLoS Genet* 7: e1001314.

Frese, S.A., Mackenzie, D.A., Peterson, D.A., Schmaltz, R., Fangman, T., Zhou, Y., et al. (2013) Molecular characterization of host-specific biofilm formation in a vertebrate gut symbiont. *PLoS Genet* 9: e1004057.

Gärtner, K. (2001) The forestomach of rats and mice, an effective device supporting digestive metabolism in muridae (Review). *J Exp Anim Sci* 42: 1–20.

Kakimoto, S., Sumino, Y., Kawahara, K., Yamazaki, E., and Nakatsui, I. (1990) Purification and characterization of acid urease from *Lactobacillus fermentum*. Appl Microbiol Biotechnol 32: 538–543.

Kallipolitis, B.H., and Ingmer, H. (2001) *Listeria monocytogenes* response regulators important for stress tolerance and pathogenesis. *FEMS Microbiol Lett* 204: 111–115.

Konings, W. (2002) The cell membrane and the struggle for life of lactic acid bacteria. *Antonie Van Leeuwenhoek* 82: 3–27.

Kristian, S.A., Datta, V., Kansal, R., Fedtke, I., Peschel, A., Gallo, R.L., et al. (2005) dalanylation of teichoic acids promotes group A *Streptococcus* antimicrobial peptide resistance, neutrophil survival, and epithelial cell invasion. *J Bacteriol* 187: 6719–6725.

Lebeer, S., Vanderleyden, J., and De Keersmaecker, S.C.J. (2008) Genes and molecules of lactobacilli supporting probiotic action. *Microbiol Mol Biol Rev* 72: 728–764.

Lo, R., Turner, M.S., Barry, D.G., Sreekumar, R., Walsh, T.P., and Giffard, P.M. (2009) Cystathionine gamma-lyase is a component of cystine-mediated oxidative defense in *Lactobacillus reuteri* BR11. *J Bacteriol* 191: 1827–1837.

Lowry, O.H., Rosebrough, N.J., Farr, L.A., and Randall, R.J. (1951) Protein measurement with the folin phenol reagent. *J Biol Chem* 193: 265–275.

McConnell, E.L., Basit, A.W., and Murdan, S. (2008) Measurements of rat and mouse gastrointestinal pH, fluid and lymphoid tissue, and implications for *in-vivo* experiments. *J Pharm Pharmacol* 60: 63–70.

McConnell, M., Mercer, A., and Tannock, G. (1991) Transfer of plasmid pAMβl between members of the normal microflora inhabiting the murine digestive tract and modification of the plasmid in a *Lactobacillus reuteri host*. *Microb Ecol Health Dis* 4: 343–355.

Mora, D., and Arioli, S. (2014) Microbial urease in health and disease. *PLoS Pathog* 10: e1004472.

O'Callaghan, J., and O'Toole, P. (2013) *Lactobacillus*: Host- microbe relationships. In, Between Pathogenicity and Commensalism. *Berlin, Germany: Springer*, pp. 119–154.

Oh, P.L., Benson, A.K., Peterson, D.A., Patil, P.B., Moriyama, E.N., Roos, S., and Walter, J. (2010) Diversification of the gut symbiont *Lactobacillus reuteri* as a result of host-driven evolution. *ISME J* 4: 377–387.

Peschel, A., Otto, M., Jack, R.W., Kalbacher, H., Jung, G., and Götz, F. (1999) Inactivation of the dlt operon in *Staphylococcus aureus* confers sensitivity to defensins, protegrins, and other antimicrobial peptides. *J Biol Chem* 274: 8405–8410.

Pfaffl, M.W. (2001) A new mathematical model for relative quantification in real-time RT-PCR. *Nucleic Acids Res* 29: e45.

Price, C.E., Zeyniyev, A., Kuipers, O.P., and Kok, J. (2012) From meadows to milk to mucosa – adaptation of *Streptococcus* and *Lactococcus* species to their nutritional environments. *FEMS Microbiol Rev* 36: 949–971.

Rattanaprasert, M., Roos, S., Hutkins, R.W., and Walter, J. (2014) Quantitative evaluation of synbiotic strategies to improve persistence and metabolic activity of *Lactobacillus reuteri* DSM 17938 in the human gastrointestinal tract. *J Funct Foods* 10: 85–94.

Rollan, G., Lorca, G.L., and Font de Valdez, G. (2003) Arginine catabolism and acid tolerance response in *Lactobacillus reuteri* isolated from sourdough. *Food Microbiol* 20: 313–319.

Russell, W.M., and Klaenhammer, T.R. (2001) Efficient system for directed integration into the *Lactobacillus acidophilus* and *Lactobacillus gasseri* chromosomes via homologous recombination. *Appl Environ Microbiol* 67: 4361–4364.

Sánchez, B., Champomier-Vergès, M.-C., Collado, M.D.C., Anglade, P., Baraige, F., Sanz, Y., *et al.* (2007) Low-pH adaptation and the acid tolerance response of *Bifidobacterium longum* biotype *longum*. *Appl Environ Microbiol* 73: 6450–6459.

Schwab, C., Tveit, A.T., Schleper, C., and Urich, T. (2014) Gene expression of lactobacilli in murine forestomach biofilms. *Microb Biotechnol* 7: 347–359.

Sekirov, I., Russell, S.L., Antunes, L.C.M., and Finlay, B.B. (2010) Gut microbiota in health and disease. *Physiol Rev* 90: 859–904.

Senouci-Rezkallah, K., Schmitt, P., and Jobin, M.P. (2011) Amino acids improve acid tolerance and internal pH maintenance in *Bacillus cereus* ATCC14579 strain. *Food Microbiol* 28: 364–372.

Sims, I.M., Frese, S.A., Walter, J., Loach, D., Wilson, M., Appleyard, K., et al. (2011) Structure and functions of exopolysaccharide produced by gut commensal *Lactobacillus reuteri* 100-23. *ISME J* 5: 1115–1124. Su, M.S., and Gänzle, M.G. (2014) Novel two-component regulatory systems play a role in biofilm formation of *Lactobacillus reuteri* rodent isolate 100-23. *Microbiology* 160 (Pt 4): 795–806.

Su, M.S., Schlicht, S., and Gänzle, M.G. (2011) Contribution of glutamate decarboxylase in *Lactobacillus reuteri* to acid resistance and persistence in sourdough fermentation. *Microb Cell Fact* 10: S8.

Tannock, G.W., Ghazally, S., Walter, J., Brooks, H., Cook, G., Surette, M., et al. (2005) Ecological behavior of *Lactobacillus reuteri* 100-23 is affected by mutation of the luxS gene. *Appl Environ Microbiol* 71: 8419–8425.

Teixeira, J.S., Seeras, A., Sanchez-Maldonado, A.F., Zhang, C., Su, M.S.W., and Gänzle, M.G. (2014) Glutamine, glutamate, and arginine-based acid resistance in *Lactobacillus reuteri*. *Food Microbiol* 42: 172–180.

Tennant, S.M., Hartland, E.L., Phumoonna, T., Lyras, D., Rood, J.I., Robins-Browne, R.M., and Van Driel, I.R. (2008) Influence of gastric acid on susceptibility to infection with ingested bacterial pathogens. *Infect Immun* 76: 639–645.

Vrancken, G., Rimaux, T., Wouters, D., Leroy, F., and De Vuyst, L. (2009) The arginine deiminase pathway of *Lactobacillus fermentum* IMDO 130101 responds to growth under stress conditions of both temperature and salt. *Food Microbiol* 26: 720–727.

Walter, J. (2008) Ecological role of lactobacilli in the gastro- intestinal tract: implications for fundamental and biomedical research. *Appl Environ Microbiol* 74: 4985–4996.

Walter, J., Chagnaud, P., Tannock, G.W., Loach, D.M., Dal Bello, F., Jenkinson, H.F., et al. (2005) A high- molecular-mass surface protein (Lsp) and methionine sulfoxide reductase B (MsrB) contribute to the ecological performance of Lactobacillus reuteri in the murine gut. *Appl Environ Microbiol* 71: 979–986.

Walter, J., Loach, D.M., Alqumber, M., Rockel, C., Hermann, C., Pfitzenmaier, M., and Tannock, G.W. (2007) D-alanyl ester depletion of teichoic acids in *Lactobacillus reuteri* 100-23 results in impaired colonization of the mouse gastrointestinal tract. *Environ Microbiol* 9: 1750–1760.

Walter, J., Schwab, C., Loach, D.M., Gänzle, M.G., and Tannock, G.W. (2008) Glucosyltransferase A (GtfA) and inulosucrase (Inu) of *Lactobacillus reuteri* TMW1.106 con- tribute to cell aggregation, in vitro biofilm formation, and colonization of the mouse gastrointestinal tract. *Microbiology* 154: 72–80.

Walter, J., Britton, R.A., and Roos, S. (2011) Host-microbial symbiosis in the vertebrate gastrointestinal tract and the *Lactobacillus reuteri* paradigm. *Proc Natl Acad Sci USA* 108 (Suppl.): 4645–4652.

Wang, R. (2002) Two's company, three's a crowd: can H2S be the third endogenous gaseous transmitter? *FASEB J* 16: 1792–1798.

Ward, F.W., and Coates, M.E. (1987) Gastrointestinal pH measurement in rats: influence of the microbial flora, diet and fasting. *Lab Anim* 21: 216–222.

Wesney, E., and Tannock, G.W. (1979) Association of rat, pig, and fowl biotypes of lactobacilli with the stomach of gnotobiotic mice. *Microb Ecol* 5: 35–42.

Wilson, C.M., Loach, D., Lawley, B., Bell, T., Sims, I., O'Toole, P.W., et al. (2014) *Lactobacillus reuteri* 100-23 modulates urea hydrolysis in the murine stomach. *Appl Environ Microbiol* 80: 6104–6113.

Zavros, Y., Rieder, G., Ferguson, A., Samuelson, L.C., and Merchant, J.L. (2002) Genetic or chemical hypochlorhydria is associated with inflammation that modulates parietal and G-cell populations in mice. *Gastroenterology* 122: 119–133.

Zheng, J., Ruan, L., Sun, M., and Gänzle, M. (2015) A genomic view of lactobacilli and pediococci demonstrates that phylogeny matches ecology and physiology. *Appl Environ Microbiol* 81: 7233–7243.

# **Chapter 3**

# *In vivo* selection to identify bacterial strains with enhanced ecological performance in synbiotic applications.

# Preface

This chapter has been previously published: *In Vivo* Selection to identify bacterial strains with enhanced ecological performance in synbiotic applications. Krumbeck, Janina A., María X. Maldonado-Gomez, Inés Martínez, Steven A. Frese, Thomas E. Burkey, Karuna Rasineni, Amanda E. Ramer-Tait, Edward N. Harris, Robert W. Hutkins, and Jens Walter. *In vivo* selection to identify bacterial strains with enhanced ecological performance in synbiotic applications. *Applied and Environmental Microbiology* 81, no. 7 (2015): 2455-2465

#### **3.1 Abstract**

One strategy for enhancing the establishment of probiotic bacteria in the human intestinal tract is via the parallel administration of a prebiotic, which is referred to as a synbiotic. Here we present a novel method that allows a rational selection of putative probiotic strains to be used in synbiotic applications: in vivo selection (IVS). This method consists of isolating candidate probiotic strains from fecal samples following enrichment with the respective prebiotic. To test the potential of IVS, we isolated bifidobacteria from human subjects who consumed increasing doses of galactooligosaccharides (GOS) for 9 weeks. A retrospective analysis of the fecal microbiota of one subject revealed an 8-fold enrichment in Bifidobacterium adolescentis strain IVS-1 during GOS administration. The functionality of GOS to support the establishment of IVS-1 in the gastrointestinal tract was then evaluated in rats administered the bacterial strain alone, the prebiotic alone, or the synbiotic combination. Strain-specific quantitative real-time PCR showed that the addition of GOS increased *B. adolescentis* IVS-1 abundance in the distal intestine by nearly 2 logs compared to rats receiving only the probiotic. Illumina 16S rRNA sequencing not only confirmed the increased establishment of IVS-1 in the intestine but also revealed that the strain was able to outcompete the resident *Bifidobacterium* population when provided with GOS. In conclusion, this study demonstrated that IVS can be used to successfully formulate a synergistic synbiotic that can substantially enhance the establishment and competitiveness of a putative probiotic strain in the gastrointestinal tract.

#### **3.2 Introduction**

The mechanistic role of the gastrointestinal (GI) microbiota and its metabolites in maintaining human health has been well demonstrated (1–3). Gut microbes provide several important benefits for their host, including provision of nutrients, development and maturation of the immune system, and protection against pathogens via colonization resistance (4). However, the gut microbiota may also contribute to obesity, inflammatory and autoimmune diseases, and other chronic disease states (5–7). Such diseases are often associated with compositional alterations in the fecal microbiota, a condition referred to as "dysbiosis" (8). Given that the presence of specific types of bacteria and their relative abundance within the gut are considered to affect host health, there is much interest in devising strategies that modulate gut microbiota composition and potentially redress disease related dysbiotic patterns (9).

Dietary approaches currently available to modulate the gut microbiota include prebiotics (10–12), fermentable fibers (13, 14), probiotics (or live biotherapeutics) (15), and synbiotics, which are a combination of a probiotic and a prebiotic (11, 16). According to Kolida and Gibson (16), synbiotics can be either complementary or synergistic. Complementary synbiotics consist of a probiotic and a prebiotic selected to independently confer benefits to the host. In contrast, synergistic synbiotics are comprised of a prebiotic chosen specifically for the selected probiotic to stimulate its growth, activity, and survival in the gastrointestinal tract (16).

Synergistic synbiotics therefore hold the potential to improve the establishment of a specific bacterial strain when introduced into the gastrointestinal tract. Unfortunately, successful synergistic synbiotic combinations are not well established in the literature despite a large number of studies. To our knowledge, only two reports describe a synbiotic combination in which the prebiotic significantly enhanced the stability, persistence, or metabolic activity of a specific probiotic strain *in vivo* (17–19). As noted by Kolida and Gibson (16), this low success rate may be explained by the selection of most synbiotic combinations on an arbitrary basis, including shelf life, industrial performance, availability, and cost. Indeed, few synbiotic preparations are formulated based on a rational selection of both the prebiotic and the probiotic (12, 16), such as via *in vitro* or *in vivo* screens assessing the ability of the probiotic to utilize the prebiotic (17–21). Even if synbiotic formulations were based on these criteria, synergism between the probiotic strain and the prebiotic was rarely observed in human and animal trials (22–24). These observations suggest that the probiotic strains were unable to utilize the selected prebiotic to expand their populations under the prevailing ecological conditions in the gastrointestinal tract. We therefore propose that synergistic synbiotics are likely to be more successful if selection of the probiotic organism is based on ecological criteria.

In this report, we introduce the concept of *in vivo* selection (IVS) to identify putative probiotic strains with enhanced ecological performance when used in synbiotic applications. The concept consists of isolating putative probiotic strains from fecal or intestinal samples after enriching for them with dietary administration of the prebiotic. We reasoned that such strains would likely be able to successfully utilize the prebiotic *in vivo* within the constraints of the competitive gastrointestinal environment. To test IVS, we isolated bifidobacteria from fecal samples of human individuals who had consumed the prebiotic galactooligosaccharide (GOS) during a previous human trial (25). A combination of approaches was used to select a candidate probiotic strain (*Bifidobacterium adolescentis*)

strain IVS-1) enriched by GOS *in vivo*. We then tested the synergistic potential of this strain and GOS when administered as a synbiotic combination in a rat model of high- fatdiet-induced nonalcoholic fatty liver disease (NAFLD). A NAFLD model without severe inflammatory disease was chosen, as inflammation would potentially confound the ecological analysis due to its effects on gut microbiota composition. Although no direct physiological benefits were observed in the rats, the results from the gut microbiota analysis demonstrated that IVS can be used to select a synergistic synbiotic combination that substantially increases the ecological performance of the bacterial strain *in vivo*.

#### 3.3 Materials and methods

#### 3.3.1 Isolation of *in vivo*-enriched bifidobacteria from humans.

In a previous study (25), fecal samples were collected from subjects who consumed cumulative doses of GOS (0, 2.5, 5, and 10 g per day for 3 weeks each). Throughout the study, fresh fecal samples were collected and immediately plated onto Rogosa LS agar to enumerate bifidobacteria. Bacterial counts were used to identify GOS responders (i.e., individuals who experienced significant increases in numbers of bifidobacteria), and colonies were picked during the period in which 10 g GOS day<sup>-1</sup> was consumed. Colonies were purified by successive liquid and plate cultures, and stock cultures were prepared and stored at -80°C. A total of 28 individual colonies (2 to 3 per subject) were propagated. To classify isolates, DNA was extracted by using the phenol-chloroform extraction method (26), and the 16S rRNA gene was amplified by using the 8F and 1391R universal primers. The amplification product was purified (QIAquick PCR purification kit; Qiagen Inc., MD) and sequenced by a commercial provider (Eurofins MWG Operon, Huntsville, AL).

Identity was determined by comparing sequences to sequences in the GenBank database; species were assigned based on the best match.

#### 3.3.2 In vitro growth on GOS.

Each isolate was screened for its ability to use GOS as a growth substrate in an MRS broth culture. Growth experiments were performed with basal MRS broth containing 2% (wt/vol) glucose or GOS (Purimune; GTCNutrition, Golden, CO). The latter contained 92% GOS, with residual carbohydrates being mainly lactose. Control cultures were therefore also grown on basal MRS broth supplemented with the same amount of lactose as that present in the commercial GOS (giving a final concentration of 0.16% lactose). Cultures were incubated anaerobically at 37°C, and growth was determined by optical density measurement at 600 nm. Strains that grew on GOS to cell densities similar to those on glucose were considered GOS fermenters.

### 3.3.3 Strain-specific primer design and validation.

The genome of *B. adolescentis* IVS-1 was sequenced to draft status by using a standard shotgun library prep kit on a Roche GS FLX sequencer at the former Core for Applied Genomics and Ecology (CAGE) (University of Nebraska, Lincoln, NE). Sequencing resulted in 65,460 reads that were assembled *de novo* by using the gsAssembler (Newbler) module of the GS-FLX Off- Instrument software suite. This resulted in draft sequences of 148 contigs with ~15-fold coverage.

Unique genes in *B. adolescentis* IVS-1 were identified by comparing the annotated genome with other available *B. adolescentis* genomes in the JGI database (using the Phylogenetic Profiler for Single Genes tool in IMG). From this analysis, the clustered regularly interspaced short palindromic repeat (CRISPR)-associated helicase Cas3 was selected as the target gene, and a putative primer pair was designed by using Primer 3 software (27). Candidate primers were evaluated for hairpin and dimer formation by using Netprimer (Premier Biosoft International, Palo Alto, CA). The selected forward (F) primer TTGCTTTTGCTCTGGAACATAC and reverse (R) primer GTAATGAGGTAATACTGCGTCC were validated in silico by performing a BLAST search against the NCBI database. These primers were also validated experimentally by quantitative real-time PCR (qRT- PCR) using DNA from 10 different Bifidobacterium strains related to strain IVS-1 (each having >96% identity at the 16S rRNA gene level). These strains included Bifidobacterium adolescentis ATCC 15703, Bifidobacterium adolescentis L2-32, *Bifidobacterium* longum subsp. longum ATCC 15707. DJO10A, *Bifidobacterium Bifidobacterium* longum longum ATCC 15697. Bifidobacterium longum subsp. longum F8, Bifidobacterium longum subsp. longum JDM301, Bifidobacterium sp. strain 113, Bifidobacterium sp. strain 12\_1\_47BFAA, and *Bifidobacterium* sp. strain HMLN14. Furthermore, to test if primers could select against fecal bacterial communities in both humans and rats, DNA from 23 human fecal samples and 10 Sprague-Dawley rat fecal samples from an independent study were tested. Human fecal materials analyzed included the baseline samples (i.e., before GOS supplementation) from 18 subjects from a previous study by Davis et al. (25) as well as five other human fecal samples from an independent study.

#### **3.3.4 Quantitative real-time PCR.**

qRT-PCR was performed by using a Mastercycler Realplex2 instrument (Eppendorf AG, Hamburg, Germany). Each PCR was performed with 25-μl volumes using real-time master mix containing SYBR(5 Prime Inc., Gaithersburg, MD) and either genus-specific primers for *Bifidobacterium*, F primer TCGCGTC(C/T)G GTGTGAAAG and R primer CACATCCAGC(A/G)TCCAC (25, 26), or the strain-specific primers for *B. adolescentis* IVS-1 (described above), each at a concentration of 0.8 μM. Annealing temperatures of 58°C and 61°C were used for the genus- and strain-specific PCRs, respectively. Standard curves for absolute quantification of bacterial cell numbers were prepared by using cultures of *B. adolescentis* IVS-1 grown overnight (14 h), as described previously (25, 26).

# 3.3.5 Administration of the probiotic, prebiotic, and synbiotic to rats.

A freeze-dried powder of *Bifidobacterium adolescentis* IVS-1 was produced by a contract manufacturer (Culture Systems, Mishawaka, IN). The powder contained  $5 \times 10^{10}$  CFU g<sup>-1</sup> and was stable during the entire course of the study. For delivery to the rats, the powder was suspended in drinking water (double-distilled water) to reach a concentration of  $3 \times 10^7$  cells ml<sup>-1</sup>. GOS was diluted in water at a concentration of 0.033 g ml<sup>-1</sup>, and the synbiotic was prepared by mixing both IVS-1 and GOS in the abovementioned concentrations. All preparations were prepared fresh daily in drinking water for the duration of the experiment. Cell viability and stability were validated by plating samples on MRS medium at different time points. This analysis revealed that IVS-1 was highly

stable in drinking water, with levels dropping <1 log over 24 h. The addition of GOS did not influence the viability of the probiotic in drinking water (data not shown).

#### 3.3.6 Rat study design.

Synergism of the synbiotic preparation was tested in a rat model of NAFLD (28). Four-week-old male Sprague-Dawley rats were obtained from Charles River Laboratories (Wilmington, MA) and acclimated for five days prior to study initiation. All animals were housed in pairs in individually vented cages mounted on a rack with positive airflow. The room environment was maintained at 20°C to 21°C with a 12-h light-dark cycle. Prior to the start of the study, all rats received a standard rat chow and autoclaved, double-distilled water ad libitum during the five day acclimation period. All animal procedures were approved by University of Nebraska—Lincoln IACUC.

Rats were randomly assigned to one of five treatments, with three to six rats per group. Groups one through four were fed a high-fat diet (60%kcal from fat) (AIN-58G9 TestDiet) (see Table 3.S1 in the supplemental material), while group five received a standard diet (12% fat) (AIN-58G7 TestDiet) for eight weeks. After four weeks of feeding, groups were assigned to one of the following supplement treatments. Rats in groups one and five received no additional treatment. Group two rats received drinking water supplemented with 3.3% GOS to give ~1 g of GOS day<sup>-1</sup> rat<sup>-1</sup>. Group three rats were given drinking water supplemented with ~1 x 10<sup>9</sup> CFU of *B. adolescentis* IVS-1 day<sup>-1</sup> rat<sup>-1</sup>. Group four rats received both the GOS and IVS-1 (synbiotic mixture), at the same doses as those given to groups two and three. All treatments were prepared fresh daily and

administered for four weeks. The daily water intake per rat was significantly different among groups and was used to calculate the absolute doses of probiotic cells per day (P =0.001) (see Table 3.S2 in the supplemental material). Rats fed the probiotic drank significantly more water (41.9 ± 8.6 ml) than did rats fed the synbiotic (35.4 x 4.5 ml), resulting in a significantly higher dose of IVS-1 in the probiotic group (1.26 x 109 CFU versus 1.06 x 109 CFU; P = 0.0001). GOS consumption was not significantly different between the prebiotic- and synbiotic-fed groups (P = 0.2063) (see Table 3.S2 in the supplemental material).

Body weights were determined weekly throughout the study. All rats were necropsied after eight weeks of study. Blood, cecum, colon content, liver, and epididymal fat pads were collected, and the cecum and colon content were immediately frozen in liquid nitrogen and stored at -80°C until further use.

#### **3.3.7 Evaluation of host physiological parameters in rats.**

Liver lipid extraction was performed according to methods described previously by Folch and colleagues (29). Aliquots of lipid extract were saponified to quantify triglycerides (TGs) by using the TG diagnostic kit (Thermo dimethyl adipimidate kit; Thermo Electron Clinical Chemistry, Louisville, CO). Data are reported as µg TGmg<sup>-1</sup> (wet weight) liver tissue. To evaluate liver damage, plasma alanine aminotransferase (ALT) and alkaline phosphatase (ALP) enzyme levels were measured, which are indicators of hepatocyte damage/leakage and cholangiocyte stress, respectively (30, 31). Blood was collected into heparinized tubes at necropsy, and ALT and ALP levels were quantified by using a Mammalian Liver Profile rotor in a VetScan VS2 analyzer (Abaxis, Union City, CA). Levels of tumor necrosis factor alpha (TNF- $\alpha$ ) and monocyte chemoattractant protein 1 (MCP-1) were quantified as measures of systemic inflammation by using a Milliplex rat magnetic bead multiplex assay (Merck Millipore, Billerica, MA) according to the manufacturer's protocol.

#### **3.3.8 Illumina 16S RNA sequencing and sequence analysis.**

Colonic and cecal contents were flash-frozen in liquid nitrogen at necropsy, and DNA was extracted as described previously (26), with one modification: the lysis buffer contained 20mM Tris-HCl (pH 8), 2mM EDTA, 1.2% Triton X-100 (pH 8.0), and 20 mg ml<sup>-1</sup> Lysozyme (MP Biomedicals, Solon, OH). Amplicon sequencing of colonic contents was performed by the University of Minnesota Genomics Center, and all samples were sequenced together in the same run. First, theV5-V6region of the 16SrRNA gene was amplified with primer pair 784F (5'-RGGATTAGATACCC-3') and 1064R (5'-CGACRRCCATGCANCACCT-3') in a 25µl PCR mixture containing 5 µl of template DNA, 5µl of 2x HotStarTaqPCRmaster mix, a final concentration of primers of 500 nM, and 0.025 U µl<sup>-1</sup> HotStarTaq polymerase (Qiagen Inc.). Amplification reactions included an initial denaturation step at 95°C for 5 min followed by 20 to 25 cycles of denaturation (50 s at 94°C), annealing (30 s at 40°C), and elongation (30 s at 72°C). Next, samples were diluted 1:100 in water for input into library tailing PCR. The PCR was analogous to the one conducted for initial amplification except for a *Taq* polymerase concentration of 0.25 U  $\mu$ l<sup>-1</sup>, and the PCR conditions consisted of an initial denaturation step at 95°C for 5 min followed by 10 to 15 cycles of denaturation (50 s at 94°C), annealing (30 s at 40°C), and elongation (1 min at 72°C).

PCR products were quantified by using the Quant-iT PicoGreen double-stranded DNA (dsDNA) assay kit (Life Technologies). A subset of the amplicon libraries was spot checked on a Bioanalyzer High-Sensitivity DNA chip (Agilent Technologies, Santa Clara, CA) for correct amplicon size. Next, samples were normalized to 2 nM and pooled. The total volume of the libraries was reduced by the use of a SpeedVac, and amplicons were size selected at 420 bp ± 20% by using the Caliper XT system (PerkinElmer, Waltham, MA). Afterwards, library pools were cleaned with 1.8 x AMPureXP beads (Beckman Coulter, Brea, CA) and eluted in water. The amount of DNA in the final pool was quantified with PicoGreen and normalized to 2 nM for input into the Illumina MiSeq platform (v3 kit) to produce 300-bp paired-end sequencing products. Clustering was done at 10 pM with a 5% spike of PhiX. The generated sequences were quality filtered with Illumina software at the University of Minnesota Genomics Center. Twenty-two of 24 samples met all quality control criteria and were used for the microbial community analysis.

#### 3.3.9 Microbial community analysis.

Reads were trimmed to 240 bp with the FASTX-Toolkit (http://hannonlab.cshl.edu/fastx\_toolkit/), and paired- end reads were merged with the merge-illumina-pairs application (https: //github.com/meren/illumina-utils/) (P value of 0.03, enforced Q30 check, perfect matching to primers, and no ambiguous nucleotides al-

lowed). Files exceeding 30,000 reads were subsampled to this number in Mothur v.1.31.162 to standardize the sequencing depth across samples. Subsequently, USEARCH v7.0.100163 was used to generate operational taxonomic units (OTUs) with a 98% similarity cutoff. OTU generation included the removal of putative chimeras identified against the Gold reference database, in addition to the chimera removal inherent to the OTU clustering step in UPARSE. After quality control and chimera removal, samples contained an average of  $25,718 \pm 941$  sequences. The resulting sequences were also taxonomically characterized from phylum to genus levels with Ribosomal Database Project (RDP) Classifier with the MultiClassifier v1.1 tool. All phylotypes were computed as percent proportions based on the total number of sequences in each sample.

#### **3.3.10** Statistical analysis.

Results are expressed as means  $\pm$  standard deviations (SD) unless otherwise stated. To analyze bacterial composition, diversity differences, and host physiological parameters, one-way analyses of variance (ANOVA) with repeated measures in combination with Tukey's post hoc tests were applied. To achieve normality for data that were not normally distributed, values were subjected to  $log_{10}$  transformations. If only two groups were compared, Student's t tests were performed. Spearman's correlations were used to assess correlations between bacterial groups. To account for type I errors, the false discovery rate was used. A P value of < 0.05 and correlation coefficient (r) values of > 0.60 (in absolute values) were considered significant. Analyses of variance and false discovery rate control were performed by using SAS/STAT (SAS Institute Inc., Cary, NC, USA), while correlations were determined by using GraphPad Prism version 5.0 (GraphPad Software, La Jolla, CA, USA).

# 3.3.11 Nucleotide sequence accession number.

The genome sequence of B. adolescentis IVS-1 has been deposited in the DDBJ/EMBL/GenBank data- base under accession number JRNZ01000000.

# **3.4 Results**

#### 3.4.1 In vivo selection of B. adolescentis IVS-1.

In a previous study (25, 32), we reported a significant and remarkably specific enrichment of *Bifidobacterium* populations in human subjects during dietary supplementation with GOS (as demonstrated by 454 sequencing, genus-specific qRT-PCR, and quantitative culture), which is in agreement with data from other GOS feeding studies (33–38). Cultural enumeration of fecal samples during the human trial allowed us to identify individuals in which bifidobacteria were enriched by GOS and from whom strains likely to utilize GOS *in vivo* could be selected. This novel strategy for selection and recovery of autochthonous strains enriched by a prebiotic is referred to as *in vivo* selection (IVS) (Fig. 3.1A). Using the IVS approach, a total of 28 presumptive bifidobacterial colonies from 11 subjects were isolated and classified by sequencing of the 16S rRNA genes. Eight isolates were classified as *Bifidobacterium adolescentis*, eight were classified as *Bifidobacterium longum*, three were classified as *Bifidobacterium pseudocatenulatum*, and one was classified as *Bifidobacterium bifidum*. Of the remaining isolates, four belonged to the *Coriobacterium* genus, one could be classified only to the family level (*Lachnospiraceae*), and three could not be sequenced due to insufficient growth. All strains resulting in pure cultures were also screened for their ability to ferment GOS during *in vitro* growth, and 13 were classified as GOS fermenters, 12 were classified as non-fermenters, and three could not be propagated to be tested (data not shown). Out of the 13 strains able to ferment GOS, five were classified as *B. longum*, five were classified as *B. adolescentis*, one was classified as *B. bifidum*, one was classified as *B. pseudocatenulatum*, and another one was classified as *Lachnospiraceae*. None of the isolated *Coriobacterium* strains were classified as fermenters.

Based on the culture data, 454 sequencing (32), and the GOS fermentation tests, we selected one strain and designated it IVS-1. This strain originated from a subject who showed a strong bifidogenic response to GOS (Fig. 3.1B). Based on 16S rRNA sequencing, IVS-1 had 98.4% identity (100% query coverage and an E value of zero) with the 16S rRNA gene of *B. adolescentis* ATCC15703T and was therefore allotted to this species. However, the strain belongs to a distinct phylogenetic cluster (*Bifidobacterium* species II cluster) detectable by using the V1-V3 region of the 16S rRNA gene (32). This cluster was significantly enriched by GOS in all subjects, including the individual from whom IVS-1 was isolated (Fig. 3.1C). The ability of *B. adolescentis* IVS-1 to utilize GOS was demonstrated by growth in MRS broth containing 2% GOS (see Fig. 3.S1 in the supplemental material). The established metabolic benefits of the species *B. adolescentis* serve as another rationale for the selection of IVS-1 for future applications (39, 40).



**Fig. 3.1.** *In vivo* selection to identify putative probiotic strains to be used in synbiotic applications. A. Concept of *in vivo* selection. B. Proportion of fecal bifidobacteria in a human individual consuming GOS (included in chews) in four increasing doses (0, 2.5, 5, and 10 g) during a human feeding trial (25), as determined by 454 pyrosequencing of 16S rRNA tags. C. Proportion of *Bifidobacterium* lineage species II in the same individual, as determined by pyrosequencing. D. Cell numbers of *B. adolescentis* IVS-1 in the same individual, as quantified by strain-specific qRT-PCR.

To verify that *B. adolescentis* strain IVS-1 was specifically enriched by GOS *in vivo*, we devised a strain-specific qRT-PCR approach with primers based on the genome sequence of IVS-1. Primer specificity was validated against ten closely related

*Bifidobacterium* strains, fecal DNA from all subjects included in the human feeding trial (25) and five additional human individuals, and ten fecal samples from Sprague-Dawley rats from an independent experiment. A detectable PCR product was obtained only with DNA from *B. adolescentis* IVS-1 and the fecal sample from which the strain was isolated. This finding indicated that the primers were highly strain specific and that strain IVS-1 was present only in the human subject from whom it was isolated.

The strain-specific qRT-PCR system was then used to quantify the abundance of IVS-1 in fecal samples from this subject during the GOS feeding study. This analysis revealed that IVS-1 levels were increased 8-fold during both the 5-g and 10-g GOS dose periods compared to the 0-g period (P < 0.001) (Fig. 3.1D), before returning to baseline levels immediately after GOS consumption ended. Collectively, these results demonstrated the utility of IVS to select a bacterial strain enriched in the human gastrointestinal tract through dietary administration of a prebiotic.



**Fig. 3.2.** Test of a synbiotic combination of *B. adolescentis* IVS-1 and GOS in a high-fatdiet rat model. A. Experimental design of the rat study. Rats were fed either a standard diet or a high-fat diet for 8 weeks, supplemented with or without a probiotic (IVS-1), a prebiotic (GOS), or a synbiotic (IVS-1 plus GOS) for the last 4 weeks. B. Quantification of absolute cell numbers of bifidobacteria in colonic and cecal contents by genus-specific qRT-PCR. C. Strain-specific qRT-PCR was used to quantify absolute numbers of *B. adolescentis* IVS-1 in colonic and cecal contents.

# 3.4.2 Test of the synbiotic combination using rats on a high-fat diet.

We systematically tested synergism between strain IVS-1 and GOS when used as a synbiotic in rats fed a high-fat diet (Fig. 3.2A). Decreases in numbers of bifidobacteria are often observed during high-fat-diet feeding (41–43). To determine if our synbiotic

strategy could redress this decrease, we employed a rat model of high-fat-diet-induced NAFLD where rats develop steatosis (fatty liver) but do not show an increase in body weight, develop liver inflammation, or progress to nonalcoholic steatohepatitis (NASH) (28). In our study, all high-fat-diet-fed rats developed steatosis (i.e., liver triglyceride levels of  $> 50 \ \mu g \ mg^{-1}$  of tissue) and had slightly increased plasma ALP levels compared to rats fed a standard diet (see Table 3.S2 in the supplemental material). Dietary supplements significantly influenced triglyceride liver contents; however, high-fat-diet-fed rats did not develop the histopathological liver inflammation characteristic of NASH (data not shown) and did not have increased plasma ALT levels (see Table 3.S2 in the supplemental material). Plasma tumor necrosis factor alpha (TNF- $\alpha$ ) and monocyte chemoattractant protein 1 (MCP-1) levels were not significantly elevated in the high-fatfed rats compared to the controls (see Table 3.S2 in the supplemental material), indicating a lack of systemic inflammation. Together, these data indicated that all rats receiving a high-fat diet developed NAFLD but not severe inflammatory disease that would confound the evaluation of synbiotic synergy and gut microbial ecology.

#### 3.4.3 Experiments in rats demonstrate strong synergism between IVS-1 and GOS.

To test the functionality of the prebiotic to support the establishment of *B*. *adolescentis* IVS-1 in the rat intestine, rats fed a high-fat diet were administered either IVS-1 alone, GOS alone, or the synbiotic combination; all findings were compared to results for the high-fat controls (Fig. 3.2A). Consistent with data from previous studies (41, 42), high-fat feeding decreased the abundance of bifidobacteria in both the colon and cecum of the rats, although this reduction did not reach statistical significance (Fig. 3.2B and Table

3.1). Genus-specific qRT-PCR analysis revealed that the prebiotic, but not IVS-1, significantly increased the total number of bifidobacteria in the cecum (Fig. 3.2B). These findings indicate that the introduction of IVS-1 alone did not increase *Bifidobacterium* abundance above baseline levels (~10<sup>8</sup> cells/g), whereas the prebiotic substrate was able to support the resident population. Compared to IVS-1 and GOS alone, the combination of the two dramatically increased the total number of bifidobacteria in the cecum (P < 0.01 between synbiotic and prebiotic treatments; P < 0.001 between synbiotic and probiotic treatments; P < 0.001 between synbiotic synbio

Strain-specific qRT-PCR analysis of *B. adolescentis* IVS-1 clearly demonstrated a synergistic effect of IVS-1 and GOS in the colon and in the cecum. Even though rats receiving IVS-1 alone consumed significantly more IVS-1 on a daily basis than did rats given the synbiotic due to increased drinking water consumption (P < 0.0001) (see Table 3.S2 in the supplemental material), the synbiotic led to an almost 2-log increase in the level of IVS-1 in the colon and cecum (9.47 ± 0.2 log<sub>10</sub> cells g<sup>-1</sup> and 9.43 ± 0.2 log<sub>10</sub> cells g<sup>-1</sup>, respectively) compared with the probiotic treatment (7.9 ± 0.1 and 7.44 ± 0.3 log<sub>10</sub> cells g<sup>-1</sup> in the cecum and colon, respectively) (P < 0.0001) (Fig. 3.2C). No IVS-1 was detected in rats fed the standard diet, the high-fat diet, or the prebiotic alone.

#### 3.4.4. 16S rRNA sequencing confirms synergism between IVS-1 and GOS in vivo.

We analyzed the 16S rRNA tags obtained via Illumina sequencing to gain a community-wide perspective on treatment effects on the resident gut microbiota. The ability of probiotic and synbiotic treatments to establish IVS-1 in rats was assessed based

on the abundance of an operational taxonomic unit (OTU) representing the species *B. adolescentis* (OTU\_2). This species was undetectable in rats that did not receive the probiotic treatment but constituted 3.4% of the microbiota in rats fed IVS-1 (Fig. 3.3A and Table 3.1). This finding indicates that the *B. adolescentis* population observed in rats was due solely to the administration of IVS-1. This finding was expected, as this species is not a member of the normal rat microbiota. Sequences representing *B. adolescentis* were enriched to 37.0% in rats receiving the synbiotic treatment, indicating a significant enhancement of the probiotic (in terms of abundance) due to the addition of the prebiotic (P = 0.0159). Without GOS, IVS-1 was only the eighth most abundant OTU in the rats' colonic microbiota, while it became the most abundant OTU when given together with GOS, having an abundance almost four times higher than that of the second most abundant OTU (a *Blautia* species, at 9.7%) (Fig. 3.3A). This finding demonstrated that IVS-1 could be introduced as the dominant member of the rat gut microbiota when GOS was also provided.
	Mean % bacterial abundance ± SD <sup>c</sup>						
Taxonomic group	Standard diet	Control High-Fat Diet (HF)	Prebiotic (HF)	Probiotic (HF)	Synbiotic (HF)	ANOVA <i>P-</i> value	
Phylum							
Firmicutes Actinobacteria	87.6±5 A 8.9±6 AB	88.8±9 A 3.6±2 A	76.9±8 A 19.6±8 BC	87.8±5 A 7.6±4 A	59.3±7 B 39.1±7 C	<0.0001 <0.0001	
Family							
Clostridiaceae	3.9±6	0.5±1	$0.8\pm1$	5.4±6 A	0.2±0 B	0.0061	
Incertae Sedis XIV	3.7±6	7.5±10	11.0±15	1.1±2 A	17.3±12 B	0.0342	
Streptococcaceae	12.7±5	21.3±5 A	9.3±1	8.9±6 B	6.6±2 B	0.0045	
Erysipelotrichaceae	16.8±11	21.3±17	9.2±1 A	26.5±10 B	8.3±3 A	0.0226	
Bifidobacteriaceae	5.9±7 A	1.3±1 A	17.0±9	4.1±2 A	37.8±7 B	0.0017	
Coriobacteriaceae	$0.6\pm0$	0.3±0	$1.0{\pm}1$	1.9±3 A	0.2±0 B	0.0263	
Rikenellaceae	0.9±1 A	0.1±0	0.3±0	0.1±0	0.0±0 B	0.0181	
Genus							
Clostridium	3.9±6	0.5±1 A	$0.8 \pm 1$	5.3±6 B	0.2±0 A	0.0122	
Blautia	3.4±6	7.4±10	$11.0{\pm}15$	0.9±1 A	17.2±12 B	0.0431	
Holdemania	0.1±0	1.0±2 A	0.9±0	0.0±0 B	0.0±0 B	0.0117	
Bifidobacterium	5.9±7 A	1.3±1 A	17.0±9	4.1±2 A	37.8±7 B	0.0017	
Lactococcus	12.4±4	21.0±5 A	9.1±1	8.7±6 B	6.3±2 B	0.0045	
Alistipes	0.9±1 A	0.1±0	0.3±0	0.1±0	0.0±0 B	0.0181	
OTUs <sup>a</sup>							
OTU_2 (B. adolescentis, 99%)	0.0±0 A	0.0±0 A	0.0±0 A	3.4±2 BC	37.0±7 BD	< 0.0001	
OTU_1 (L. lactis, 100%)	12.4±4	21.0±5 A	9.1±1	8.6±6 B	6.3±2 B	0.0045	
OTU_626 ( <i>Lachnospiracea</i> e <sup>b</sup> )	0.4±1 A	0.1±0 A	0.0±0 B	ND	0.1±0	0.0002	
OTU_7 ( <i>Turicibacter</i> sanguinis, 97%)	3.7±3	2.3±3	2.3±2	9.1±7 A	0.5±1 B	0.0279	
OTU_14 ( <i>Blautia</i> <sup>b</sup> )	0.0±0 A	0.1±0 AB	1.5±1 BC	0.0±0 AB	9.7±6 CF	0.0003	
OTU_33 (L. intestinalis, 99%)	ND	0.0±0 A	ND	1.1±2 B	0.0±0 A	0.0022	
OTU_9 ( <i>Clostridium</i> sp. <sup>b</sup> )	3.8±6	0.5±1 A	0.7±1	5.3±6 B	0.2±0 A	0.0128	
OTU_6 (B. pseudolongum, 97%)	5.8±7	0.9±1	16.6±8 A	0.6±1	0.0±0 B	0.0293	
OTU_44 ( <i>C. cocleatum</i> , 99%)	ND	1.0±1 A	ND	0.0±0 B	ND	0.0121	

Table 3.1: Proportions of bacterial taxa significantly influenced by dietary treatments

<sup>a</sup> Percent homologies to the closest type strain in the database are shown in parentheses. If the strain could not be assigned to a type strain (<97% homology), RDP Classifier was used to determine the most likely genus, and the RDP Classifier value is shown (80% cutoff).

<sup>b</sup> OTU without closely related type strain (<97% homology) classified with RDP Classifier. <sup>c</sup> Values with different uppercase letters are significantly different from each other. HF, high fat; ND, not detected.

#### 3.4.5 Community-wide characterization of effects on gut microbiota.

GOS treatment alone promoted a remarkably specific bifidogenic response, leading to an increase in the abundance of only one OTU related to *Bifidobacterium pseudolongum* (OTU\_6) (Table 3.1 and Fig. 3.3A). These findings confirm the highly specific bifidogenic response of GOS, which was previously demonstrated in humans (32).

Although IVS-1 treatment alone did not significantly increase the abundance of the genus *Bifidobacterium*, it induced a significant increase in the abundance of *Bifidobacterium adolescentis* at the species level (Table 3.1). Of note, several unexpected changes were also detected, such as enrichment of the family *Clostridiaceae*, the genus *Clostridium*, and an OTU within this genus (OTU\_9). Furthermore, the abundance of an OTU related to *Lactobacillus intestinalis* (OTU\_33) increased, while that of an OTU related to *Lactococcus lactis* (OTU\_1) decreased (Table 3.1).

Synbiotic treatment significantly increased the proportion of *Actinobacteria* (P < 0.0001), the family *Bifidobacteriaceae* (P < 0.0017), and the genus *Bifidobacterium* (P < 0.0017) (Table 3.1). These shifts were almost completely equivalent to shifts of OTU\_2, showing that the above-described alterations at higher taxonomic levels were due to the enrichment of IVS-1. The establishment of IVS-1 was associated with an increase in the abundances of the genus *Blautia* and one OTU within this genus (OTU\_14). In addition, there was a reduction in the abundances of the phylum *Firmicutes* (P < 0.0001) and families within this phylum, including *Clostridiaceae*, *Streptococcaceae*, and *Erysipelotrichaceae*. The abundances of the genera *Clostridium* and *Lactococcus* and OTUs within these genera were also decreased (Table 3.1).



**Fig. 3.3.** Characterization of the rat colonic microbiota composition by Illumina sequencing of 16S rRNA tags. A. Analysis of colonic microbiota at the OTU level. OTUs representing at least 1% of total sequences are shown individually, while OTUs representing <1% are grouped. OTUs in colors other than light blue were significantly influenced by the dietary treatment. B. Principal coordinate analysis (Bray-Curtis distance) of beta diversity. C. NMDS plot of beta diversity based on Bray-Curtis distance. SD, standard diet.

To assess both the alpha and beta diversities of the community in the colon, different diversity indexes were calculated from the data. Specifically, Shannon's index and the number of observed OTUs were used to determine the alpha diversity, and principal coordinate analysis (PCoA) and nonmetric multidimensional scaling (NMDS) plots based on Bray-Curtis distance were used to visualize the similarity between samples for each treatment.

On average,  $135.41 \pm 34.4$  OTUs per sample were identified. Alpha diversity based on Shannon's index was not significantly influenced by the treatment; however, there was a tendency for reduced diversity in the synbiotic group. This was caused by a slight reduction in community evenness, likely due to the expansion of a single species (*B. adolescentis*) (data not shown). Two independent approaches were used to analyze the beta diversity of the microbiota communities among treatments. PCoA and NMDS, based on Bray-Curtis distances of beta diversity, revealed that communities from rats fed the synbiotic clustered separately from the microbiomes of rats fed all the other treatments, which clustered together (Fig. 3.3B and C). This finding demonstrated that only the synbiotic treatment caused a global shift in microbiota structure.

#### 3.4.6 Systematic analyses of associations between members of the gut microbiota.

To identify potential interactions between IVS-1 and members of the gut microbiota, and among other bacterial members, we performed correlation analyses on all taxon combinations in the data set. Correlations were performed by using bacterial abundance data from all treatments. Strong negative correlations between the family *Bifidobacteriaceae* and the family *Clostridiaceae* (Fig. 3.4A), the genera *Bifidobacterium* and *Lactococcus* (Fig. 3.4B), and the genera *Bifidobacterium* and *Akkermansia* (Fig. 3.4C) were observed. In addition, strain IVS-1 levels (OTU\_2) showed a negative correlation with *Lactococcus lactis* (Fig.3.4D) and a very tight negative association with resident *B. pseudolongum* (r = -0.64; *P* = 0.0004) (Fig. 3.4E). These negative associations suggest direct or indirect competition between these bacterial taxa. Positive associations between both *Bifidobacteriaceae* and *Bifidobacterium* and the genus *Blautia* were detected, suggesting a synergistic relationship, which may be supported by the addition of GOS (Table 3.1 and Fig. 3.4F).



**Fig. 3.4.** Correlation analysis of colonic taxa present in rats fed a high-fat diet supplemented with or without a probiotic (IVS-1), a prebiotic (GOS), or a synbiotic (IVS-1 plus GOS) or a standard diet. Bacterial quantities are expressed as percent abundances of total bacteria as determined by 16S rRNA sequencing. Spearman's correlations between *Bifidobacteriaceae* and *Clostridiaceae* (A), *Bifidobacterium* and *Lactococcus* (B), *Bifidobacterium* and *Akkermansia* (C), *Bifidobacterium adolescentis* IVS-1 and *Lactococcus lactis* (D), *Bifidobacterium adolescentis* IVS-1 and *Bifidobacterium* (E), and *Bifidobacterium* and *Blautia* (F) were determined.

#### **3.5 Discussion**

Synergistic synbiotics are a promising concept to modulate the composition of the gut microbiota and promote the establishment of probiotic organisms in the gut (16). Despite this potential, however, there are few *in vivo* human or animal studies providing evidence that prebiotics can be used to support specific probiotic strains. Unfortunately, most synbiotic studies, including work in rats (44–49), mice (50–52), pigs (53–57), chickens (58, 59), and humans (60), did not employ strain-specific detection methods and therefore did not provide information on the potential synergism between pre- and probiotics. Of the *in vivo* studies that did discriminate the probiotic strain, most still did not demonstrate that *in vivo* performance could be enhanced by a prebiotic. This accounts for experiments using synbiotic formulations in humans (61), rats (62, 63), and other animal models (64). These findings suggest that, with few exceptions (17–19), probiotic strains are unable to compete against the resident gut microbiota, which is inherently resistant to outside colonizers (65), even when an exogenous growth substrate in the form of a prebiotic is provided.

Several reasons may explain the low success rates of synergistic synbiotics when evaluated *in vivo*, even for combinations in which the probiotic strain is able to utilize the prebiotic substrate *in vitro*. First, to become established in the gut, the probiotic strain must be able to occupy an ecological niche. This means that strains must not only outcompete the resident microbiota for the prebiotic substrate but also secure other nutrients that might be growth limiting (such as amino acids, lipids, vitamins, minerals, and nucleotides, etc.). In addition, probiotic strains must tolerate the prevailing environmental conditions in the digestive tract (including pH, bile acids, IgA, and defensins). Ultimately, *in vitro* tests are unable to predict the ability of a probiotic to benefit from a prebiotic substrate within the constraints of the competitive gastrointestinal environment. In contrast, the IVS approach described here overcomes many limitations of *in vitro* tests used to formulate synbiotics because it provides a basis for identifying bacterial strains that are able to utilize the prebiotic substrate under the same ecological conditions in which they are intended to function.

In this study, we employed IVS and selected a synbiotic combination that was tested in a rat model of NAFLD. Although the synbiotic did not influence host phenotypes, it was highly efficient at enhancing population levels of the probiotic strain, making it the most dominant OTU in the gut (Fig. 3.3A and Table 3.1). These findings provide a proof of concept for the potential of *in vivo* selection to identify synbiotic combinations that are, in ecological terms, highly synergistic. In addition to enhancing the abundance of strain IVS-1, the synbiotic used here also redressed the high-fat-diet-induced reduction in the level of bifidobacteria detected in rats that is often reported in the literature (41–43). Therefore, although no metabolic benefits were seen in the rat model used in our study, the synbiotic may be beneficial in other scenarios, as bifidobacteria are considered health-promoting organisms (6, 26, 66–68).

The community-wide analysis provided evidence that synergism between GOS and strain IVS-1 increased the competitive fitness of the strain in the rat intestinal tract. *B. pseudolongum*, which is a natural member of the rat GI tract (69), was detected in relative abundances of 5.8 % and 0.9 % in rats fed the standard and high-fat diets, respectively. Although the probiotic treatment did not affect levels of *B. pseudolongum*, the prebiotic treatment increased the abundance of this species to 16.6 %, indicating that *B*.

pseudolongum utilized GOS in vivo. However, the parallel addition of strain IVS-1 with GOS completely excluded *B. pseudolongum*, and a strong negative correlation between this species and IVS-1 was observed (r = -0.67; P = 0.0006) (Fig. 3.4E). These findings indicate that IVS-1 not only had a higher affinity for GOS in vivo than the resident Bifidobacterium species but also utilized GOS to increase its competitiveness and effectively outcompete a closely related resident species. This finding is consistent with the niche exclusion model, which states that the organism most efficient at using limited nutrients outcompetes its competitors for the same niche (70). Strong inverse correlations between bifidobacteria and *Clostridiaceae*, *Lactococcus*, and *Akkermansia* (Fig. 3.4A to C) were also observed. It is likely that these associations are also due to niche competition and are potentially enhanced by GOS administration. Bifidobacteria produce short-chain fatty acids that are inhibitory to other bacteria either by lowering the pH or via direct antimicrobial effects (e.g., acetic acid) (71). In summary, these findings demonstrate that the competitive fitness of strain IVS-1 was increased by GOS, which supports the conclusion that IVS can select synbiotic combinations with extremely high synergism. To what degree the increased competitive interactions between IVS-1 and the resident microbiota impact host health is difficult to predict and likely context dependent, but they clearly should be considered in future studies.

Correlation analyses revealed only one positive association among members of the rat microbiota, between the bifidobacteria (at the family and genus levels) and the genus *Blautia*. The abundance of OTU\_14, an uncultured *Blautia* strain, was also significantly increased by GOS and in the synbiotic treatment (Table 3.1). The positive correlation between *Bifidobacterium* and *Blautia* (Fig. 3.4F) indicates a synergistic effect between the

two taxa. The significant increase in the abundance of *Blautia* in the synbiotic treatments further suggests a syntrophic interaction between IVS-1 and *Blautia*, as GOS is consumed mainly by bifidobacteria (72), and the genus *Blautia* is not reported to utilize GOS. In contrast, the genus *Blautia* contains bacteria that are hydrogenotrophic acetogens, which utilize  $H_2$  and  $CO_2$  as energy sources (73). Although bifidobacteria do not produce these gases, cross-feeding between bifidobacteria and butyrate-producing colon bacteria can result in  $H_2$  and  $CO_2$  production (74), which might explain the positive correlations between *Bifidobacterium* and *Blautia*. However, additional experiments are necessary to establish the mechanism by which GOS can enhance the populations of *Blautia* in the gut and the positive associations between this genus and IVS-1.

In this study, we have shown how IVS can be used to formulate a highly synergistic synbiotic that can substantially enhance population levels and the competitiveness of a putative probiotic strain in the gastrointestinal tract and establish it as the dominant member of the gut microbiota in a conventional animal model. To our knowledge, this has not yet been reported in the probiotic literature. The process of IVS is broadly applicable and can easily be extended to other host species, body sites, prebiotic substrates (or dietary fibers), or target organisms. For example, it may be possible to use IVS to enhance other putative health-promoting genera such as *Akkermansia*, which has been shown to respond to prebiotics *in vivo* (75). While we selected *B. adolescentis* IVS-1 during a human trial that did not determine the physiological effect of GOS, IVS might be especially powerful when combined with a human clinical trial that determines the beneficial effect of a prebiotic on the host as the primary selection criterion. Therefore, to develop synbiotics for specific health applications, the IVS concept should be extended to select bacterial strains that not

only responded to the prebiotic but whose expansion correlated with beneficial physiological effects for the host. Such an approach would have the potential to identify health-promoting strains whose metabolic activity *in vivo* could be increased through a prebiotic. This might also result in synbiotic applications with greater health effects than those of the prebiotic alone, especially in the subset of humans who do not respond to the prebiotic (14, 32). A human study testing the synbiotic combination identified here (and comparing it with a synbiotic that includes a *Bifidobacterium* strain that can ferment GOS but was not selected by IVS) is currently in progress. Clearly, the application of IVS is likely to enhance the ecological performance of probiotic strains or live biotherapeutics within the habitats in which they are thought to function, and the technology could be readily applied in the design of microbiota-modulating therapies.

## 3.6 Supplemental material

Nutritional profile	High-Fat Diet	Standard Diet
Protein [%]	24.2	17.6
Fat [%]	34.7	5.2
Fiber (max) [%]	5.5	3.9
Carbohydrates [%]	27.8	68.3
Energy (kcal/g)	Kcal / %	Kcal / %
From Protein	0.969 / 18.6	0.705 / 18.3
From Fat (ether extract)	3.122 / 59.9	0.464 / 12.1
From Carbohydrates	1.113 / 21.4	2.733 / 71.0

 Table 3.S1. Composition of standard and high-fat diets.

	Standard Diet	High-Fat Diet	High-Fat Diet Prebiotic	High-Fat Diet Probiotic	High-Fat Diet Synbiotic	P - value ANOVA
Average body weight [g rat <sup>-1</sup> ]	475 ±19	449 ±46	499 ±81	496 ±62	502 ±51	0.5446
Average relative epididymal fat pad weight [% rat <sup>-1</sup> ]	0.84 ±0.2	1.04 ±0.4	1.12 ±0.3	1.01 ±0.2	1.08 ±0.5	0.8267
Average water consumption [ml rat <sup>-1</sup> day <sup>-1</sup> ]	29.3 ±4.1 <sup>b</sup>	35.6 ±4.5 <sup>a</sup>	36.3 ±4.9 <sup>a</sup>	$41.9 \pm 8.6^{\circ}$	35.4 ±4.5ª	<0.0001
Average IVS-1 consumption [CFU rat <sup>-1</sup> day <sup>-1</sup> ]	NA	NA	NA	1.26 x 10 <sup>9 a</sup>	1.06 x 10 <sup>9 b</sup>	0.0001#
Average GOS consumption [g rat <sup>-1</sup> day <sup>-1</sup> ]	NA	NA	1.20	NA	1.17	0.2063#
Triglyceride content [µg TG mg <sup>-1</sup> tissue]	17.8 ±4.3 <sup>a</sup>	$70.7 \pm 1.4^{b}$	62.7 ±1.2 <sup>bc</sup>	53.5 ±0.8 <sup>c</sup>	$92.4 \pm 2.4^{d}$	<0.0001
Alkanine Phosphatase (ALP) [units liter <sup>-1</sup> ]	227 ±83	399 ±55	464 ±127	458 ±167	460 ±120	0.0646
Alanine Transferase (ALT) [units liter <sup>-1</sup> ]	34 ±13	42 ±11	36 ±5	37±7	41 ±4	0.6418
TNF-α [pg ml <sup>-1</sup> ]	14.2 ±2.3	14.1 ±2.6	16.2 ±3.2	16.7 ±7.9	13.8 ±1.4	0.4718
MCP-1 [pg ml <sup>-1</sup> ]	224 ±25	228 ±37	239 ±21	248 ±63	243 ±36	0.6345

**Table 3.S2.** Body weight, relative epididymal fat pad weight, consumed drinking water,consumed IVS-1 and GOS, and host physiological markers.

\* as a threshold for steatosis was a liver triglyceride levels greater than 50 µg/mg of tissue NA: not applicable

<sup>#</sup> Student's *t*-test was applied

Values with different letters are significantly different from each other



**Figure 3.S1.** Growth of Bifidobacterium adolescentis IVS-1 in basal MRS supplemented with GOS, 0.16% lactose (residual sugar present in the commercial GOS), or basal MRS without carbohydrates.

#### **3.7 Acknowledgements**

This work was supported by USDA grant 2012-67017-19344. This work was also supported by the Nebraska Research Initiative's Proof of Concept Program, administered by the University of Nebraska.

We thank the UNL Institutional Animal Care Program staff for excellent animal care and Carol Casey at the Veterans Administration Medical Center for her advice and comments. We also acknowledge Catherine Muller for her contribution to the animal experiments.

#### **3.8 References**

1. Hildebrand F, Nguyen TLA, Brinkman B, Yunta RG, Cauwe B, Vandenabeele P, Liston A, Raes J. 2013. Inflammation-associated entero-types, host genotype, cage and interindividual effects drive gut microbiota variation in common laboratory mice. *Genome Biol* 14:R4. http://dx .doi.org/10.1186/gb-2013-14-1-r4.

2. Lee DK, Park SY, An HM, Kim JR, Kim MJ, Lee SW, Cha MK, Kim SA, Chung MJ, Lee KO, Ha NJ. 2011. Antimicrobial activity of Bifidobacterium spp. isolated from healthy adult Koreans against cariogenic microflora. *Arch Oral Biol* 56:1047–1054. http://dx.doi.org/10 .1016/j.archoralbio.2011.03.002.

3. Clemente JC, Ursell LK, Parfrey LW, Knight R. 2012. The impact of the gut microbiota on human health: an integrative view. *Cell* 148:1258– 1270. http://dx.doi.org/10.1016/j.cell.2012.01.035.

4. Round JL, Mazmanian SK. 2009. The gut microbiota shapes intestinal immune responses during health and disease. *Nat Rev Immunol* 9:313–323. http://dx.doi.org/10.1038/nri2515.

5. Tlaskalová-Hogenová H, Ste <sup>\*</sup>pánková R, Kozáková H, Hudcovic T, Vannucci L, Tucková L, Rossmann P, Hrnc<sup>\*</sup> T, Kverka M, Zákostelská<sup>\*</sup>ír Z, Klimešová K, Pr<sup>\*</sup>ibylová J, Bártová J, Sanchez D, Fundová P, Borovská D, Srùtková D, Zídek Z, Schwarzer M, Drastich P, Funda DP. 2011. The role of gut microbiota (commensal bacteria) and the mucosal barrier in the pathogenesis of inflammatory and autoimmune diseases and cancer: contribution of germ-free and gnotobiotic animal models of human diseases. *Cell Mol Immunol* 8:110–120. http://dx.doi.org/10.1038 /cmi.2010.67. 6. Cani PD, Delzenne NM. 2009. Interplay between obesity and associated metabolic disorders: new insights into the gut microbiota. *Curr Opin Pharmacol* 9:737–743. http://dx.doi.org/10.1016/j.coph.2009.06.016.

7. Hooper LV, Littman DR, Macpherson AJ. 2012. Interactions between the microbiota and the immune system. *Science* 336:1268–1273. http://dx .doi.org/10.1126/science.1223490.

8. Walker AW, Lawley TD. 2013. Therapeutic modulation of intestinal dysbiosis. *Pharmacol Res* 69:75–86. http://dx.doi.org/10.1016/j.phrs .2012.09.008.

9. Frank DN, Pace NR. 2008. Gastrointestinal microbiology enters the metagenomics era. *Curr Opin Gastroenterol* 24:4–10. http://dx.doi.org/10 .1097/MOG.0b013e3282f2b0e8.

10. Al-Sheraji SH, Ismail A, Manap MY, Mustafa S, Yusof RM, Hassan FA. 2013.
Prebiotics as functional foods: a review. *J Funct Foods* 5:1542–1553.
http://dx.doi.org/10.1016/j.jff.2013.08.009.

11. Gibson GR, Roberfroid MB. 1995. Dietary modulation of the human colonic microbiota: introducing the concept of prebiotics. *J Nutr* 125: 1401–1412.

12. Tuohy KM, Probert HM, Smejkal CW, Gibson GR. 2003. Using probiotics and prebiotics to improve gut health. *Drug Discov Today* 8:692–700. http://dx.doi.org/10.1016/S1359-6446(03)02746-6.

13. Hooda S, Boler BMV, Serao MCR, Brulc JM, Staeger MA, Boileau TW, Dowd SE, Fahey GC, Jr, Swanson KS. 2012. 454 pyrosequencing reveals a shift in fecal microbiota

of healthy adultmenconsuming polydextrose or soluble corn fiber. *J Nutr* 142:1259–1265. http://dx.doi.org/10.3945/jn .112.158766.

14. Martínez I, Kim J, Duffy PR, Schlegel VL, Walter J. 2010. Resistant starches types 2 and 4 have differential effects on the composition of the fecal microbiota in human subjects. *PLoS One* 5:e15046. http://dx.doi.org/10.1371/journal.pone.0015046.

15. Hill C, Guarner F, Reid G, Gibson GR, Merenstein DJ, Pot B, Morelli L, Canani RB, Flint HJ, Salminen S, Calder PC, Sanders ME. 2014. Expert consensus document: the International Scientific Association for Probiotics and Prebiotics consensus statement on the scope and appropri- ate use of the term probiotic. *Nat Rev Gastroenterol Hepatol* 11:506–514. http://dx.doi.org/10.1038/nrgastro.2014.66.

16. Kolida S, Gibson GR. 2011. Synbiotics in health and disease. *Annu Rev Food Sci Technol* 2:373–393. http://dx.doi.org/10.1146/annurev-food -022510-133739.

17. Tanaka R, Takayama H, Morotomi M, Kuroshima T, Ueyama S, Matsumoto K, Kuroda A, Masahiko M. 1983. Effects of administration of TOS and *Bifidobacterium breve* 4006 on the human fecal flora. Bifidobact Microflora 2:17–24. http://dx.doi.org/10.12938/bifidus1982.2.1\_17.

18. Ogawa T, Asai Y, Yasuda K, Sakamoto H. 2005. Oral immunoadjuvant activity of a new synbiotic *Lactobacillus casei* subsp *casei* in conjunction with dextran in BALB/c mice. *Nutr Res* 25:295–304. http://dx.doi.org/10 .1016/j.nutres.2004.10.012.

19. Ogawa T, Asai Y, Tamai R, Makimura Y, Sakamoto H, Hashikawa S, Yasuda K. 2006. Natural killer cell activities of synbiotic *Lactobacillus casei* ssp. *casei* in conjunction with dextran. *Clin Exp Immunol* 143:103–109. http://dx.doi.org/10.1111/j.1365-2249.2005.02975.x.

20. Rabiu BA, Jay AJ, Gibson GR, Rastall RA. 2001. Synthesis and fermentation properties of novel galacto-oligosaccharides by beta-galactosidases from Bifidobacterium species. *Appl Environ Microbiol* 67:2526–2530. http://dx.doi.org/10.1128/AEM.67.6.2526-2530.2001.

21. Modesto M, D'Aimmo MR, Stefanini I, Trevisi P, De Filippi S, Casini L, Mazzoni M, Bosi P, Biavati B. 2009. A novel strategy to select *Bifidobacterium* strains and prebiotics as natural growth promoters in newly weaned pigs. *Livest Sci* 122:248–258. http://dx.doi.org/10.1016/j.livsci .2008.08.017.

22. Malinen E, Mättö J, Salmitie M, Alander M, Saarela M, Palva A. 2002. Analysis of *Bifidobacterium* populations in human faecal samples from a consumption trial with *Bifidobacterium lactis* Bb-12 and a galacto-oligosaccharide preparation. *Syst Appl Microbiol* 25:249–258. http://dx .doi.org/10.1078/0723-2020-00117.

23. Alander M, Mättö J, Kneifel W, Johansson M, Koegle B, Crittenden R, Mattila-Sandholm T, Saarela M. 2001. Effect of galacto-oligosaccharide supplementation on human faecal microflora and on survival and persistence *of Bifidobacterium lactis* Bb-12 in the gastrointestinal tract. *Int Dairy* J 11:817–825. http://dx.doi.org/10.1016/S0958-6946(01)00100-5.

24. Rattanaprasert M, Roos S, Hutkins RW, Walter J. 2014. Quantitative evaluation of synbiotic strategies to improve persistence and metabolic activity of *Lactobacillus reuteri* 

DSM17938 in the human gastrointestinal tract. *J Funct Foods* 10:85–94. http://dx.doi.org/10.1016/j.jff.2014.05.017.

25. Davis LMG, Martínez I, Walter J, Hutkins R. 2010. A dose dependent impact of prebiotic galacto-oligosaccharides on the intestinal microbiota of healthy adults. *Int J Food Microbiol* 144:285–292. http://dx.doi.org/10.1016/j.ijfoodmicro.2010.10.007.

26. Martínez I, Wallace G, Zhang C, Legge R, Benson AK, Carr TP, Moriyama EN, Walter J. 2009. Diet-induced metabolic improvements in a hamster model of hypercholesterolemia are strongly linked to alterations of the gut microbiota. *Appl Environ Microbiol* 75:4175–4184. http://dx .doi.org/10.1128/AEM.00380-09.

27. Rozen S, Skaletsky H. 2000. Primer3 on the WWW for general users and for biologist programmers. *Methods Mol Biol* 132:365–386.

28. Ahmed U, Redgrave TG, Oates PS. 2009. Effect of dietary fat to produce non-alcoholic fatty liver in the rat. *J Gastroenterol Hepatol* 24:1463–1471. http://dx.doi.org/10.1111/j.1440-1746.2009.05870.x.

29. Folch J, Lees M, Sloan Stanely G. 1957. A simple method for the isolation and purification of total lipids from animal tissues. *J Biol Chem* 226:497–509.

30. Simental-Mendía LE, Rodríguez-Hernández H, Rodríguez-Morán M, Guerrero-Romero F. 2012. The alanine aminotransferase to triglycerides ratio as a marker to identify nonalcoholic fatty liver disease. *Eur J Gastro- enterol Hepatol* 24:1173–1177. http://dx.doi.org/10.1097/MEG.0b013e32 83564ee5. 31. Lee J-T, Pao L-H, Lee M-S, Liao J-W, Shih C-M, Hsiong C-H, Yin F-Y, Shih T-Y, Hu OY-P. 2013. A new approach to facilitate diagnosis of nonalcoholic fatty liver disease through a galactose single point method in rats with fatty liver. *Dig Liver Dis* 45:134–141. http://dx.doi.org/10.1016/j.dld.2012.08.019.

32. Davis LMG, Martínez I, Walter J, Goin C, Hutkins RW. 2011. Barcoded pyrosequencing reveals that consumption of galacto-oligosaccharides results in a highly specific bifidogenic response in humans. *PLoS One* 6:e25200. http://dx.doi.org/10.1371/journal.pone.0025200.

33. Ito M, Deguchi Y, Miyamori A, Matsumoto K, Kikuchi H, Kobayashi Y, Yajima T, Kan T. 1990. Effects of administration of galacto-oligosaccha- rides on the human faecal microflora, stool weight and abdominal sensation. *Microb Ecol Health Dis* 3:285–292. http://dx.doi.org/10.3109 /08910609009140251.

34. Gibson GR. 1999. Dietary modulation of the human gut microflora using the prebiotics oligofructose and inulin. *J Nutr* 129:1438–1441.

35. Sako T, Matsumoto K, Tanaka R. 1999. Recent progress on research and applications of non-digestible galacto-oligosaccharides. *Int Dairy* J 9:69– 80. http://dx.doi.org/10.1016/S0958-6946(99)00046-1.

36. Macfarlane S, Macfarlane GT, Cummings JH. 2006. Review article: prebiotics in the gastrointestinal tract. *Aliment Pharmacol Ther* 24:701–714. http://dx.doi.org/10.1111/j.1365-2036.2006.03042.x.

37. Silk DBA, Davis A, Vulevic J, Tzortzis G, Gibson GR. 2009. Clinical trial: the effects of a trans-galacto-oligosaccharide prebiotic on faecal microbiota and symptoms in irritable

bowel syndrome. *Aliment Pharmacol Ther* 29: 508–518. http://dx.doi.org/10.1111/j.1365-2036.2008.03911.x.

38. Ben X-M. 2008. Low level of galacto-oligosaccharide in infant formula stimulates growth of intestinal *Bifidobacteria* and *Lactobacilli*. *World J Gastroenterol* 14:6564–6568. http://dx.doi.org/10.3748/wjg.14.6564.

39. Chen J, Wang R, Li X-F, Wang R-L. 2012. *Bifidobacterium adolescentis* supplementation ameliorates visceral fat accumulation and insulin sensitivity in an experimental model of the metabolic syndrome. *Br J Nutr* 107:1429–1434. http://dx.doi.org/10.1017/S0007114511004491.

40. Reichold A, Brenner SA, Spruss A, Förster-Fromme K, Bergheim I, Bischoff SC. 2014. *Bifidobacterium adolescentis* protects from the development of nonalcoholic steatohepatitis in a mouse model. *J Nutr Biochem* 25:118–125. http://dx.doi.org/10.1016/j.jnutbio.2013.09.011.

41. Cani PD, Neyrinck AM, Fava F, Knauf C, Burcelin RG, Tuohy KM, Gibson GR, Delzenne NM. 2007. Selective increases of bifidobacteria in gut microflora improve high-fat-diet-induced diabetes in mice through a mechanism associated with endotoxaemia. *Diabetologia* 50:2374–2383. http://dx.doi.org/10.1007/s00125-007-0791-0.

42. Everard A, Lazarevic V, Gaïa N, Johansson M, Ståhlman M, Backhed F, Delzenne NM, Schrenzel J, François P, Cani PD. 2014. Microbiome of prebiotic-treated mice reveals novel targets involved in host response during obesity. *ISME J* 8:2116–2130. http://dx.doi.org/10.1038/ismej.2014.45.

43. Cani PD, Delzenne NM, Amar J, Burcelin R. 2008. Role of gut microflora in the development of obesity and insulin resistance following high- fat diet feeding. *Pathol Biol* 56:305–309. http://dx.doi.org/10.1016/j .patbio.2007.09.008.

44. LiongMT,Shah NP. 2006. Effects of a *Lactobacillus casei* synbiotic on serum lipoprotein, intestinal microflora, and organic acids in rats. *J Dairy Sci* 89: 1390–1399. http://dx.doi.org/10.3168/jds.S0022-0302(06)72207-X.

45. Yang S-C, Chen J-Y, Shang H-F, Cheng T-Y, Tsou SC, Chen J-R. 2005. Effect of synbiotics on intestinal microflora and digestive enzyme activities in rats. *World J Gastroenterol* 11:7413–7417. http://dx.doi.org/10.3748 /wjg.v11.i47.7413.

46. Le Leu RK, Brown IL, Hu Y, Bird AR, Jackson M, Esterman A, Young GP. 2005. A synbiotic combination of resistant starch and *Bifidobacterium lactis* facilitates apoptotic deletion of carcinogen-damaged cells in rat colon. *J Nutr* 135:996–1001.

47. Igarashi M, Liyama Y, Kato R, Tomita M, Asami N. 1994. Effect of Bifidobacterium longum and lactulose on the strength of bone in ovariectomized osteoporosis model rats. *Bifidus* 7:139–147.

48. Bielecka M, Biedrzycka E, Majkowska A. 2002. Selection of probiotics and prebiotics for synbiotics and confirmation of their *in vivo* effectiveness. *Food Res Int* 35:125–131. http://dx.doi.org/10.1016/S0963-9969 (01)00173-9.

49. Bomhof MR, Saha DC, Reid DT, Paul HA, Reimer RA. 2014. Combined effects of oligofructose and *Bifidobacterium animalis* on gut microbiota and glycemia in obese rats. *Obesity* 22:763–771. http://dx.doi.org/10.1002 /oby.20632.

50. Frece J, Kos B, Svetec IK, Zgaga Z, Beganovic' J, Lebos A, Suskovic 'J. 2009. Synbiotic effect of *Lactobacillus helveticus* M92 and prebiotics on the intestinal microflora and immune system of mice. *J Dairy Res* 76:98–104. http://dx.doi.org/10.1017/S0022029908003737.

51. Foye OT, Huang I-F, Chiou CC, Walker WA, Shi HN. 2012. Early administration of probiotic *Lactobacillus acidophilus* and/or prebiotic inulin attenuates pathogen-mediated intestinal inflammation and Smad 7 cell signaling. *FEMS Immunol Med Microbiol* 65:467–480. http://dx.doi .org/10.1111/j.1574-695X.2012.00978.x.

52. Ogawa T, Hashikawa S, Asai Y, Sakamoto H, Yasuda K, Makimura Y. 2006. A new synbiotic, *Lactobacillus casei* subsp. *casei t*ogether with dextran, reduces murine and human allergic reaction. *FEMS Immunol Med Microbiol* 46:400–409. http://dx.doi.org/10.1111/j.1574-695X .2006.00046.x.

53. Piva A, Casadei G, Gatta PP, Luchansky JB, Biagi G. 2005. Effect of lactitol, lactic acid bacteria, or their combinations (synbiotic) on intestinal proteolysis *in vitro*, and on feed efficiency in weaned pigs. *Can J Anim Sci* 85:345–353. http://dx.doi.org/10.4141/A04-087.

54. Bird AR, Vuaran M, Crittenden R, Hayakawa T, Playne MJ, Brown IL, Topping DL. 2009. Comparative effects of a high-amylose starch and a fructo-oligosaccharide on fecal bifidobacteria numbers and short-chain fatty acids in pigs fed *Bifidobacterium animalis*. *Dig Dis Sci* 54:947–954. http://dx.doi.org/10.1007/s10620-008-0451-3.

55. Böhmer BM, Branner GR, Roth-Maier DA. 2005. Precaecal and faecal digestibility of inulin (DP10-12) or an inulin /*Enterococcus faecium* mix and effects on nutrient

digestibility and microbial gut flora. *J Anim Physiol Anim Nutr* (Berl) 89:388–396. http://dx.doi.org/10.1111/j.1439-0396.2005.00530.x.

56. Bomba A, Nemcová R, Gancarcíková S, Herich R, Guba P, Mudronová D. 2002. Improvement of the probiotic effect of microorganisms by their combination with maltodextrins, fructo-oligosaccharides and polyunsaturated fatty acids. *Br J Nutr* 88 (Suppl 1):S95–S99. http://dx.doi.org/10.1079/BJN2002634.

57. Shim SB, Verstegen MWA, Kim IH, Kwon OS, Verdonk JMAJ. 2005. Effects of feeding antibiotic-free creep feed supplemented with oligofructose, probiotics or synbiotics to suckling piglets increases the preweaning weight gain and composition of intestinal microbiota. *Arch Anim Nutr* 59:419–427. http://dx.doi.org/10.1080/17450390500353234.

58. Baffoni L, Gaggìa F, Di Gioia D, Santini C, Mogna L, Biavati B. 2012. A *Bifidobacterium*-based synbiotic product to reduce the transmission of C. jejuni along the poultry food chain. *Int J Food Microbiol* 157:156–161. http://dx.doi.org/10.1016/j.ijfoodmicro.2012.04.024.

59. Abdel-Raheem S, Abd-Allah S, Hassanein K. 2012. The effects of prebiotic, probiotic and synbiotic supplementation on intestinal microbial ecology and histomorphology of broiler chickens. *Int J Agro Vet Med Sci* 6:277–289. http://dx.doi.org/10.5455/ijavms.156.

60. Furrie E, Macfarlane S, Kennedy A, Cummings JH, Walsh SV, O'Neil DA, Macfarlane GT. 2005. Synbiotic therapy (*Bifidobacterium longum*/ Synergy 1) initiates resolution of inflammation in patients with active ulcerative colitis: a randomised controlled pilot trial. *Gut* 54:242–249. http://dx.doi.org/10.1136/gut.2004.044834.

61. Gopal PK, Prasad J, Gill HS. 2003. Effects of the consumption of *Bifidobacterium lactis* HN019 (DR10) and galacto-oligosaccharides on the microflora of the gastrointestinal tract in human subjects. *Nutr Res* 23:1313–1328. http://dx.doi.org/10.1016/S0271-5317(03)00134-9.

62. Schultz M, Munro K, Tannock GW, Melchner I, Göttl C, Schwietz H, Schölmerich J, Rath HC. 2004. Effects of feeding a probiotic preparation (SIM) containing inulin on the severity of colitis and on the composition of the intestinal microflora in HLA-B27 transgenic rats. *Clin Diagn Lab Immunol* 11:581–587. http://dx.doi.org/10.1128/CDLI.11.3.581-587.2004.

63. Nakanishi S, Kataoka K, Kuwahara T, Ohnishi Y. 2003. Effects of high amylose maize starch and *Clostridium butyricum* on metabolism in colonic microbiota and formation of azoxymethane-induced aberrant crypt foci in the rat colon. *Microbiol Immunol* 47:951–958. http://dx.doi.org/10.1111/j.1348-0421.2003.tb03469.x.

64. Krause DO, Bhandari SK, House JD, Nyachoti CM. 2010. Response of nursery pigs to a synbiotic preparation of starch and an anti-*Escherichia coli* K88 probiotic. *Appl Environ Microbiol* 76:8192–8200. http://dx.doi .org/10.1128/AEM.01427-10.

65. Stecher B, Hardt W-D. 2008. The role of microbiota in infectious disease. *Trends Microbiol* 16:107–114. http://dx.doi.org/10.1016/j.tim.2007.12 .008.

66. Kalliomäki M, Collado MC, Salminen S, Isolauri E. 2008. Early differences in fecal microbiota composition in children may predict overweight. *Am J Clin Nutr* 87:534–538.

67. Ewaschuk JB, Diaz H, Meddings L, Diederichs B, Dmytrash A, Backer J, Looijer-van Langen M, Madsen KL. 2008. Secreted bioactive factors from *Bifidobacterium infantis* 

enhance epithelial cell barrier function. *Am J Physiol Gastrointest Liver Physiol* 295:G1025–G1034. http://dx.doi.org/10.1152/ajpgi.90227.2008.

68. Wang Z, Xiao G, Yao Y, Guo S, Lu K, Sheng Z. 2006. The role of bifidobacteria in gut barrier function after thermal injury in rats. *J Trauma* 61:650–657. http://dx.doi.org/10.1097/01.ta.0000196574.70614.27.

69. Mitsuoka T. 1990. Bifidobacteria and their role in human health. *J Ind Microbiol* 6:263–267. http://dx.doi.org/10.1007/BF01575871.

70. Petrof EO, Khoruts A. 2014. From stool transplants to next-generation microbiotatherapeutics.Gastroenterology146:1573–1582.http://dx.doi.org/10.1053/j.gastro.2014.01.004.

71. Fukuda S, Toh H, Hase K, Oshima K, Nakanishi Y, Yoshimura K, Tobe T, Clarke JM, Topping DL, Suzuki T, Taylor TD, Itoh K, Kikuchi J, Morita H, Hattori M, Ohno H. 2011.
Bifidobacteria can protect from enteropathogenic infection through production of acetate. *Nature* 469: 543–547. http://dx.doi.org/10.1038/nature09646.

72. Maathuis AJH, van den Heuvel EG, Schoterman MHC, Venema K. 2012. Galactooligosaccharides have prebiotic activity in a dynamic *in vitro* colon model using a (13)Clabeling technique. *J Nutr* 142:1205–1212. http://dx.doi.org/10.3945/jn.111.157420.

73. Liu C, Finegold SM, Song Y, Lawson PA. 2008. Reclassification of *Clostridium* coccoides, *Ruminococcus hansenii*, *Ruminococcus hydrogenotrophicus*, *Ruminococcus luti*, *Ruminococcus productus* and *Ruminococcus schinkii* as *Blautia coccoides gen. nov.*, comb. nov., *Blautia hansenii comb. nov.*, *Blautia hydroge. Int J Syst Evol Microbiol* 58:1896–1902. http://dx.doi.org/10.1099/ijs.0.65208-0.

74. De Vuyst L, Leroy F. 2011. Cross-feeding between bifidobacteria and butyrateproducing colon bacteria explains bifdobacterial competitive- ness, butyrate production, and gas production. *Int J Food Microbiol* 149: 73–80. http://dx.doi.org/10.1016/j.ijfoodmicro.2011.03.003.

75. Everard A, Belzer C, Geurts L, Ouwerkerk JP, Druart C, Bindels LB, Guiot Y, Derrien M, Muccioli GG, Delzenne NM, de Vos WM, Cani PD. 2013. Cross-talk between *Akkermansia muciniphila* and intestinal epithelium controls diet-induced obesity. *Proc Natl Acad Sci USA* 110: 9066–9071.

#### Chapter 4

# Functional characterization of a rationally selected synbiotic application in obese adults.

### 4.1 Abstract

One strategy for enhancing the establishment of probiotic bacteria in the human intestinal tract is via the parallel administration of a prebiotic. Such combinations are referred to as synbiotics. We have developed a rationally formulated synbiotic combination based on *in vivo* selection (IVS). This approach employed ecological criteria to select a highly synergistic synbiotic combination, specifically, Bifidobacterium adolescentis strain IVS-1 and galactooligosaccharides (GOS). This synbiotic was previously shown to be synergistic in a rat model (1), and we have now tested the ability of this synbiotic to improve the abundance of bifidobacteria, and specifically of the probiotic strain, in obese human subjects. The study was designed as a randomized, placebo controlled, parallel arm clinical trial. When the rationally selected synbiotic was compared to a commercial synbiotic (Bifidobacterium lactis Bb12 and GOS), the synergistic synbiotic combination led to significantly higher levels of the probiotic strain in the gastrointestinal tract of the subjects than the control. In conclusion, this study demonstrated that IVS can be used to successfully formulate a synbiotic that can enhance the establishment and competitiveness of a putative probiotic strain in the human gastrointestinal tract.

#### **4.2 Introduction**

The contribution of the gut microbiome to human and animal health is now well established (2–4). Indeed, there are substantial efforts aimed at designing dietary strategies that modify the composition of the gut microbiota with the goal of preventing disease and promoting health (5, 6). In particular, probiotic bacteria and prebiotic fibers, as well as in a combined form as synbiotics, have been tested in clinical trials to treat a range of conditions, including IBS (7, 8), IBD (9–12), lactose intolerance (13, 14), and other gastrointestinal (GI) diseases and disorders (15–18).

In addition to GI diseases, the development of several systemic conditions, including metabolic endotoxemia and metabolic syndrome, are also associated with a gut dysbiosis that could potentially be redressed through dietary modulations. Evidence from human and animal studies suggests that a constant low grade inflammation of the GI lining may precede or initiate the development of metabolic disorders (19). The origin of this inflammation may be caused by alterations in the composition of the gut microbiota, which is directly involved in controlling the host's gut barrier function, and increases systemic exposure to pro-inflammatory free fatty acids (20, 21). While the exact process of decreased barrier function and increased permeability due to the gut microbiota remains unclear, data from animal studies suggest that several mechanisms are included. For example, changes in the abundance of certain members of the gut microbiota lead to changes in the production and availability of short chain fatty acids, which are absorbed by the gut epithelial cells. Additionally, the gut microbiota affects the distribution of tight junction proteins, such as ZO-1 and Occludin, and influences the endocannabinoid system tone, leading to a higher expression of anandamide and cannabinoid receptor 1 (22, 23). These factors lead to increased intestinal permeability and subsequently to an increased

160

translocation of microbe-derived lipopolysaccharides (LPS) into the bloodstream. This induces metabolic endotoxemia, which eventually leads to metabolic syndrome (22, 24, 25).

Several studies have established that *Bifidobacterium* spp. have beneficial effects in the pathology associated with impaired barrier function by reducing gut permeability and improving epithelial cell barrier function (6, 24, 26–29). Specifically, *Bifidobacterium* adolescentis administration to rats ameliorated insulin sensitivity, white fat accumulation, and liver weight (30), and significantly lowered rates of bacterial translocation (27). This species was also shown to attenuate the formation of reactive oxygen species, activate nuclear factor  $\kappa B$  (NF $\kappa B$ ), and reduced markers of inflammation in the rodent liver (31). Similarly, administration of Bifidobacterium longum reduced GI inflammation and metabolic syndrome, and reduced levels of LPS and interleukin beta in a rat model (32). Another species, *Bifidobacterium breve* reduced triacylglycerol content, decreased serum TNF- $\alpha$  levels, and restored serum LPS levels to levels that were observed in control rats (33). It also suppressed accumulation of epididymal fat pad and body weight, improved fasting levels of glucose and insulin, and improved total cholesterol values in a mouse model (34). Finally, supplementation of newborn mice with *Bifidobacterium infantis* and Bifidobacterium bifidum significantly lowered intestinal endotoxin levels compared to control mice (35), and *B. infantis* normalized gut permeability and decreased colonic IFN- $\gamma$  secretion in IL-10-deficient mice (26).

Collectively, these reports suggest that dietary strategies that support both the size and physiologic activity of *Bifidobacterium* populations in the human GI tract could be effective for a range of metabolic disorders. Ordinarily, the abundance of these bacteria in adults is relatively low (< 3%) (36), and negatively correlates with high-fat and low fiber diets, which are common in Westernized societies (24, 37). This low abundance of bifidobacteria may be addressed by dietary treatments such as probiotics, prebiotics, or synbiotics. The consumption of prebiotic carbohydrates, such as galactooligosaccharide (GOS) and other fibers, have been shown to increase autochthonous bifidobacteria in infants (38, 39) and adults (40–42). However, not all subjects respond to prebiotic interventions, even at high doses (36, 43, 44), and the subjects may not possess a particular *Bifidobacterium* strain of interest. One strategy to enrich for bifidobacteria, even in non-responders, would be to administer the prebiotic together with a *Bifidobacterium* strain that is capable of metabolizing the prebiotic component *in vivo*. Such pairings are referred to as synergistic synbiotics (45).

However, the rational selection of these synbiotic strains is critical as the potential of the probiotic strains to become established *in vivo* is significantly limited due to colonization resistance conferred by the resident or commensal microbial population, the host, and other abiotic factors. Allochthonous strains may, for example, lack adaptive traits necessary to become competitive and physiologically active in the GI environment. Their ability to compete with the autochthonous microbiota in the GI tract is also compromised by niche exclusion, colonization resistance, nutrient availability, the host's immune system, and the prevailing environmental conditions in the digestive tract (46). These abiotic and biotic ecological factors have a major influence on the ability of allochthonous organisms (including most probiotics) to reach, and then become established in the human GI tract, even if only transiently. Indeed, studies have shown that bacterial strains that are allochthonous to the GI tract are washed out shortly after administration is discontinued, and pre-treatment conditions are quickly re-established (36, 47–50). By selecting bacterial strains that are autochthonous, adapted to the host GI environment, the colonization

resistance towards the strains based on these abiotic and biotic factors may be attenuated. Moreover, incorporating strains autochthonous to the adult GI tract, such as *B*. *adolescentis*, as a synbiotic might also be expected to enhance their probiotic function and colonization success.

We recently described a novel method for the selection of an autochthonous strain of B. adolescentis that was enriched in vivo by GOS (1). When this strain, B. adolescentis IVS-1, was fed to rats, its abundance reached 3.4 % of the total bacterial gut microbiota. However, when combined with GOS as a synbiotic, abundance of strain IVS-1 increased to more than 35 %. To determine if this rationally selected synbiotic would also show enhanced colonization in humans, we tested it in a parallel arm placebo controlled clinical trial with obese adults. Our goal was to assess the potential of this *in vivo* selected strain, combined with GOS as a synbiotic to establish bifidobacteria and redress metabolic aberrancies, with gut permeability as the primary endpoint. We compared establishment with a widely used commercial strain, *B. animalis* subsp. *lactis* Bb12, which has been used in synbiotic applications with GOS (48, 51, 52). Each strain was provided as individual treatments as well as combined with GOS as synbiotics. We compared the ability of the *in* vivo selected rationally designed synbiotic to the commercial synbiotic to alter the gut microbiota in obese individuals, and tested if GOS supported the colonization of the probiotic strains. This report focusses on the impact of the dietary treatments on the gut microbiota, while an assessment of the ability of each synbiotic to improve intestinal permeability and endotoxemia is currently ongoing.

Subjects. This study was designed as a randomized, placebo-controlled, parallel-arm clinical trial conducted at Rush University Medical Center (RUMC). Women and men between 18 and 60 years with a BMI of  $30.0 - 40.0 \text{ kg/m}^2$  were recruited. Subjects were permitted to have elevated liver enzymes due to fatty liver and metabolic syndrome, but were otherwise considered as healthy. Exclusion criteria included (1) prior intestinal resection, (2) patient history of GI diseases except for hiatal hernia, GERD, hemorrhoids, (3) severe renal disease defined by creatinine more than twice normal, (4) markedly abnormal liver function defined by ALT/AST over 4 times normal levels or elevated bilirubin (5) antibiotic use within the last 12 weeks prior to enrollment, (6) lean or overweight (BMI < 30 kg/m<sup>2</sup>), (7) intolerant to aspirin, (8) regular use of aspirin, (9) excessive alcohol intake (>2 drinks for men, 1 drink for women daily), (10) presence of chronic metabolic disease (cardiovascular disease, insulin requiring diabetes or uncontrolled diabetes, cancer, etc.), (11) a plan to have a major change in dietary habit during the following 6 months, (12) consumption of probiotics, prebiotics or synbiotics without an appropriate 4 week washout period, (13) lactose intolerance or malabsorption, (14) subjects younger than 18 or older than 60, (15) unwillingness to consent to the study.

**Study design.** A total of 105 subjects were recruited and randomly assigned to one of six treatment groups (Table 4.1). The randomization was controlled for age and race. Three visits were required for each subject (Figure 4.1). At Visit 1, potential subjects were screened for eligibility and provided with a form for a 3-day diet record, all supplies for

stool collection, and instructions for specimen handling and for completing these tasks before the next visit.

#### **Table 4.1.** Treatment groups

Group A	Lactose control
Group B	1 x 10 <sup>9</sup> CFU <i>B. adolescentis</i> IVS-1
Group C	1 x 10 <sup>9</sup> CFU <i>B. animalis</i> subsp. <i>lactis</i> Bb12
Group D	1 x 10 <sup>9</sup> CFU <i>B. adolescentis</i> IVS-1 + GOS
Group E	1 x 10 <sup>9</sup> CFU <i>B. animalis</i> subsp. <i>lactis</i> Bb12 + GOS
Group F	Galactooligosaccharide (GOS)

#### Treatment group Treatment

Subjects were instructed to store stool samples in Styrofoam coolers with freezer packs until delivery to the hospital. The samples were not allowed to be older than 24 hours if stored at -20°C, and not older than 5 hours if stored at room temperature. At visit 2, study subjects provided the completed food record and the baseline stool sample. Subjects were provided with one of the six treatments and consumed their randomly assigned supplement daily for three weeks as instructed. At the end of three weeks, subjects returned to the clinic to provide stool, as previously described. At the visit after 3 weeks of supplementation, the subject provided a stool sample, 3-day food records, and completed GI symptom questionnaires regarding adverse events. The latter included a standardized survey that rates bowel movement, stool consistency, discomfort, flatulence, abdominal pain, and bloating on a scale from 1 (best) to 10 (worst) (44). Weight, height, waist circumference, and blood pressure were measured, and BMI was calculated at each visit. Blood pressure was measured using an automated cuff with the average of three assessments used for statistical comparisons. Four weeks after the last treatment was consumed subjects provided a wash-out stool sample. Subjects gave a written informed consent before the study procedure.

Eleven subjects (two in group A, four in group B, two in group C, one in group D, and two in group E) did not follow all protocol requirements and were subsequently excluded from the per protocol analysis presented here. These subjects were included in an intend to treat (ITT) analysis, which can be found in the supplements.



Fig. 4.1. Time line for synbiotic trial.

**Dietary treatments.** The prebiotic, GOS, was obtained from Friesland Foods (sold under the trade name, Vivinal®). This product contained 72.5 % of GOS, 22.8 % lactose, and 4.7 % mono-sugars (galactose and glucose). It was previously established that a dose of 5 g per day of GOS was sufficient to induce a bifidogenic response (40). Therefore the total amount of GOS powder was raised to 6.9 g to achieve a 5 g GOS treatment. The material was packaged in sachets in the Food Processing Product Development Lab (UNL). An additional 0.1 g of lactose was added to achieve the same weight as the other preparations. The sachet material was impermeable to oxygen and moisture.

The two organisms that were used in this feeding study were *Bifidobacterium adolescentis* IVS-1 and *Bifidobacterium animalis* subsp. *lactis* Bb-12. The latter was obtained from Chrs. Hansen as a high cell density powder. Strain IVS-1 was produced from a contract manufacturer (Danwell Technology, Garden Grove, CA). The probiotic powder was stored for up to six month at -18°C and showed stable CFU numbers (data not shown). Probiotic mixtures were portioned into sachets, each contained 0.1 g of cell powder ( $10^{10}$  CFU/g). In addition, 6.9 g of lactose were added as a carrier/control, for a total dose of 7.0 g. Synbiotics contained 6.9 g of Vivinal® and 0.1 g of probiotic (either *B. adolescentis* IVS-1 or *B. animalis* Bb12), for a total dose of 7.0 g. Placebo samples contained 7.0 g of lactose. Subjects were provided with enough samples for the entire length of the study and were instructed to consume each dose on a daily basis, either mixed with food or liquid, but no tab water. The subjects were instructed to store samples in a cold (-18 °C) environment.

**Analysis of fecal microbiota.** Fecal samples were aliquoted and stored at -80°C until further analysis. The DNA was extracted as previously described (29). Amplicon sequencing was performed at the University of Minnesota Genomics Center. All samples were amplified and sequenced in a single run. The V5-V6 region of the 16S rRNA gene was amplified as previously described (1). Quality filtering performed by the University of Minnesota Genomics Center showed that more than 96 % of the samples met all quality control criteria. All reads were trimmed to 240 base pairs using the FASTX-Toolkit. The
reads were quality controlled, merged, OTU clustered, and taxonomically assigned, as previously described (1). If a sample exceeded 37,000 reads it was subsampled using Mothur v.1.31.162 to standardize the sequencing depth across samples. After processing samples contained an average of  $22,012.7 \pm 6,623$  sequences.

Quantitative real-time PCR (qPCR). qPCR was performed by using a Mastercycler Realplex2 instrument (Eppendorf AG, Hamburg, Germany). Each PCR was performed with 25-µl volumes using real-time master mix containing SYBR (5 Prime Inc., Gaithersburg, MD) and either genus-specific primers for Bifidobacterium (40), or the strain-specific primers for *B. adolescentis* IVS-1 (1), as described previously. For strain specific detection of *B. animalis* subsp. *lactis* Bb12 the PCR mixture contained 25 µl of PCR reaction mix (QuantiFast<sup>®</sup> Probe PCR Kit, QIAGEN, Hilden, Germany), 0.3 µmol of each primer (BAL-23S-F 5'-CAGGTGGTCTGGTAGAGTATACCG-3' and BAL-23S-R 5'-ACGGCGACTTGCGTCTTG-3'), 0.25 µmol of probe (BAL-23S-P 5'-FAM-CGCCCACGACCCGCAAG-TAMRA-3'), and 5 µl DNA as previously described (53). The target of these primers and probe is the elongation factor Tu (tuf) gene of Bb12. The specificity of the primers and probe for Bb12 was validated experimentally by qPCR using DNA from 11 different *Bifidobacterium* strains related to strain Bb12. These strains included Bifidobacterium adolescentis IVS-1, Bifidobacterium adolescentis ATCC 15703, Bifidobacterium adolescentis L2-32, Bifidobacterium longum subsp. longum ATCC 15707, Bifidobacterium longum DJO10A, Bifidobacterium longum ATCC 15697, Bifidobacterium longum subsp. longum F8, Bifidobacterium longum subsp. longum JDM301, Bifidobacterium sp. strain 113, Bifidobacterium sp. strain 12\_1\_47BFAA, and

*Bifidobacterium* sp. strain HMLN14. Furthermore, to test if primers could strain specifically select against fecal bacterial communities in humans, baseline DNA samples from subjects in groups C (Bb12 + lactose) and group E (Bb12 + GOS), and 10 randomly selected samples from other subjects, were tested.

Standard curves for absolute quantification of bacterial cell numbers were prepared by using cultures of *B. animalis* subsp. *lactis* Bb12 and *B. adolescentis* IVS-1 grown overnight (14 h).

**Statistical analysis.** All data present in the main body of the text was analyzed based on a per protocol analysis. Subjects that were recruited but violated the protocol in any way, for example took antibiotics, stored the treatments at room temperature, etc., were excluded from the analysis. These subjects were included in an intend to treat (ITT) analysis, which can be found in the supplements. Data is presented as mean  $\pm$  SEM for variables that can be considered normally distributed (or median and range for variables not normally distributed). Group means were compared by ANOVA and post-hoc tests except when data was not normally distributed, in which case nonparametric analyses of medians was done using the Kruskal-Wallis test. Correlation analysis were done using the Spearman's correlation test for parametric analysis. Chi-square tests or Fisher's Exact Tests were used for incidence data. P < 0.05 is considered statistical significant. If only two groups were compared, Student's t tests were performed. For analysis of the gut microbiota the data was normalized by  $\log_{10}$  transformation. To account for type I errors, a false discovery rate was used. A P value of < 0.05 and correlation coefficient (r) values of > 0.50 (in absolute values) were considered significant. Genera above 0.5 % and OTUs above 1 % abundance on average were considered for correlations. Correlation graphs were generated for parameters that showed significant correlations and were visually inspected. If the removal of one single data-point caused the association to become non-significant, the data point was considered an outlier and removed. All analyses used SPSS (Chicago, IL), GraphPad Prism version 5.0 (GraphPad Software, La Jolla, CA, USA), R studio (R Core Team, 2014), or SAS/STAT (SAS Institute Inc., Cary, NC, USA).

#### 4.4 Results

### 4.4.1 Subject demographics and clinical characteristics

The subject cohort for this study was primarily female (71 %), middle-aged, non-Hispanic or Latino, and African American (Table 4.2). All subjects were obese, with participants in group D (IVS-1 + GOS) classifying as extreme obese, in group E (Bb12 + GOS) and F (GOS) as Class II obese, and in group A (Lactose), B (IVS-1+ lactose), and C (Bb12 + lactose) as class I obese. Subjects in group D had a significantly higher body mass index (BMI) than subjects in group B (P = 0.049). The average waist circumference was 40 inches.

### 4.4.2 Test of synergy of GOS in addition to IVS-1 or Bb12.

Our first goal was to determine if the ability of two strains of bifidobacteria to become established in the GI tract of obese adults would be enhanced by the addition of GOS. The strains included *B. adolescentis* IVS-1 that had previously been isolated from a GOS-enriched subject, and *B. animalis* subsp. *lactis* Bb12, a widely used commercial probiotic. Both strains were capable of fermenting GOS *in vitro* (1, 54). Treatment groups

included subjects who consumed each of the two test strains alone (groups B and C), subjects who consumed the synbiotics (i.e., test strains combined with GOS; groups D and E), a GOS-only group (F), and a placebo (lactose) group (A). Strain-specific qPCRs were used to quantify bacterial cell levels for those groups that received either one of the test strains alone or as synbiotics using the corresponding strain specific primers for IVS-1 and Bb12. Because *Bifidobacterium adolescentis* is a common species in the adult GI tract, it was necessary to establish that the IVS-1 primers were specific for this strain.

Results confirmed that no signal was detected in any baseline sample in subjects of group B (IVS-1 alone) or D (IVS-1 + GOS) (Figure 4.2A), or in the baseline of another additional 20 subjects that were randomly selected (data not shown). Therefore, it was concluded that any IVS-1 that was detected by qPCR after the treatment was given to the subjects, was indeed the probiotic strain selected for this study.

A similar approach was used to test the specificity of the Bb12 primers and probe. The primers and probe had been previously designed (53) and are the standard used by Chr. Hansen to identify their probiotic product. The strain Bb12 is a commonly used probiotic in dairy products. It was detected in three subjects before the treatment was started. Analysis of food diaries of these subjects, however, did not identify any indication of Bb12 product consumption.

	Total Sample (n=94)	Group A Control (n=17)	Group B IVS-1 (n=18)	Group C Bb12 (n=164)	Group D IVS- 1+GOS (n=16)	Group E Bb12+GOS (n=17)	Group F GOS (n=16)
Demographic Characteristics							
Gender							
Female, N (%)	71 (75.5)	13 (76.5)	9 (64.3)	12 (75.0)	11 (78.6)	12 (75.0)	12 (75.0)
Age, years, mean ± SD	44.3±11.2	44.3±11.2	44.3±11.2	44.3±11.2	44.3±11.2	44.3±11.2	45.9±9.6
Race, n (%)							
Hispanic/ Latino	9 (9.6)	0 (0.0)	2 (14.3)	2 (14.3)	2 (12.5)	2 (11.8)	1 (6.3)
Non- Hispanic/ Latino	85 (90.4)	17 (100.0)	12 (85.7)	12 (85.7)	14 (87.5)	15 (88.2)	15 (93.8)
Ethnicity, n (%)							
White	31 (33.0)	5 (29.4)	8 (57.1)	4 (28.6)	3 (18.8)	6 (35.3)	5 (31.3)
African American	58 (61.7)	10 (58.8)	6 (42.9)	7 (50.0)	13 (81.3)	11 (64.7)	11 (68.8)
Other <sup>2</sup>	5 (5.3)	2 (11.8)	0 (0.0)	3 (21.4)	0 (0.0)	0 (0.0)	0 (0.0)
Education, n (%)							
$\leq$ 12 years	20 (21.3)	3 (17.6)	1 (7.1)	2 (14.3)	6 (37.5)	2 (11.8)	6 (37.5)
> 12 years	74 (78.7)	14 (82.4)	13 (92.9)	12 (85.7)	10 (62.5)	15 (88.2)	10 (62.5)
Clinical Characteristics <sup>3</sup>							
Body weight, kg <sup>4</sup>	100.0 (25.7)	96.8 (17.7)	94.8 (14.6)	98.5 (32.2)	118.0 (36.7)	112.8 (31.8)	102.3 (18.3)
BMI, kg/m <sup>2,5</sup>	36.7 (8.5)	34.0 (4.5)	33.9 (6.2) A	35.5 (10.3)	41.6 (12.4) B	40.5 (7.1)	36.8 (5.6)
Waist circum- ference, inches	45.0 (7.3)	44.0 (11.0)	43.5 (4.4)	43.0 (9.9)	47.8 (12.3)	47.0 (8.7)	45.0 (3.2)

 Table 4.2. Demographic and metabolic characteristics of study subjects<sup>1</sup>

<sup>1</sup>Sample based on those that were randomized to treatment and completed the post-treatment visit (Visit 4)

<sup>2</sup>White ethnicity includes 2 Hispanic and 1 Middle Eastern participant; Black/African American ethnicity incudes 1 mixed ethnicity participant

<sup>3</sup>All clinical characteristics are listed as median (IQR)

<sup>4</sup>Significant difference based on Kruskal-Wallis, but no differences based on post-hoc pairwise comparisons after adjustment for multiple comparisons.

<sup>5</sup>Different letters indicate significant differences between treatment groups

The results showed that compared to the baseline and washout in the fecal samples, both strains reached significant increases in total numbers (P < 0.001 in group B (IVS-1 + lactose), D (IVS-1 + GOS), and E (Bb12 + GOS); P < 0.01 in group C (Bb12 + lactose)), in the presence as well as absence of GOS (Figure 4.2A). In group B (IVS-1 alone), an average of 6.99 ± 1.2 log<sub>10</sub> of IVS-1 cells g<sup>-1</sup> was detected and in group D, receiving the synbiotic IVS-1 plus GOS, cell levels reached 7.22 ± 1.6 log<sub>10</sub> of IVS-1 cells g<sup>-1</sup>. Group C received the commercial strain Bb12, which was detected at absolute numbers of 5.83 ± 0.7 log<sub>10</sub> of Bb12 cells g<sup>-1</sup> and in group E (Bb12 + GOS) Bb12 numbers reached 6.11 ± 0.7 log<sub>10</sub> of Bb12 cells g<sup>-1</sup>.

The numbers of the two probiotic strains for each treatment at each time point were then compared directly (Figure 4.2B). IVS-1 colonized the GI tract at a significantly higher number than the commercial probiotic Bb12 in the probiotic only treatments (P = 0.0056). Although GOS enriched for both strains, Bb12 and IVS-1, relative to the test strains alone, in both cases this trend was not significant (P = 0.7382 and P = 0.3034, respectively). In contrast, when comparing the two synbiotics, GOS significantly increased the number of IVS-1 by more than one log compared to the commercial Bb12 synbiotic  $(7.22 \pm 1.6 \log_{10})$ and 6.11  $\pm$  0.7 log<sub>10</sub> respectively, P = 0.0195). This result demonstrates an ecological advantage of IVS-1 over Bb12 and a limited effect of GOS supplementation on either strain. After a four week washout period, Bb12 could not be detected in group C subjects, and only in three subjects at a very low number in the synbiotic group (group E). One of these subjects already had Bb12 present in the baseline sample. In the IVS-1 + GOSsynbiotic group D, IVS-1 persisted in six subjects during the washout period, and the number of IVS-1 was significantly higher than the number of Bb12 in group E (P =0.0057).

In addition to strain-specific quantification, the absolute numbers of bifidobacteria were measured for all six groups at the three time points by genus-specific quantitative real time PCR (Figure 4.3A). Even though both *Bifidobacterium* strains were significantly enriched as determined by strain-specific qPCR, neither IVS-1 nor Bb12 supplementation increased the total number of bifidobacteria. Only groups D (IVS-1 + GOS) and F (GOS) showed a significant increase of bifidobacteria due to the treatments (P = 0.0203 and P = 0.0191, respectively).

The *Bifidobacterium* numbers at the baseline varied greatly within each group, ranging from the detection limit ( $\log_{10} 4.67$  cells g<sup>-1</sup> feces) to  $\log_{10} 10.42$  cells g<sup>-1</sup> feces in group F (GOS). Therefore, the change in cell number was calculated for each subject (Figure 4.3B). The highest increases in *Bifidobacterium* numbers were for group F (GOS) with  $1.30 \pm 1.7 \log_{10}$ , followed by group B (IVS-1 only) with  $1.22 \pm 1.4 \log_{10}$ . Interestingly, the lactose group also had an increase in bifidobacteria ( $0.51 \pm 0.9 \log_{10}$ ). However, none of these differences reached statistical significance.





**Fig. 4.2.** A. Test of *in vivo* selected synergistic synbiotic application compared to a commercial synbiotic. Quantification of absolute cell numbers of bifidobacteria in fecal samples by qPCR using strain-specific primers for strains *B. adolescentis* IVS-1 and *B. lactis* Bb12. Shown are probiotic and synbiotic treatment groups at baseline, treatment and washout time points. Significance of  $P \le 0.05$  is denoted by a single asterisk (\*),  $P \le 0.01$  as two asterisks (\*\*), and  $P \le 0.001$  by three asterisks (\*\*\*). B. Direct comparison of abundances of *B. adolescentis* IVS-1 and *B. lactis* Bb12 at each time point. Different letters indicate significant differences between groups ( $P \le 0.05$ ).





**Fig. 4.3.** A. Quantification of absolute cell numbers of bifidobacteria in fecal samples by genus-specific qPCR. B. Change in abundance of bifidobacteria for each subject due to treatment consumption.

# 4.4.3 Community-wide characterization of effects on gut microbiota confirms ecological advantage of GOS and IVS-1 compared to Bb12 and GOS.

Community-wide changes that were introduced by the dietary treatments to the resident gut microbiota were assessed by sequencing 16S rRNA tags. Interestingly, only one phylum was significantly influenced by the dietary treatments as *Actinobacteria* was significantly higher in subjects treated with IVS-1 + lactose (group B, P = 0.0181, Table 4.2, Figure 4.S1), and the genus *Bifidobacterium* in groups B (IVS-1 + lactose) and C (Bb12 + lactose). Apart from this, only at the OTU level were significant changes introduced to the gut microbiota. However, the abundance of these OTUs was extremely low in most cases, and high variations in the sample populations were also observed. Accordingly, the alpha and beta diversity of each group and between groups was not significantly different (data not shown).

The ability of the test strains (alone and as synbiotics) to become established in the GI tract was based on the abundance of operational taxonomic units (OTU) representing the species *B. adolescentis* (OTU\_1) and *B. lactis* subsp. *animalis* (OTU\_167) (Table 4.3). *B. adolescentis* is an autochthonous species in the human GI tract and was detected at low average abundances in the GI tract of the subjects at baseline  $(2.27 \pm 4.7 \%)$ . In contrast, *B. animalis* subsp. *lactis* was detected in baseline samples for only seven subjects. In four of these subjects the abundance of OTU\_167 was below 0.01 %. The other three subjects are identical to the ones identified in the qPCR analysis (Figure 4.2).

Based on the sequencing analysis,  $OTU_1$  was significantly enriched by the probiotic-alone and synbiotic treatments to 3.7 and 7.3 %, respectively (Table 4.2). In groups D (IVS-1 + GOS), E (Bb12 + GOS), and F (GOS)  $OTU_1$  *B. adolescentis* became

the most abundant OTU, representing an average of 7.3 %, 7.2 %, and 6.7 % of the microbiota, respectively, and the second most abundant in group B (3.4 %). This increase was clearly based on the presence of IVS-1 and/or GOS in the treatments, as the abundance of OTU\_1 *B. adolescentis* was only 0.01 % in group A (Lactose) and 1.2 % in group C (Bb12 + lactose). A comparison of the abundance of OTU\_1 between group B (IVS-1 + lactose), C (IVS-1 + GOS), and group F (GOS) showed no significant difference (P > 0.1). There was a significant increase in the relative abundance of OTU\_167 *B. animalis* between baseline and treatment in group C (Bb12 + lactose) and E (Bb12 + GOS) (0.4 % and 0.1 %, respectively).

# 4.4.4 Systematic analyses of members of the gut microbiota.

To identify potential interactions between the test strains and other members of the gut microbiota, we performed correlation analyses at the treatment time point. As a cutoff, the genus had to have an average abundance of at least 0.5 %, and the OTU an average abundance of at least 1 %, with the exception of OTU\_167 (*B. animalis*) which was included even though its average abundance was below 1 %. No significant correlations between the genus *Bifidobacterium*, or any *Bifidobacterium* OTUs, and other genera or OTU could be identified once outliers were removed. However, several other significant correlations between other members of the gut microbiota were observed (Figure 4.S2 and 4.S3).

		Mean % bacterial abundance $\pm$ SD <sup>b</sup>						
Treatments	Taxonomic Group	Baseline	Treatment	Washout	P value			
Group A	OTUs <sup>a</sup>							
Lactose control	OTU_2050 (Ruminococcus2 sp.)	0.01±0.0 AB	0.01±0.0 A	0.02±0.0 B	0.0139			
Group B	Phylum							
IVS-1 Probiotic	Actinobacteria	9.16±5.5 A	15.44±6.2 B	14.39±5.9 AB	0.0181			
	Genus							
	Bifidobacterium	7.85±7.8 A	14.66±7.2 B	13.43A±6.9 B	0.0378			
	OTUs <sup>a</sup>							
	OTU_2353 (Dorea sp.))	0.01±0.0 A	0.02±0.0 AB	0.02±0.0 B	0.0068			
	OTU_1 (B. adolescentis)	1.30±4.4 A	3.69±5.2 B	3.56±4.7 B	0.0010			
Group C	Genus							
Bb12 Probiotic	Bifidobacterium	9.53±5.3 A	10.18±8.3 B	9.96±6.1 AB	0.0378			
	OTUs <sup>a</sup>							
	OTU_167 (B. animalis subsp. lactis)	0.04±0.1 AB	0.04±0.1 A	0.00±0.0 B	0.0328			
Group D IVS-1 GOS	OTUs <sup>a</sup>							
	OTU_469 (Anaerotruncus sp.)	0.03±0.1 A	$0.00\pm0.0$ B	0.01±0.0 AB	0.0188			
	OTU_102: (Lachnospiracea incertae sedis sp.)	0.35±0.5 A	0.01±0.2 B	0.24±0.6 AB	0.0444			
	OTU_1 (B. adolescentis)	2.48±4.5 A	7.34±7.2 B	4.75±5.1 B	0.0343			
Group E Bb12 GOS	OTUs <sup>a</sup>							
	OTU_167 (B. animalis subsp. lactis)	0.00±0.0 A	0.12±0.2 B	0.01±0.0 A	0.0043			
	OTU_1800 (unclass. Lachnospiraceae sp.)	0.00±0.0 AB	0.01±0.0 A	0.00±0.0 B	0.0241			
Group F	OTUs <sup>a</sup>							
GOS	OTU_643 (Bacteroides sp.)	0.02±0.0 A	0.00±0.0 B	0.00±0.0 AB	0.0372			
	OTU_319 (Butyricimonas sp.)	0.00±0.0 AB	0.00±0.0 A	0.05±0.1 B	0.0382			

Table 4.3. Proportions of bacterial taxa significantly influenced by dietary treatments

<sup>a</sup> If the strain could not be assigned to a type strain (< 97 % homology), RDP Classifier was used to determine the most likely genus (80 % cutoff). <sup>b</sup> Values with different uppercase letters are significantly different from each other.

The ratio of *Prevotella* to *Bacteroides* has been previously suggested to be strongly associated with diet, especially diets rich in animal fat (*Bacteroides*) versus carbohydrates (*Prevotella*) (55). Other studies have reported a significant change in the ratio between *Prevotella* and *Bacteroides* due to dietary treatments (56), but an analysis of the *Prevotella* and *Bacteroides* ratio in this study showed no significant difference within a treatment group, or when groups were compared (Figure 4.S4). Furthermore, the abundance of butyrate producing genera such as *Faecalibacterium*, *Eubacterium*, *Roseburia*,

*Lachnobacterium*, and *Ruminococcus* was not significantly influenced by any of the treatments, nor was their combined abundance changed by any of the treatments (data not shown).

# **4.4.5** Systematic analyses of subjects that showed IVS-1 persistence after treatment termination.

We observed that for nine subjects IVS-1 persisted during the four week washout period. Three subjects were from group B and six were from group D (IVS-1 alone and IVS-1 + GOS, respectively). The microbiota of these nine persisters was compared with the microbiota of all the other subjects in groups B and D (referred to as non-persisters). The aim of this analysis was to determine if the persister status could be predicted before the treatment had begun (i.e., from the baseline samples), based on the composition of the gut microbiota. In order to identify genera and OTUs that affected persistence of IVS-1 in the GI tract of these two groups, a Random Forest classification was performed. This analysis shows the importance that the relative abundances of different taxa have in predicting persistence. Output of this analysis is a "value of mean decrease in accuracy". The higher the value of mean decrease in accuracy of the taxa, the stronger the prediction of persistence of IVS-1. All genera and OTUs with a value of mean decrease in accuracy of at least one are reported here (Figure 4.4). The random forest analysis identified *Slackia* as the most important predictor of persistence. When the abundance of *Slackia* was compared between persisters and non-persisters, Slackia abundance was higher in nonpersisters with a ratio of almost 200. However, despite the high ratio, the absolute abundance of *Slackia* was only 0.23 % in non-persisters and almost absent in persisters.

Other members of the *Coriobacteriales* order are also predictive for the ability of IVS-1 to persist in the subjects, including Asaccharobacter, Collinsella, and Olsenella. Except for Asaccharobacter, all of these Coriobacteriales were present at a higher abundance in nonpersisters than in persisters. Three members of the *Clostridiales* order, *Ruminococcus*, Eubacterium and Mogibacterium, one member of the Bacteroides order, Prevotella, and one member of the *Erysipelotrichales* order, *Holdemania*, also appeared to have the greatest impact on IVS-1 persistence. At the OTU level, eight OTUs had a value of mean decrease in accuracy of at least one; six belonged to the order *Clostridiales*, one to Coriobacteriales, and one to Lactobacillales. The prediction value of the OTUs for persistence appears to be strain dependent. For example, OTU\_21 Blautia sp. and OTU\_98 *Blautia* sp. were present in higher abundance in persisters then in non-persisters, while OTU\_103 *Blautia* sp. had a significantly higher abundance in non-persisters (P = 0.0353). Three Lachnospiraceae OTUs were also identified. OTU\_61 Lachnospiraceae incertea sedis and OTU 2093 Lachnospiraceae incertea sedis share 96 % identity and are both in higher abundance in persisters than in non-persisters (P = 0.0365 and P = 0.0805, respectively). OTU\_76, an unclassified *Lachnospiraceae* sp., on the other hand was present in significantly higher numbers in non-persisters (P = 0.0177) and shared 91 % and 90 % identity with OTU\_61 and OTU\_2093, respectively. OTU\_1 B. adolescentis was only present in two of the persister subjects at the baseline and was not detected by the Random Forest analysis (< -1.0).

Even though no *Bifidobacterium* species had a Random Forest importance value above 1, three *Bifidobacterium* OTUs had values above zero and were further analyzed (Figure 4.6A). Those three OTUs consisted of OTU\_2055 *B. pseudocatenulatum* (99% identity, e-value  $6 \cdot 10^{-120}$ , 100% query cover), OTU\_2202 *B. pseudocatenulatum* (98% identity, e-value  $2 \cdot 10^{-119}$ , 100% query cover), and OTU\_7 *B. ruminantium/adolescentis* (100 % identity, e-value  $4 \cdot 10^{-126}$ , 100% query cover in each case according to NCBI). OTU\_7 had a 144-fold higher relative abundance in non-persisters than in persisters on average (*P* = 0.0454), even though it was not present in all non-persisters at the baseline. OTU\_7 is very closely related to IVS-1 (98 % identity) and possibly the same species (Figure 4.6B). OTU\_2055 *B. pseudocatenulatum* and OTU\_2202 *B. pseudocatenulatum*, were higher in persisters by a ratio of 24 and 5.5, respectively, but their abundance was not significantly different between the groups.



**Fig. 4.4.** Prediction analysis of IVS-1 persisters and non-persisters. Random Forest variable importance plots and relative abundance of genera (A) and OTUs (B) with a variable importance of at least 1 in mean decrease in accuracy. The ratio of the given taxa between responders and non-responders is shown on the right. Taxa are color coded by order.



**Fig. 4.5.** A. Prediction analysis of IVS-1 persisters and non-persisters. Random Forest variable importance plots and relative abundance of *Bifidobacterium* OTUs with a variable importance of at least zero in mean decrease in accuracy. B. The ratio of the given *Bifidobacterium* OTUs between responders and non-responders.

While different markers for alpha diversity such as Shannon index, Simpson index, and the number of observed species was not significantly different among persisters and non-persisters in the baseline sample (Figure 4.S5), the beta diversity of the samples tended to separate (Figure 4.6). However, this trend was not significant (P = 0.2637)



**Fig. 4.6.** NMDS plot of beta diversity analysis based on Bray-Curtis distance of baseline samples between IVS-1 persisters and non-persisters.

## 4.4.6. Analysis of anthropometric markers and gastrointestinal symptoms

Anthropometric markers were analyzed at the baseline and after three weeks of treatment, and the change (as percent) is reported in Table 4.4. There was no change detected in anthropometrics between groups.

	Total Sample (n=94)	Group A Lactose (n=17)	Group B IVS-1 (n=14)	Group C Bb12 (n=14)	Group D IVS-1 + GOS (n=16)	Group E Bb12 + GOS (n=17)	Group F GOS (n=16)
Anthropome	trics						
Body weight, kg	0.4 (2.9)	0.4 (5.0)	-0.1 (2.9)	0.6 (2.4)	-0.3 (3.4)	-0.1 (3.6)	1.3 (3.6)
BMI, kg/m <sup>2</sup>	0.4 (2.9)	0.4 (5.0)	-0.1 (2.9)	0.6 (2.4)	-0.3 (3.4)	-0.1 (3.6)	1.3 (3.6)
Waist circum- ference, inches	0.0 (4.6)	-1.3 (3.8)	0.0 (6.9)	0.2 (5.2)	0.0 (5.5)	1.8 (3.0)	0.0 (5.1)

**Table 4.4.** Percent change in anthropometrics in participants compared to the baseline<sup>1</sup>

<sup>1</sup>Per-protocol sample based on those that were randomized to treatment and were considered compliant to the treatment

<sup>2</sup>All clinical characteristics are listed as median (IQR)

A structured 34-item questionnaire with a Likert scale from 0 (no symptoms) to 10 (very severe symptoms) was used to assess gastrointestinal symptoms at baseline and any changes in these symptoms with supplementation. At baseline, subjects reported no symptoms to 28 of 34 (82.4 %) GI symptoms included on the questionnaire; this number of symptoms increased to 85.3 % (29 of 34 items) at the end of the treatment. The most common symptoms at baseline were bloating, passing gas, hard stools, and watery stools, with 60.6 %, 85.1 %, 46.8 %, and 43.6 % overall indicating presence of these symptoms, respectively. The median symptom score for each of these four symptoms both before and after treatment are listed in Table 4.5. Overall, low median scores indicated that most subjects either had minimal GI symptoms or low severity of that symptom. No significant differences existed between median symptom score for the six groups at baseline. At the end of treatment, those in the GOS group had significantly more hard stools than the Bb12 + GOS group (Kruskal-Wallis; P = 0.024). Passing gas increased from a median of 2.0 to 5.0 with lactose supplementation, but this was not significantly different (Mann-Whitney,

P = 0.15). The severity of passing gas significantly reduced from 4.0 to 1.0 in the Bb12 + GOS group (P = 0.040), and severity of hard stools increased from 1.0 to 3.5 in the GOS group (Wilcoxon Signed Ranks Test, P = 0.030).

	Total Sample (n=92)	Group A Lactose (n=16)	Group B IVS-1 (n=14)	Group C Bb12 (n=14)	Group D IVS-1 + GOS (n=15)	Group E Bb12 + GOS (n=17)	Group F GOS (n=16)	
Baseline <sup>2</sup>								
Bloating	2.0 (4.8)	2.0 (4.8)	1.0 (3.3)	2.0 (4.0)	2.0 (5.0)	4.0 (6.0)	2.5 (7.8)	
Passing Gas	3.0 (5.0)	2.5 (4.8)	2.5 (3.5)	2.5 (4.5)	3.0 (5.0)	4.0 (3.5) *	2.5 (5.8)	
Hard Stools	1.0 (2.8)	0.5 (3.0)	1.0 (2.3)	0.0 (2.3)	0.0 (3.0)	1.0 (3.5)	1.0 (3.8) *	
Watery Stools	1.0 (4.8)	1.0 (3.8)	1.0 (3.5)	1.5 (3.5)	2.0 (5.0)	1.0 (4.5)	2.0 (5.8)	
Treatment End <sup>3,4</sup>								
Bloating	1.0 (4.0)	0.0 (4.0)	1.5 (4.0)	1.0 (2.0)	1.0 (3.0)	2.0 (4.0)	2.5 (5.0)	
Passing Gas	3.0 (5.0)	5.0 (4.8)	3.0 (2.0)	3.0 (6.3)	2.0 (3.0)	1.0 (5.0) *	3.0 (5.8)	
Hard Stools	1.0 (3.0)	0.0 (3.0)	1.0 (1.3)	1.0 (3.3)	1.0 (3.0)	0.0 (1.0) A	3.5 (4.8) B *	
Watery Stools	1.0 (3.0)	1.0 (2.0)	1.0 (3.3)	0.5 (1.5)	1.0 (4.0)	1.0 (2.5)	2.5 (4.8)	

**Table 4.5.** Gastrointestinal symptoms by supplementation group<sup>1,2,3</sup>

<sup>1</sup>Per-protocol sample based on those that were randomized to treatment and were considered compliant to the treatment

<sup>2</sup>Change in gastrointestinal symptoms are listed as median (IQR). Only the most common symptoms experienced are listed

<sup>4</sup>Different letters indicate a significant difference in distribution between groups within symptom; identical symbol indicate differences in symptoms before and after treatment within a treatment group

#### **4.5 Discussion**

There is much interest in the health promoting capabilities of bifidobacteria in the human GI tract. Synbiotics have advantages in promoting bifidobacteria as they could result in improved establishment of a specific *Bifidobacterium* strains in the human gut when compared to the probiotic alone, and increase bifidobacteria in individuals that do not possess them or that do not respond to probiotics alone. However, if synbiotics are in fact more successful than their parts has not adequately been studied in humans. Here we

systematically assessed the ability of two *Bifidobacterium* strains, administered alone and combined with GOS as synbiotics, to become established in the gastrointestinal tract of obese adults. In addition, community sequencing was used to identify other changes introduced to the gastrointestinal microbiota by the dietary treatments.

The first objective of this study was to test if the prebiotic carbohydrate GOS included in the two synbiotic preparations supported the establishment of the test probiotic strains in the human gut. Both strains were significantly enriched during the treatment period compared to baseline and washout levels, whether consumed alone or as synbiotics according to the qPCR analysis (Figure 4.2). The absolute numbers of IVS-1 were significantly higher than those detected for Bb12, independently of the presence of GOS. This suggests that the autochthonous strain IVS-1 has an ecological advantage over the allochthonous commercial strain Bb12. This is an important finding as it emphasizes the necessity to consider ecological requirements of the probiotic strains when synbiotic combinations are formulated. Bb12 and IVS-1 are both capable of utilizing GOS in vitro (1, 54), however enhanced Bb12 colonization of the human GI tract is not supported by GOS as demonstrated here, and as also previously reported (48–50). There was a tendency of IVS-1 being specifically enriched by GOS. However, this synergistic effect between IVS-1 and GOS did not reach significance in this study cohort. Interestingly, the synbiotic of IVS-1 and GOS led to significantly higher numbers of IVS-1 than the commercial synbiotic of Bb12 and GOS. This indicates that IVS can be used to formulate a synergistic synbiotic that can enhance population levels and the competitiveness of a putative probiotic strain in the gastrointestinal tract compared to a commercial synbiotic. Indeed, this is only the second report of a prebiotic specifically enriching for a putative probiotic strain in humans (57).

Secondly, we compared the ability of the two strains and their respective synbiotic combinations to alter the composition of the gut microbiota in obese individuals. Our analyses showed that the probiotic, prebiotic, and synbiotic treatments altered the gut microbiota of the study subjects to a very limited extent (Table 4.3). Only in treatment group B (IVS-1 + lactose) the phylum Actinobacteria was significantly higher in the treatment sample than at the baseline, and the genus *Bifidobacterium* was significantly higher in groups B (IVS-1 + lactose) and C (Bb12 + lactose) compared to the baseline based on 16S rRNA sequencing. One limitation of this sequencing approach is that it returns the relative abundance of taxa, instead of the absolute numbers present in the GI tract. Therefore a qPCR approach was done, which is more quantitative. This analysis showed that the absolute number of bifidobacteria was significantly increased by the IVS-1 + GOS synbiotic (group D) and GOS alone (group F) (Figure 4.3). This increase in numbers and abundance of bifidobacteria due to GOS feeding has been previously shown (1, 36). Interestingly, however, the treatment of Bb12 and GOS did not significantly enrich for bifidobacteria. This may be caused by a large variance in the subject cohort's microbiota, as there was no other member of the gut microbiota identified that was enriched by this treatment and could have outcompeted the bifidobacteria for GOS. While both strains, IVS-1 and Bb12, were significantly enriched by the probiotic dietary treatments in group B (IVS-1 + lactose) and C (Bb12 + lactose) (Figure 4.2), but the absolute number of bifidobacteria was not increased in these groups (Figure 4.3).

According to the 16S rRNA sequencing analysis, in all other cases only OTUs were significantly influenced in their abundances by the treatments (Table 4.2). This overall resilience of the gut microbiota to the treatments was characterized by a great variability between the subjects. For example, the bifidogenic response to GOS treatment varied

between an increase of almost 5 logs in one subject and a decrease of 0.6 logs in a different individual. This responder and non-responder phenomenon was previously described for GOS treatment in healthy adult subjects (40), and was also observed in this study, despite the presence of the added probiotic strain. The stability of the gut microbiota was also confirmed by the lack of change in the alpha and beta diversities, no change in the abundance of butyrate producers, and a consistency of the *Prevotella* and *Bacteroides* ratio. Few significant correlations were found between members of the gut microbiota, and none between bifidobacteria and any other taxa.

Overall these analyses showed that a strain of a core species of the human gut microbiota, *B. adolescentis*, can be established in almost all of the subjects by probiotic and synbiotic consumption, and at significantly higher numbers than an allochthonous strain. However, this establishment had little effect on the resident community in the GI tract.

Strain-specific qPCR analysis at the four week washout time point showed that IVS-1 did not only reach higher colonization levels compared to Bb12, but IVS-1 was also significantly more persistent than Bb12 (Figure 4.2 B). This finding supports the concept of *in vivo* selection to select for ecologically more competitive probiotic strains. The autochthonous strain IVS-1 had a significant advantage to become established in the GI tract and avoid niche exclusion from the resident microbiota compared to the allochthonous strain Bb12.

Interestingly, IVS-1 persisted in twice as many subjects when consumed as a synbiotic with GOS compared to IVS consumption alone. It has to be taken into consideration, however, that the number of subjects that showed persistence was very low,

only three in group B (IVS-1 + lactose) and six in group D (IVS-1 + GOS), and the washout sample was taken four weeks after the last consumption of the treatment. This study was not designed to specifically test persistence and a strain detection at only four weeks after the end of the treatment does not allow any conclusion about long-term persistence of IVS-1. In this study the addition of GOS did not significantly support the persistence of IVS-1, but because of the small sample size a final conclusion about the synergistic effect of GOS and IVS-1 in regard to persistence cannot be drawn. A future study may consider to extend the consumption of GOS after IVS-1 consumption has been terminated in order to establish if GOS specifically supports IVS-1 colonization and competitive fitness.

In order to get insight into the ecological niche that IVS-1 may be occupying in the GI tract of persisters, the composition of the gut microbiota was compared between persisters and non-persisters in order to determine if a persistence of IVS-1 could be predicted before the beginning of the treatment. By Random Forest analysis, mostly members of the Coriobacteriaceae family were identified (Figure 4.4). Coriobacteriaceae are frequently found in patients suffering from overweight (58) or inflammatory bowel diseases, but there is no corresponding quantitative or functional data available yet (59). *Slackia* was identified as the most important genus predictor of persistence (Figure 4.4). The genus *Slackia* is part of the family *Coriobacteriaceae* and contains five species (60– 63). In vitro analyses suggest that none of these species are capable of utilizing GOS (62, 63). Additionally, *Slackia* has been characterized as asaccharolytic (as well as Eubacterium and Mogibacterium) (64). Therefore it is unlikely that Slackia would have been competing with IVS-1 for GOS. This genus is known as a commensal of the mammalian microbiota (59) and two of its members, S. isoflavoniconvertens and S. equolifaciens. are known equol producers (63). Whether or not the Slackia detected here produces equol remains unknown. Overall *Slackia* is present at very low abundance in non-persisters (0.2 %) but almost absent in persisters.

Other members of the *Coriobacteriaceae* family, *Asaccharobacter*, *Collinsella*, and *Olsenella*, were also predictors of persistence. OTU\_3 Collinsella aerofaciens and the genus *Collinsella* were both present at approximately 3 % abundance in persisters. Therefore the genus *Collinsella* mostly likely consists only of OTU\_3 *Collinsella aerofaciens* in this case. This species is considered to be a member of the core human gut microbiome (65). Interestingly, the identification of *Collinsella*, *C. aerofaciens*, and *Coriobacteriaceae* has been consistent in comparison to another study (Maldonado-Gomez et al., unpublished). In that case, these taxa were predictive of long-term persistence for *B. longum* in healthy human subjects. In this study no significant correlation between IVS-1 or bifidobacteria and any of these taxa could be identified that could explain this phenomenon. For now it remains speculative which effect the presence of *Coriobacteriaceae* has on the persistence of IVS-1.

Three *Blautia* OTUs were identified by Random Forest. Two of the three *Blautia* OTUs were in higher abundance in persisters than in non-persisters. *Blautia* has not been reported to utilize GOS, but a strong positive correlation and potentially cross-feeding between *Blautia* and *Bifidobacterium* was previously reported (1). On the other side, OTU\_23, a *Streptococcus* sp., was also identified to be predictive of IVS-1 persistence. It has been previously shown that this species is capable of utilizing complex carbohydrates like GOS (66), so this species could have been competing with IVS-1. However, this OTU was in significantly higher abundance in persisters than in non-persisters (P = 0.0266). Again, the number of persister subjects was very low for this analysis and, unfortunately,

for now it cannot be explained how the taxa identified by Random Forest shape the GI niche that can be occupied by IVS-1.

Interestingly, three *Bifidobacterium* OTUs were identified as predictors of IVS-1 colonization (Figure 4.5). While the mean decrease in accuracy values were quite low and therefore these results have to be considered with caution, two OTUs that potentially contributed to IVS-1 colonization were identified (both *B. pseudocatenulatum*), and one that was possibly preventing it (OTU\_7). While the identity of OTU\_7 could not be clearly established, OTU\_7 was a very closely related strain to IVS-1 with 98 % identity. Therefore, these two strains may have very similar ecological niche preferences and requirements. Based on niche exclusion theory (67) it could be possible that OTU\_7 was occupying a niche in the GI tract that could have been occupied by IVS-1 as well, but not by both strains at the same time. IVS-1 was not capable of outcompeting the closely related resident strain in this case.

## 4.6 Conclusion

This study has provided novel insights into the complex interactions between the gut microbiota and dietary regimens consisting of prebiotics, probiotics, and synbiotics. The potential of synbiotic applications on human health has been previously established (45). However, many synbiotic formulas lack synergistic activity in that the probiotic is not enriched by the prebiotic. Thus, the probiotic strain is established in the GI tract not more effectively had it been introduced by itself. While this lack of success may be due to the arbitrary selection of the synbiotic components, we have previously shown that *in vivo* selection can overcome the ecological limitations imposed on the probiotic strains in the

GI tract (1). By specifically enriching a *B. adolescentis* strain, IVS-1, with its cognate prebiotic substrate, we have previously validated this synergistic synbiotic concept in rats (1). We have now demonstrated the potential of *in vivo* selected probiotics and synbiotics in a human trial as well. IVS-1 was successfully established in the GI tract of human subjects in significantly higher numbers than a commercial probiotic, *B. animalis* subsp. *lactis* Bb12, and also when the two synbiotic combinations were compared. IVS-1, as a strain of a core species of the human gut microbiota, was established in all but one subject, and OTU\_1 *B. adolescentis* became the dominant member of the gut microbiota. This study has clearly established that *in vivo* selection can identify autochthonous probiotic strains that are highly competitive in the GI environment when introduced as probiotics or synbiotics. While there was a trend for GOS specifically and synergistically enriching for IVS-1 in this study, this trend did not reach significance.

This study provided important proof of concept that a rational selection of synbiotic combinations based on the ecological requirements of the probiotic strain can significantly enhance the colonization, and persistence of probiotic strains.

## **4.7 References**

- Krumbeck JA, Maldonado-Gomez MX, Martínez I, Frese SA, Burkey TE, Rasineni K, Ramer-Tait AE, Harris EN, Hutkins RW, Walter J. 2015. *In Vivo* Selection to identify bacterial strains with enhanced ecological performance in synbiotic applications. Applied and Environmental Microbiology 81:2455–2465.
- Joyce SA, Gahan CGM. 2014. The gut microbiota and the metabolic health of the host. Current Opinion in Gastroenterology 30:120–7.
- Cho I, Blaser MJ. 2012. The human microbiome: at the interface of health and disease. Nature Reviews Genetics 13:260–270.
- Guarner F, Malagelada J. 2003. Gut flora in health and disease. Lancet 361:512– 519.
- Musso G, Gambino R, Cassader M. 2010. Gut microbiota as a regulator of energy homeostasis and ectopic fat deposition: mechanisms and implications for metabolic disorders. Current Opininion in Lipidology 21:76–83.
- 6. Cani PD, Neyrinck AM, Fava F, Knauf C, Burcelin RG, Tuohy KM, Gibson GR, Delzenne NM. 2007. Selective increases of bifidobacteria in gut microflora improve high-fat-diet-induced diabetes in mice through a mechanism associated with endotoxaemia. Diabetologia 50:2374–2383.
- Fujimori S, Gudis K, Mitsui K, Seo T, Yonezawa M, Tanaka S, Tatsuguchi A, Sakamoto C. 2009. A randomized controlled trial on the efficacy of synbiotic versus probiotic or prebiotic treatment to improve the quality of life in patients with ulcerative colitis. Nutrition 25:520–525.

- Ishikawa H, Matsumoto S, Ohashi Y, Imaoka A, Setoyama H, Umesaki Y, Tanaka R, Otani T. 2011. Beneficial effects of probiotic *Bifidobacterium* and galacto-oligosaccharide in patients with ulcerative colitis: A randomized controlled study. Digestion 84:128–133.
- Cappello C, Tremolaterra F, Pascariello A, Ciacci C, Iovino P. 2013. A randomised clinical trial (RCT) of a symbiotic mixture in patients with irritable bowel syndrome (IBS): Effects on symptoms, colonic transit and quality of life. International Journal of Colorectal Disease 28:349–358.
- Bittner AC, Croffut RM, Stranahan MC, Yokelson TN. 2007. Prescript-Assist TM probiotic-prebiotic treatment for irritable bowel syndrome: An open-label, partially controlled, 1-year extension of a previously published controlled clinical trial. Clinical Therapeutics 29:1153–1160.
- Dughera L, Elia C, Navino M, Cisaro F, Aimo G, Bertele A, Calabrese C, De Felici I, Diligente G, Febbraro I, Franze A, Fries W, Martorano M, Scifo V, Villardo L. 2007. Effects of symbiotic preparations on constipated irritable bowel syndrome symptoms. Acta Biomedica de l'Ateneo Parmense 78:111–116.
- 12. Andriulli A, Neri M, Loguercio C, Terreni N, Merla A, Cardarella MP, Federico A, Chilovi F, Milandri GL, De Bona M, Cavenati S, Gullini S, Abbiati R, Garbagna N, Cerutti R, Grossi E. 2008. Clinical trial on the efficacy of a new symbiotic formulation, Flortec, in patients with irritable bowel syndrome: a multicenter, randomized study. Journal of clinical gastroenterology 42 Suppl 3:S218–223.
- 13. Savaiano DA, Ritter AJ, Klaenhammer TR, James GM, Longcore AT,

**Chandler JR**, **Walker WA**, **Foyt HL**. 2013. Improving lactose digestion and symptoms of lactose intolerance with a novel galacto-oligosaccharide (RP-G28): a randomized, double-blind clinical trial. Nutrition journal **12**:160–169.

- Pakdaman MN, Udani JK, Molina JP, Shahani M. 2015. The effects of the DDS1 strain of *Lactobacillus* on symptomatic relief for lactose intolerance a randomized, double-blind, placebo-controlled, crossover clinical trial. Nutrition Journal 15:56–67.
- Asemi Z, Khorrami-Rad A, Alizadeh SA, Shakeri H, Esmaillzadeh A. 2014. Effects of synbiotic food consumption on metabolic status of diabetic patients: A double-blind randomized cross-over controlled clinical trial. Clinical Nutrition 33:198–203.
- 16. Fernandes R, Beserra BTS, Mocellin MC, Kuntz MGF, da Rosa JS, de Miranda RCD, Schreiber CSO, Froede TS, Nunes EA, Trindade EBSM. 2016. Effects of Prebiotic and Synbiotic Supplementation on Inflammatory Markers and Anthropometric Indices After Roux-en-Y Gastric Bypass. A randomized, Tripleblind, Placebo-controlled Pilot Study. Journal of Clinical Gastroenterology 50:208– 217.
- Dehghan P, Pourghassem Gargari B, Asghari Jafar-abadi M, Aliasgharzadeh
   A. 2014. Inulin controls inflammation and metabolic endotoxemia in women with
   type 2 diabetes mellitus: a randomized-controlled clinical trial. Journal of Food
   Sciences and Nutrition 65:117–123.
- Pourghassem Gargari B, Dehghan P, Aliasgharzadeh A, Asghari Jafar-Abadi
   M. 2013. Effects of high performance inulin supplementation on glycemic control

and antioxidant status in women with type 2 diabetes. Diabetes Metab J **37**:140–148.

- Shoelson SE, Goldfine AB. 2009. Fanning the flames of obesity-induced inflammation. Nature medicine 15:373–374.
- Geurts L, Neyrinck AM, Delzenne NM, Knauf C, Cani PD. 2014. Gut microbiota controls adipose tissue expansion, gut barrier and glucose metabolism: Novel insights into molecular targets and interventions using prebiotics. Beneficial Microbes 5:3–17.
- 21. Serino M, Luche E, Gres S, Baylac A, Berge M, Cenac C, Waget A, Klopp P, Iacovoni J, Klopp C, Mariette J, Bouchez O, Lluch J, Ouarne F, Monsan P, Valet P, Roques C, Amar J, Bouloumie A, Theodorou V, Burcelin R. 2011. Metabolic adaptation to a high-fat diet is associated with a change in the gut microbiota. Gut 61:543–553.
- Cani PD, Bibiloni R, Knauf C, Waget A, Neyrinck AM, Delzenne NM, Burcelin R. 2008. Changes in Gut Microbiota Control Metabolic Diet –Induced Inflammation in High-Fat Diet- Induced Obesity and Diabetes in Mice. Diabetes 57:1470–1481.
- 23. **Cani PD**, **Osto M**, **Geurts L**, **Everard A**. 2012. Involvement of gut microbiota in the development of low-grade inflammation and type 2 diabetes associated with obesity. Gut microbes 3:279–288.
- 24. Cani PD, Amar J, Iglesias MA, Poggi M, Knauf C, Bastelica D, Neyrinck AM, Fava F, Tuohy KM, Chabo C, Waget A, Delmée E, Cousin B, Sulpice T, Chamontin B, Ferrières J, Tanti J-F, Gibson GR, Casteilla L, Delzenne NM,

Alessi MC, Burcelin R. 2007. Metabolic endotoxemia initiates obesity and insulin resistance. Diabetes **56**:1761–1772.

- Neves AL, Coelho J, Couto L, Leite-Moreira A, Roncon-Albuquerque R. 2013. Metabolic endotoxemia: A molecular link between obesity and cardiovascular risk. Journal of Molecular Endocrinology 51:R51–R64.
- 26. Ewaschuk JB, Diaz H, Meddings L, Diederichs B, Dmytrash A, Backer J, Looijer-van Langen M, Madsen KL. 2008. Secreted bioactive factors from *Bifidobacterium infantis* enhance epithelial cell barrier function. American Journal of Physiology- Gastrointestinal and Liver Physiology 295:G1025–1034.
- 27. Wang Z, Xiao G, Yao Y, Guo S, Lu K, Sheng Z. 2006. The Role of Bifidobacteria in Gut Barrier Function After Thermal Injury in Rats. Journal of Trauma-Injury Infection & Critical Care 61:pp 650–657.
- Cani P, Delzenne N. 2009. The Role of the Gut Microbiota in Energy Metabolism and Metabolic Disease. Current Pharmaceutical Design 15:1546–1558.
- 29. Martínez I, Wallace G, Zhang C, Legge R, Benson AK, Carr TP, Moriyama EN, Walter J. 2009. Diet-induced metabolic improvements in a hamster model of hypercholesterolemia are strongly linked to alterations of the gut microbiota. Applied and Environmental Microbiology 75:4175–4184.
- Chen J, Wang RR-L, Li X-F, Wang RR-L. 2012. *Bifidobacterium adolescentis* supplementation ameliorates visceral fat accumulation and insulin sensitivity in an experimental model of the metabolic syndrome. The British journal of nutrition 107:1429–1434.

- Reichold A, Brenner SA, Spruss A, Förster-Fromme K, Bergheim I, Bischoff SC. 2014. *Bifidobacterium adolescentis* protects from the development of nonalcoholic steatohepatitis in a mouse model. The Journal of nutritional biochemistry 25:118–125.
- 32. Chen JJ, Wang R, Li X, Wang R. 2011. *Bifidobacterium longum* supplementation improved high-fat-fed-induced metabolic syndrome and promoted intestinal Reg I gene expression. Experimental biology and medicine (Maywood, NJ) **236**:823–831.
- 33. Plaza-Diaz J, Gomez-Llorente C, Abadia-Molina F, Saez-Lara MJ, Campana-Martin L, Munoz-Quezada S, Romero F, Gil A, Fontana L. 2014. Effects of *Lactobacillus paracasei* CNCM I-4034, *Bifidobacterium breve* CNCM I-4035 and *Lactobacillus rhamnosus* CNCM I-4036 on hepatic steatosis in Zucker rats. PLoS ONE 9:e98401.
- 34. Kondo S, Xiao J, Satoh T, Odamaki T, Takahashi S, Sugahara H, Yaeshima T, Iwatsuki K, Kamei A, Abe K. 2010. Antiobesity Effects of *Bifidobacterium breve* Strain B-3 Supplementation in a Mouse Model with High-Fat Diet-Induced Obesity. Bioscience, Biotechnology and Biochemistry 74:1656–1661.
- 35. Griffiths EA, Duffy LC, Schanbacher FL, Qiao H, Dryja D, Leavens A, Rossman J, Rich G, Dirienzo D, Ogra PL. 2004. *In Vivo* Effects of Bifidobacteria and Lactoferrin on Gut Endotoxin Concentration and Mucosal Immunity in Balb/c Mice. Digestive Diseases and Sciences 49:579–589.
- 36. **Davis LMG**, **Martínez I**, **Walter J**, **Goin C**, **Hutkins RW**. 2011. Barcoded pyrosequencing reveals that consumption of galactooligosaccharides results in a highly specific bifidogenic response in humans. PloS One 6:e25200.

- Milani C, Ferrario C, Turroni F, Duranti S, Mangifesta M, van Sinderen D,
   Ventura M. 2016. The human gut microbiota and its interactive connections to diet.
   Journal of Human Nutrition and Dietetics 1–8.
- 38. Fanaro S, Marten B, Bagna R, Vigi V, Fabris C, Peña-Quintana L, Argüelles F, Scholz-Ahrens KE, Sawatzki G, Zelenka R, Schrezenmeir J, De Vrese M, Bertino E. 2009. Galacto-oligosaccharides are bifidogenic and safe at weaning: a double-blind randomized multicenter study. Journal of Pediatric Gastroenterology and Nutrition 48:82–88.
- 39. Sierra C, Bernal M-J, Blasco J, Martinez R, Dalmau J, Ortuno I, Espin B, Vasallo M-I, Gil D, Vidal M-L, Infante D, Leis R, Maldonado J, Moreno JM, Romon E. 2015. Prebiotic effect during the first year of life in healthy infants fed formula containing GOS as the only prebiotic: a multicentre, randomised, doubleblind and placebo-controlled trial. European journal of nutrition 54:89–99.
- Davis LMG, Martínez I, Walter J, Hutkins R. 2010. A dose dependent impact of prebiotic galactooligosaccharides on the intestinal microbiota of healthy adults. International Journal of Food Microbiology 144:285–292.
- 41. Walton GE, van den Heuvel EGHM, Kosters MHW, Rastall RA, Tuohy KM, Gibson GR. 2012. A randomised crossover study investigating the effects of galacto-oligosaccharides on the faecal microbiota in men and women over 50 years of age. Br J Nutr 107:1466–1475.
- 42. Vulevic J, Juric A, Tzortzis G, Gibson GR. 2013. A Mixture of trans -Galactooligosaccharides Reduces Markers of Metabolic Syndrome and Modulates the Fecal Microbiota and Immune Function of Overweight Adults 1 – 3. Journal of
Nutrition 324–331.

- 43. Salonen A, Lahti L, Salojärvi J, Holtrop G, Korpela K, Duncan SH, Date P, Farquharson F, Johnstone AM, Lobley GE, Louis P, Flint HJ, de Vos WM.
  2014. Impact of diet and individual variation on intestinal microbiota composition and fermentation products in obese men. The ISME journal 8:2218–2230.
- 44. **Martínez I, Kim J, Duffy PR, Schlegel VL, Walter J**. 2010. Resistant starches types 2 and 4 have differential effects on the composition of the fecal microbiota in human subjects. PloS one **5**:e15046.
- Kolida S, Gibson GR. 2011. Synbiotics in health and disease. Annual Review of Food Science and Technology 2:373–393.
- Lawley TD, Walker AW. 2013. Intestinal colonization resistance. Immunology 138:1–11.
- 47. Rattanaprasert M, Roos S, Hutkins RW, Walter J. 2014. Quantitative evaluation of synbiotic strategies to improve persistence and metabolic activity of *Lactobacillus reuteri* DSM 17938 in the human gastrointestinal tract. Journal of Functional Foods 10:85–94.
- 48. Alander M, Mättö J, Kneifel W, Johansson M, Koegle B, Crittenden R, Mattila-Sandholm T, Saarela M. 2001. Effect of galacto-oligosaccharide supplementation on human faecal microflora and on survival and persistence of *Bifidobacterium lactis* Bb-12 in the gastrointestinal tract. International Dair 11:817–825.
- 49. **Malinen E, Mättö J, Salmitie M, Alander M, Saarela M, Palva A**. 2002. Analysis of *Bifidobacterium* populations in human faecal samples from a consumption trial

with *Bifidobacterium lactis* Bb-12 and a galacto-oligosaccharide preparation. Systematic and applied microbiology **25**:249–258.

- 50. Satokari RM, Vaughan EE, Akkermans AD, Saarela M, de Vos WM. 2001. Polymerase Chain Reaction and Denaturing Gradient Gel Electrophoresis Monitoring of Fecal *Bifidobacterium* Populations in a Prebiotic and Probiotic Feeding Trial. Systematic and applied microbiology 231:227–231.
- 51. Rafter J, Bennett M, Caderni G, Clune Y, Hughes R, Karlsson PC, Klinder A, O'Riordan M, O'Sullivan GC, Pool-Zobel B, Rechkemmer G, Roller M, Rowland I, Salvadori M, Thijs H, Van Loo J, Watzl B, Collins JK. 2007. Dietary synbiotics reduce cancer risk factors in polypectomized and colon cancer patients. Am J Clin Nutr 85:488–496.
- 52. Roller M, Clune Y, Collins K, Rechkemmer G, Watzl B. 2007. Consumption of prebiotic inulin enriched with oligofructose in combination with the probiotics *Lactobacillus rhamnosus* and *Bifidobacterium lactis* has minor effects on selected immune parameters in polypectomised and colon cancer patients. The British journal of nutrition **97**:676–684.
- 53. Taipale T, Pienihäkkinen K, Isolauri E, Larsen C, Brockmann E, Alanen P, Jokela J, Söderling E. 2010. *Bifidobacterium animalis* subsp. *lactis* BB-12 in reducing the risk of infections in infancy. The British journal of nutrition 105:409–416.
- 54. Martinez RCR, Aynaou AE, Albrecht S, Schols HA, De Martinis ECP, Zoetendal EG, Venema K, Saad SMI, Smidt H. 2011. In vitro evaluation of gastrointestinal survival of Lactobacillus amylovorus DSM 16698 alone and

combined with galactooligosaccharides, milk and/or *Bifidobacterium animalis* subsp. *lactis* Bb-12. International Journal of Food Microbiology **149**:152–158.

- 55. Wu GD, Chen J, Hoffmann C, Bittinger K, Chen Y, Keilbaugh SA, Bewtra M, Knights D, Walters WA, Knight R, Sinha R, Gilroy E, Gupta K, Baldassano R, Nessel L, Li H, Bushman FD, Lewis JD. 2011. Linking Long-Term Dietary Patterns with Gut Microbial Enterotypes. Science 334:105–109.
- 56. Kovatcheva-Datchary P, Nilsson A, Akrami R, Lee YS, De Vadder F, Arora T, Hallen A, Martens E, Bjoerck I, Baeckhed F. 2015. Dietary Fiber-Induced Improvement in Glucose Metabolism Is Associated with Increased Abundance of Prevotella. Cell Metabolism 22:971–982.
- 57. Tanaka R, Takayama H, Morotomi M, Kuroshima T, Ueyama S, Matsumoto K, Kuroda A, Masahiko M. 1983. Effects of Administration of TOS and *Bifidobacterium breve* 4006 on the Human Fecal Flora. Bifidobacteria and Microflora 2:17–24.
- 58. Walker AW, Ince J, Duncan SH, Webster LM, Holtrop G, Ze X, Brown D, Stares MD, Scott P, Bergerat A, Louis P, McIntosh F, Johnstone AM, Lobley GE, Parkhill J, Flint HJ. 2011. Dominant and diet-responsive groups of bacteria within the human colonic microbiota. The ISME journal 5:220–230.
- 59. Clavel T, Lepage P, Charrier C. 2013. The Family Coriobacteriaceae, p. 201–238. In Rosenberg, E, DeLong, EF, Lor, S, Stackebrandt, E, Thompson, F (eds.), The Prokaryotes, 4th ed. Springer.
- 60. Lawson PA, Greetham HL, Gibson GR, Giffard C, Falsen E, Collins MD. 2005.

*Slackia faecicanis* sp. nov., isolated from canine faeces. International Journal of Systematic and Evolutionary Microbiology **55**:1243–1246.

- Jin JS, Kitahara M, Sakamoto M, Hattori M, Benno Y. 2010. Slackia equolifaciens sp. nov., a human intestinal bacterium capable of producing equol. International Journal of Systematic and Evolutionary Microbiology 60:1721–1724.
- Kim KS, Rowlinson MC, Bennion R, Liu C, Talan D, Summanen P, Finegold SM. 2010. Characterization of *Slackia exigua* isolated from human wound infections, including abscesses of intestinal origin. Journal of Clinical Microbiology 48:1070–1075.
- 63. Nagai F, Watanabe Y, Morotomi M. 2010. Slackia piriformis sp. nov. and Collinsella tanakaei sp. nov., new members of the family Coriobacteriaceae, isolated from human faeces. International Journal of Systematic and Evolutionary Microbiology 60:2639–2646.
- 64. **Miyakawa H, Nakazawa F**. 2010. Role of Asaccharolytic Anaerobic Gram positive Rods on Periodontitis. Journal of Oral Bioscience **52**:240–244.
- 65. Tap J, Mondot S, Levenez F, Pelletier E, Caron C, Furet JP, Ugarte E, Muñoz-Tamayo R, Paslier DLE, Nalin R, Dore J, Leclerc M. 2009. Towards the human intestinal microbiota phylogenetic core. Environmental Microbiology 11:2574– 2584.
- 66. **Cardelle-Cobas A, Corzo N, Olano A, Pelaez C, Requena T, Avila M**. 2011. Galactooligosaccharides derived from lactose and lactulose: Influence of structure on *Lactobacillus*, *Streptococcus* and *Bifidobacterium* growth. International Journal

of Food Microbiology 149:81–87.

67. **Petrof EO, Khoruts A**. 2014. From stool transplants to next-generation microbiota therapeutics. Gastroenterology **146**:1573–1582.

#### **4.8 Supplements**



**Fig. 4.S1.** Average abundances of taxa in fecal samples of subjects consuming dietary treatments.

Bacterial quantities are expressed as percent abundances of total bacteria as determined by 16S rRNA sequencing. Significance of  $P \le 0.05$  is denoted by a single asterisk (\*).

Sporobacteriun

Anaerostipes

Anaerovorax

Eubacterium

Megasphaera

Phascolarctobacterium

Moryella

Dialister

Clostridium

Streptococcus

Lactobacillus

Bacteroides

Paraprevotella Prevotella

Barnesiella Parabacteroides

Parasutterella Sutterella

Akkermansia

others

Escherichia/Shigella Burkholderiales

Alistipes



**Fig. 4.S2.** Correlation analysis of fecal genera present with at least 0.5% abundance in subjects consuming dietary treatments. Bacterial quantities are expressed as percent abundances of total bacteria as determined by 16S rRNA sequencing.



**Fig. 4.S3.** Correlation analysis of fecal OTUs present with at least 1% abundance in subjects consuming dietary treatments. Bacterial quantities are expressed as percent abundances of total bacteria as determined by 16S rRNA sequencing.



**Fig. 4.S4.** Ratio of *Bacteroides* and *Prevetolla* taxa present in fecal samples of subjects consuming dietary treatments.



Fig. 4.S5. Alpha diversity comparison between persisters and non-persisters.

# ITT analysis

Table 4.S1.	Proportions of	bacterial tax	a significantly	influenced	by dietary	treatments
based on inte	end to treat analy	ysis.				

		Mean % bacterial abundance $\pm$ SD $^{\rm b}$						
Treatment		Baseline	Treatment	Washout	P value			
	Taxonomic group							
Group A	OTUs <sup>a</sup>							
Lactose control	OTU_2047 (Blautia sp.)	0.01±0.0 A	$0.01\pm0.0$ B	0.02±0.0 AB	0.0270			
	OTU_2511 (Blautia sp.)	0.89±1.2 A	1.06±1.6 B	0.91±1.1 AB	0.0278			
Group B IVS-1	OTUs <sup>a</sup>							
Probiotic	OTU_1 (B. adolescentis)	1.32±3.9 A	3.66±4.9 B	3.25±4.2 B	0.0130			
	OTU_2353 (Dorea sp.)	0.01±0.0 A	0.02±0.0 B	0.01±0.0 AB	0.0243			
Group C	OTUs <sup>a</sup>							
Bb12 Probiotic	OTU_2003 (unclass. Lachnospiraceae)	0.00±0.0 A	0.08±0.2 AB	0.04±0.1 B	0.0375			
Group D IVS-1 GOS	Family							
	Bifidobacteriaceae	$4.67{\pm}4.1~\mathrm{A}$	11.70±6.6 B	9.02±8.7 AB	0.0260			
	OTUs <sup>a</sup>							
	OTU_1 (B. adolescentis)	2.31±4.4 A	6.92±7.0 B	4.43±5.0 AB	0.0106			
	OTU_469 (Anaerotruncus sp.)	0.03±0.1 A	$0.00{\pm}0.0$ B	$0.01\pm0.0$ AB	0.0391			
Group E Bb12 GOS	OTUs <sup>a</sup>							
	OTU_156 (Clostridium XI sp.)	0.24±0.3 A	0.07±0.2 B	0.20±0.5 AB	0.0214			
Group F GOS	OTUs <sup>a</sup>							
	OTU_643 (Bacteroides sp.)	0.02±0.0 A	0.00±0.0 B	0.00±0.0 AB	0.0372			
	OTU_319 (Butyricimonas sp.)	0.00±0.0 AB	0.00±0.0 A	0.05±0.1 B	0.0382			

<sup>a</sup> If the strain could not be assigned to a type strain (<97% homology), RDP Classifier was used to determine the most likely genus, and the RDP Classifier value is shown (80% cutoff).

<sup>b</sup> Values with different uppercase letters are significantly different from each other.



Fig.4.S6.Averageabundances of taxa in fecalsamplesofsubjectsconsuming dietary treatmentsbasedonintendtotreatanalysis.Bacterial quantitiesareexpressedaspercentabundancesoftotalbacteriaasdeterminedby 16SrRNAsequencing.







**Fig. 4.S7.** Ratio of *Bacteroides* and *Prevetolla* taxa present in fecal samples of subjects consuming dietary treatments based on intend to treat analysis.



**Fig. 4.S8**. Alpha diversity comparison between persisters and non persisters based on intend to treat analysis.

#### Chapter 5

### Conclusions and future direction.

### **5.1 Conclusions**

The importance of gut microbiota on human health has been well demonstrated over the past two decades. New analytical methods, as well as bioinformatics tools have led to many new insights into this complex ecosystem. Food and diet have been shown to strongly associate with health and disease, and it has become increasingly recognized that human and animal health is profoundly affected by the specific types and proportions of microorganisms that inhabit the intestinal tract. There is now much interest in formulating dietary strategies to support a health associated gut microbiota. However, there is a considerable knowledge gap on how diet shapes the bacterial populations, which bacteria should be preferably enriched for to support host health, how to formulate dietary treatments in order to be most effective, and which ecological requirements have to be met in order to introduce new members into the gut microbiota.

One strategy to establish microbial members that confer health benefits to the host in the gut microbiota is with the application of probiotics, prebiotics, and synbiotics. As reviewed in Chapter 1, there is much clinical and commercial interest in the development of novel synbiotics. While the health claims made for synbiotic applications are currently outpacing the research, there is little known about the mechanisms by which probiotics and synbiotics become established in the GI tract. The development of novel synbiotics based on ecological requirements of the probiotic strains to survive and colonize the GI tract may enhance the beneficial health effects already observed for some of the synbiotic applications. However, a better understanding of the gastrointestinal niche, with its very specific biotic and abiotic factors, and the interaction of the resident microbiota is fundamental in order to develop novel probiotics and synbiotics.

The studies presented here first address ecological colonization factors that enable a truly symbiotic model organism, *L. reuteri*, to densely colonize its rodent host in Chapter 2, and secondly introduce and establish a novel method to select for putative probiotic strains and synbiotic combinations *in vivo* in Chapters 3 and 4.

Chapter 2 provided a better understanding of the phenotypic adaptations of a vertebrate gut symbiont, *L. reuteri*, that contribute to both specialization towards a particular host and a highly successful lifestyle. The analysis of this probiotic model organism provided new insights into the ecological requirements and challenges that probiotic strains face in the GI environment, specifically in the stomach. This study's objective was to systematically determine which genes of *L. reuteri* 100-23 contribute to tolerance towards host gastric acid secretion. There were three main findings of this study: (i) the urease cluster was the predominant factor in mediating resistance to gastric acid production; (ii) gene annotations and *in vitro* tests have limitations to predict the exact ecological functions of colonization factors of bacterial gut symbionts; and (iii) novel information was revealed on the mechanisms by which *L. reuteri* colonizes its gastric niche. Ultimately, the basic molecular research described in this chapter broadens our understanding of GI niches and the ecological challenges probiotic strains have to tolerate.

Chapters 3 and 4 focus on the effect of probiotic, prebiotic, and synbiotic applications on the microbial community in the GI tract. Relatively little is known about the effects of synbiotic treatments on the gut microbiota. In order to address this knowledge gap, a functional analysis of specific microbes to colonize the GI tract was done. In Chapter 3, we introduced *in vivo* selection (IVS) as a technique to select synergistic synbiotic combinations. In this approach putative probiotic strains are enriched in subjects by extended consumption of a prebiotic substrate, in this case GOS. A rat study was then conducted to assess the impact of the selected synergistic synbiotic in comparison to prebiotic and probiotic feeding alone. The main finding in this rat model of non-alcoholic fatty liver disease (NAFLD) was that even though the synbiotic did not influence host phenotypes, it was highly efficient at enhancing population levels of the probiotic strain, and making it the most dominant operational taxonomic unit in the GI tract of the rats. This was a novel finding that has not been reported in the probiotic literature. Our findings indicated that B. adolescentis IVS-1 not only had a higher affinity for GOS in vivo than the resident *Bifidobacterium* species but also utilized GOS to increase its competitiveness and effectively outcompete a closely related resident species. This study showed that IVS-1 and GOS were acting as a truly synergistic synbiotic in rats.

To test the potential of IVS-1 as a probiotic and as a synbiotic application (i.e., IVS-1 and GOS specifically) in human subjects, we conducted a clinical trial assessing the impact of the probiotic, prebiotic, and synbiotic treatments on the microbial community in the GI tract of human subjects suffering from obesity. As described in Chapter 4, we compared the rationally selected synbiotic to a commercial synbiotic that applied an allochthonous probiotic component (*B. animalis* subsp. *lactis* Bb12). This study aimed to

answer two important ecological questions: (i) does the *in vivo* selected autochthonous bacterial strain IVS-1 have an ecological advantage compared to the allochthonous bacterial strain Bb12 in the GI tract of human subjects; and (ii) does the supplementation of each strain with the prebiotic GOS support the colonization of the respective strain *in* vivo. Even though both strains were bifidobacteria, there was a significant difference in the colonization rate. IVS-1 colonized the GI tract in significantly higher numbers than Bb12 and even became most dominant operational taxonomic unit in the subjects. This suggested that the autochthonous strain IVS-1 had an ecological advantage over the allochthonous commercial strain Bb12. This finding supported our hypothesis that IVS can be used to select a bacterial strain that can substantially enhance population levels and the competitiveness of a putative probiotic strain in the gastrointestinal tract compared to a commercial probiotic. This was an important finding as it underlined the necessity to consider ecological requirements of the probiotic strains when synbiotic combinations are formulated. A clear synergistic effect between IVS-1 and GOS was previously established in rats in Chapter 3. However, in the human study the synergistic effect between IVS-1 and GOS did not reach significance.

In summary, the findings in Chapters 3 and 4 demonstrated that the competitive fitness of strain IVS-1 was increased by GOS, which supported the conclusion that *in vivo* selection can be a valuable technique to screen for synbiotic combinations with high synergism *in vivo*.

## **5.2 Future direction**

Our group is currently working on the metabolic data of the human trial described in Chapter 4. This additional data will allow us to correlate health symptoms and improvements with the gut microbiota of the study subjects. This analysis will give us important insight into to the health impacts of the prebiotic, probiotic and synbiotic treatment in comparison to a commercial synbiotic application.

The results presented in this thesis have clearly demonstrated that the application of IVS is likely to enhance the ecological performance of probiotic strains or live biotherapeutics within the habitats in which they are thought to function. The technology could be readily applied in the design of microbiota-modulating therapies, including novel and rational synbiotics.